87th Congress 1st Session }

JOINT COMMITTEE PRINT

INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

MATERIALS PREPARED FOR THE JOINT ECONOMIC COMMITTEE CONGRESS OF THE UNITED STATES

Part I

POSTWAR FLUCTUATIONS IN BUSINESS INVENTORIES



Printed for the use of the Joint Economic Committee

76626

U.S. GOVERNMENT PRINTING OFFICE WASHINGTON : 1961

For sale by the Superintendent of Documents, U.S. Government Printing Office Washington 25, D.C. - Price 50 cents

JOINT ECONOMIC COMMITTEE

(Created pursuant to sec. 5(a) of Public Law 304, 79th Cong.)

WRIGHT PATMAN, Texas, Chairman PAUL H. DOUGLAS, Illinois, Vice Chairman

HOUSE OF REPRESENTATIVES

SENATE

RICHARD BOLLING, Missouri HALE BOGGS, Louisiana HENRY S. REUSS, Wisconsin MARTHA W. GRIFFITHS, Michigan THOMAS B. CURTIS, Missouri CLARENCE E. KILBURN, New York WILLIAM B. WIDNALL, New Jersey JOHN SPARKMAN, Alabama J. W. FULBRIGHT, Arkansas WILLIAM PROXMIRE, Wisconsin CLAIBORNE PELL, Rhode Island PRESCOTT BUSH, Connecticut JOHN MARSHALL BUTLER, Maryland JACOB K. JAVITS, New York

WM. SUMMERS JOHNSON, Executive Director JOHN W. LEHMAN, Deputy Executive Director RICHAED J. BABBEB, Clerk

11

LETTERS OF TRANSMITTAL

DECEMBER 18, 1961.

To Members of the Joint Economic Committee:

Transmitted herewith for use by the Joint Economic Committee and other Members of the Congress is part I of a three-part series of papers prepared by experts from Government, the colleges, and research organizations. These materials have been assembled under the general title of "Inventory Fluctuations and Economic Stabilization."

The papers contained in this and the other two volumes will be discussed by their authors and other experts in a series of public hearings to be held early next year.

The papers have been prepared and the hearings are being arranged in accordance with the program of work set forth in the committee's annual report filed with the Congress May 2, 1961 (H. Rept. 328, 87th Cong., 1st sess., p. 47). This program provides for a "study of inventory movements, accumulation, and liquidation" in the following language:

Inventory fluctuation and behavior will be studied to try to determine the extent to which changes in inventories are causes of instability and to what extent they are in themselves affected by other forces inherent in the business cycle. The committee will be concerned with such areas as merchandising and production planning to see what influences and what can be done to regularize purchasing so that characteristically wide swings in the direction of inventory adjustments can be minimized.

The three sets of reports being transmitted are limited to the factfinding phase of the study outlined in the program.

The papers are presented in advance of the committee's hearings in accordance with a Joint Economic Committee practice of providing members of the committee and participating panelists an opportunity, whenever possible, to examine thoroughly the analyses and findings in preparation for the discussions at the hearings.

Sincerely yours,

WRIGHT PATMAN, Chairman.

DECEMBER 15, 1961.

Hon. WRIGHT PATMAN Chairman, Joint Economic Committee, House of Representatives, Washington, D.C.

DEAR MR. CHAIRMAN: Transmitted herewith is part I of a three part series of papers on the general subject of "Inventory Fluctuations and Economic Stabilization." The two papers in part I are devoted principally to a descriptive analysis of postwar inventory fluctuations. The papers in part II deal with the causative factors in movements in business inventories and the papers in part III with the relationship between inventory movements and economic instability. There are also included in part III a bibliography and a paper concerned with the availability and reliability of statistical data on inventories.

Paul G. Darling, on loan to the committee from Bowdoin College, has had major staff responsibility for formulating and directing this study.

Sincerely yours,

WM. SUMMERS JOHNSON, Executive Director. Part I consists of the following papers:

Postwar Cycles in Manufacturers' Inventories, by Thomas M. Stanback,	Page
Jr., New York University, National Bureau of Economic Research	VII
Analysis of Business Inventory Movements in the Postwar Period, pre- pared under the general direction of Louis J. Paradiso, Office of	
Business Economics, U.S. Department of Commerce	145
Part I—Factors Influencing Manufacturers' Inventories, by Mabel A. Smith	149
Part II—Inventory Investment in Trade, by Lawrence Bridge and Clement Winston	165

POSTWAR CYCLES IN MANUFACTURERS' INVENTORIES

BY

THOMAS M. STANBACK, Jr., New York University NATIONAL BUREAU OF ECONOMIC RESEARCH 1961

VII

.

nowledgments	
ntroduction and summary	
The role of inventory investment in business cycles: A preliminary	view_
Inventory investment during contractions	
Inventory investment during expansions	
Manufacturers' inventory investment	
Manufacturers' inventories and inventory investment	
Timing comparisons	
Cyclical sensitivity of durable goods	
Composition of manufacturers' stocks	
Purchased-materials inventories and inventory investment	
Behavior of inventories	
Timing and conformity of inventories: Comparisons with	n sales
turns	1 00100
Timing and conformity of inventories, Comparisons	with
reference turns	
Rehavior of inventory investment	
Timing and conformity of investment: Comparisons with	
furna	1 90109
Timing and conformity of investment: Comparisons	with
reference turns	** 1 011
Cyclical amplitude of inventory investment	
Timing and conformity of investment: Comparisons	with
reter of change in output	W1611
Factors influencing the behavior of nurchased materials	
Infiled order backlore	
Availability of materials	
Drise meyomente	
Inventory objectives and the numbering process	
Inventory objectives and the purchasing process	
Movements in new orders and purchased-materials i	nvest
ment.	
Cyclical changes in purchasing policy	
Summary and conclusions	
inished-goods inventories and inventory investment	
Composition of stocks	
The staple, made-to-stock inventory series	
Timing and conformity of stocks during activity cycles.	
Total finished-goods inventories	
Timing and conformity of stocks: Comparisons with sale	es and
reference turns	
The staple, made-to-stock inventory investment series	
Inventory investment and rates of change in activity	
Relation between staples made to stock and total ma	nufac-
turers' finished goods	
Total finished-goods inventory investment	
Investment and rates of change in sales	
Timing of investment during business cycles	
Timing and conformity of investment: Comparison with	ı sales
turns	
Timing and conformity of investment: Comparison with	h ref
erence turns	
Amplitude of durables and nondurables investment during	cycles
	-,
Summary	

m

6.	Goods-in-process inventories and inventory investment
	Abramovitz' theory of goods-in-process behavior
	Goods-in-process inventories
	Major patterns of movement in stocks
	Timing and conformity of stocks: Comparison with output and sales turns
	Timing of stocks during business cycles
	Goods in-process inventory investment
	Timing and conformity of investment: Comparison with out- put, sales, and reference turns
	Analysis and revision of Abramovitz' theorySummary
	Cyclical behavior of inventory and inventory investment movements
	Purchased materials
	Finished goods
	Cooda in process
	Total stocks and inventory investment
	A malitude of investment of investment of the second
	Amplitude of investment movements compared.
	Postwar and prewar timing compared
	Conclusion
•	How inventory movements contribute to instability
	Metzler's inventory cycle theory
	Observations on the validity of Metzler's theory
	Behavior of consumption and saving
	Timing of inventory investment and rates of change in final purchases
	Behavior of inventory investment under conditions of changing inventory objectives
	The occurrence of "extra" investment cycles
	The relative importance of changes in investment during recession
	Factors contributing to turning points
	Conclusions
	Conclusions.

APPENDIXES

A.	Sources and processing of data	128
B.	Inventory data	132

TABLES

1.	Changes in gross national product and nonfarm inventory investment,	6
2.	Ranking of eight contractions according to magnitude of change in nonfarm inventory investment relative to change in final purchases, 1920-58	9
3.	Annual changes in gross national product and nonfarm inventory investment compared, 1919-38, 1945-60.	10
4.	Changes in final purchases and nonfarm inventory investment, eight expansions, 1919-60.	11
5.	Changes in gross national product and manufacturers' inventory investment, 1919-60	13
6.	Annual changes in gross national product and manufacturers' inventory investment compared, 1919-38, 1945-60	14
7.	Analysis of relative size of cyclical changes in manufacturers' inventory investment 1919-60	15
8.	Timing and conformity of manufacturers' total inventory at reference	18
9.	Timing and conformity of manufacturers' total inventory investment	10
10.	at reference and activity turns; comprehensive series	22
201	investment accounted for by inventory investment of durables manufactures, 1919-61	26

11.	Timing and conformity of manufacturers' purchased-materials in-	Page
	ventories at sales turns	29
12.	Timing and conformity of manufacturers' purchased-materials in-	32
13.	Timing and conformity of manufacturers' purchased-materials inven-	
14	tory investment to sales turns	34-35
14.	tory investment at reference turns	38
15.	Amplitude of change in purchased-materials inventory investment	
16	cycles, by phase, 1948–58	39
10.	in output	40
17.	Timing of vendor performance series to quarterly change in unfilled	
	1947–58	49
18.	Timing of purchased-materials inventory investment to new orders	53
19.	Timing comparisons: Vendor performance, purchased-materials in-	
	1946–58	57
20.	Estimated composition of finished-goods stocks, 1939 and 1947	63 65
21.	Timing and conformity of 25 commodities at activity turns, 1947-55.	05
	sales turns	68
23.	Timing and conformity of manufacturers' finished-goods inventories at	69
24.	Distribution of timing of inventory investment turns (inverted) by	
05	quartile intervals of activity cycles, 25 finished goods, 1947–55	70
25. 26.	Timing of inished-goods investment to rate of change in sales	19
	investment at sales turns	80-81
27.	'Timing and conformity of manufacturers' finished-goods inventory in-	82
28.	Amplitude of change in finished-goods inventory investment cycles, by	-02
•••	phase, 1948–58	.83
29.	at output and sales turns	89
30.	Timing of manufacturers' goods-in-process inventories at reference	. 01
31.	Timing and conformity of manufacturers' goods-in-process inventory	91
011	investment at output and sales turns	93-94
32.	Timing of manufacturers' goods-in-process inventory investment at	94
33.	Timing of inventories and inventory investment at business cycle	01
94	turns: Summary	01-102
ə4 .	58	107
35.	Amplitude of change in inventory investment cycles, 1948-58, as	109
36	Timing of selected inventory investment series to related nurchase or	108
	sales series	117

APPENDIX TABLES

A-1. Weights a	and turnover periods assigned to industry deflators	_ 130
A-2. Index nun	nbers and weights used in constructing industry deflators	. 131
B-1. Total man	nufacturing, inventories	132–133
B-2. Durable-g	oods industries, total inventories	134-135
B-3. Nondurab	le-goods industries, total inventories	136 -1 37
B-4. Nonfarm	inventory investment	138139
B-5. Inventory	investment: Summary	140–144

.

CHARTS

Page 7	Gross national product, final purchases and inventory investment,	1.
'	Gross national product, final purchases and inventory investment.	2.
8	quarterly at annual rates, 1947-61	 0
19	goods industries, 1945–61	J.
23	Inventory investment: Total manufacturing, durable- and nondurable- goods industries, 1945-61	4.
-0	Purchased-materials inventories and sales: Total manufacturing,	5.
əə	Purchased-materials inventory investment, new orders, and change in	6
35–37	index of industrial production, 1946-58	
43-45	orders, and purchased-materials investment, 1946-58	1.
50	filled orders to sales, all manufacturers, 1946–58	8.
55	Purchasing policy, vendor performance, total manufacturers' pur-	9.
00	Finished-goods inventories and sales: Total manufacturing, durable-	10.
67 73	And nondurable-goods industries, 1940-38	11.
	activity: 25 commodities	
75	Finished-goods investment: Total manufacturing and cumulated net percentage of investment series expanding (25 commodities)	12.
77–78	in sales, 1946–58	13.
88	. Goods-in-process inventories and sales: Total manufacturing, durable-	14.
00	Goods-in-process inventory investment: Total manufacturing, durable-	15.
3-105	Manufacturers' inventory investment: Summary 10	16.
18-119	Quarter-to-quarter change in final purchases and sales, and inventory investment, selected series, 1946–5811	17.

NATIONAL BUREAU OF ECONOMIC RESEARCH

STUDIES IN BUSINESS CYCLES

- 1. "Business Cycles: The Problem and Its Setting," by Wesley C. Mitchell.
- 2. "Measuring Business Cycles," by Arthur F. Burns and Wesley C. Mitchell.
- 3. "American Transportation in Prosperity and Depression," by Thor Hultgren.
- 4. "Inventories and Business Cycles, with Special Reference to Manufacturers' Inventories," by Moses Abramovitz.
- 5. "What Happens During Business Cycles: A Progress Report," by Wesley C. Mitchell.
- 6. "Personal Income During Business Cycles," by Daniel Creamer, with the assistance of Martin Bernstein.
- "Consumption and Business Fluctuations: A Case Study of the Shoe, Leather, Hide Sequence," by Ruth P. Mack.
 "International Financial Transactions and Business Cycles," by
- 8. "International Financial Transactions and Business Cycles," by Oskar Morgenstern.
- 9. "Federal Receipts and Expenditures During Business Cycles, 1879–1958," by John M. Firestone.
- 10. "Business Cycle Indicators": Volume I, "Contributions to the Analysis of Current Business Conditions"; Volume II, "Basic Data on Cyclical Indicators"; edited by Geoffrey H. Moore.
- 11. "Postwar Cycles in Manufacturers' Inventories," by Thomas M. Stanback, Jr.

хш

LETTER OF TRANSMITTAL

NATIONAL BUREAU OF ECONOMIC RESEARCH, INC., New York, N.Y., October 17, 1961.

Hon. WRIGHT PATMAN, Chairman, Joint Economic Committee, House Office Building, Washington, D.C.

DEAR REPRESENTATIVE PATMAN: On behalf of the National Bureau of Economic Research I am happy to transmit to you, in response to your request of September 5, 1961, the report "Postwar Cycles in Manufacturers' Inventories," by Dr. Thomas M. Stanback, Jr.

Dr. Stanback's report is the culmination of several years of careful and systematic research, which he began as a research associate at the National Bureau in 1955–56 and continued since despite the pressure of teaching duties. He is presently associate professor of economics at New York University. Fortunately the completion of his work on the study earlier this year coincided with the plans of the Joint Economic Committee for a comprehensive study of the subject of inventories. We are therefore glad to make the report available to the committee. The report will also be published separately by the National Bureau as one of its studies in business cycles.

As the report itself indicates, it has had the benefit of a thorough review by members of the National Bureau's staff and board. The cooperation of the Office of Business Economics of the Department of Commerce was invaluable. Of great assistance, too, in the planning and conduct of the study, was the earlier work by Prof. Moses Abramovitz, issued by the National Bureau in 1950, under the title "Inventories and Business Cycles." Many of Abramovitz' findings, based on prewar information, are verified in Stanback's study, some 'are modified, and a number of new results are reported.

We hope that the report will be of assistance to the committee in its investigation of this important subject.

Very truly yours,

SOLOMON FABRICANT, Director of Research.

XIV

LETTER OF TRANSMITTAL

NATIONAL BUREAU OF ECONOMIC RESEARCH, 1961

OFFICERS

Harold M. Groves, Chairman Arthur F. Burns, President Albert J. Hettinger, Jr., Vice President Murray Shields, Treasurer Solomon Fabricant, Director of Research Geoffrey H. Moore, Associate Director of Research Hal B. Lary, Associate Director of Research William J. Carson, Executive Director

DIRECTORS AT LARGE

Wallace J. Campbell, Nationwide Insurance Erwin D. Canham, Christian Science Monitor Solomon Fabricant, New York University Marion B. Folsom, Eastman Kodak Co.

Crawford H. Greenewalt, E. I. du Pont de Nemours & Co.

Gabriel Hauge, Manufacturers Hanover Trust Co.

A. J. Hayes, International Association of Machinists

Albert J. Hettinger, Jr., Lazard Frères & Co. H. W. Laidler, League for Industrial Democracy George B. Roberts, Larchmont, N.Y. Harry Scherman, Book-of-the-Month Club

Boris Shishkin, American Federation of Labor and Congress of Industrial Organizations

George Soule, South Kent, Conn. Joseph H. Willits, Armonk, N.Y. Donald B. Woodward, A. W. Jones & Co.

Theodore O. Yntema, Ford Motor Co.

DIRECTORS BY UNIVERSITY APPOINTMENT

V. W. Bladen, Toronto Arthur F. Burns, Columbia Lester V. Chandler, Princeton Melvin G. de Chazeau, Cornell Frank W. Fetter, Northwestern R. A. Gordon, California Harold M. Groves, Wisconsin

Gottfried Haberler, Harvard Walter W. Heller, Minnesota Maurice W. Lee, North Carolina Lloyd G. Reynolds, Yale Theodore W. Schultz, Chicago Willis J. Winn, Pennsylvania

DIRECTORS BY APPOINTMENT OF OTHER ORGANIZATIONS

Percival F. Brundage, American Institute of Certified Public Accountants Harold G. Halcrow, American Farm Economic Association Theodore V. Houser, Committee for Economic Development S. H. Ruttenberg, American Federation of Labor and Congress of Industrial Organizations

Murray Shields, American Management Association Willard L. Thorp, American Economic Association W. Allen Wallis, American Statistical Association Harold F. Williamson, Economic History Association

DIRECTORS EMERITI

Oswald W. Knauth, Beaufort, S. C. Shepard Mor. N. I. Stone, New York City Shepard Morgan, Norfolk, Conn.

RESEARCH STAFF

Moses Abramovitz Gary S. Becker William H. Brown, Jr. Gerhard Bry Arthur F. Burns Phillip Cagan Joseph W. Conard Frank G. Dickinson James S. Earley Richard A. Easterlin Solomon Fabricant Milton Friedman

Raymond W. Goldsmith Millard Hastay Daniel M. Holland Thor Hultgren F. Thomas Juster C. Harry Kahn Simon Kuznets Hal B. Lary Robert E. Lipsey Ruth P. Mack Jacob Minaar Jacob Mincer

Ilse Mintz Geoffrey H. Moore Roger F. Murray Ralph L. Nelson G. Warren Nutter Richard T. Selden Lawrence H. Seltzer Robert P. Shay George J. Stigler Norman B. Ture ¶ Herbert B. Woolley

RELATION OF THE DIRECTORS TO THE WORK AND PUBLICATIONS OF THE NATIONAL BUREAU OF ECONOMIC RESEARCH

1. The object of the National Bureau of Economic Research is to ascertain and to present to the public important economic facts and their interpretation in a scientific and impartial manner. The Board of Directors is charged with the responsibility of insuring that the work of the National Bureau is carried on in strict conformity with this object.

2. To this end the Board of Directors shall appoint one or more Directors of Research.

3. The Director or Directors of Research shall submit to the members of the Board, or to its Executive Committee, for their formal adoption, all specific proposals concerning researches to be instituted.

4. No report shall be published until the Director or Directors of Research shall have submitted to the Board a summary drawing attention to the character of the data and their utilization in the report, the nature and treatment of the problems involved, the main conclusions, and such other information as in their opinion would serve to determine the suitability of the report for publication in accordance with the principles of the National Bureau.

5. A copy of any manuscript proposed for publication shall also be submitted to each member of the Board. For each manuscript to be so submitted a special committee shall be appointed by the President, or at his designation by the Executive Director, consisting of three Directors selected as nearly as may be one from each general division The names of the special manuscript committee shall of the Board. be stated to each Director when the summary and report described in paragraph (4) are sent to him. It shall be the duty of each member of the committee to read the manuscript. If each member of the special committee signifies his approval within 30 days, the manuscript may be published. If each member of the special committee has not signified his approval within 30 days of the transmittal of the report and manuscript, the Director of Research shall then notify each member of the Board, requesting approval or disapproval of publication, and 30 additional days shall be granted for this purpose. The manuscript shall then not be published unless at least a majority of the entire Board and a two-thirds majority of those members of the Board who shall have voted on the proposal within the time fixed for the receipt of votes on the publication proposed shall have approved.

6. No manuscript may be published, though approved by each member of the special committee, until 45 days have elapsed from the transmittal of the summary and report. The interval is allowed for the receipt of any memorandum of dissent or reservation, together with a brief statement of his reasons, that any member may wish to express; and such memorandum of dissent or reservation shall be published with the manuscript if he so desires. Publication does not, however, imply that each member of the Board has read the manuscript, or that either members of the Board in general, or of the special committee, have passed upon its validity in every detail.

7. A copy of this resolution shall, unless otherwise determined by the Board, be printed in each copy of every National Bureau book.

(Resolution adopted October 25, 1926, as revised February 6, 1933, and February 24, 1941.)

ACKNOWLEDGMENTS

This study was begun in the academic year 1955-56, while the author was a research associate at the National Bureau of Economic Research, and was continued thereafter under Bureau sponsorship. At almost every stage the work benefited from the assistance, criticism, and encouragement of members of the Bureau staff. My obligations are many—more than I can fully acknowledge here.

My greatest debt is to Geoffrey H. Moore, who followed the study from the beginning, giving unstintingly of his time in reading and criticizing the various versions of the manuscript. Ruth Mack, Moses Abramovitz, and Victor Zarnowitz comprised the staff reading committee; R. A. Gordon, Maurice Lee, and George B. Roberts, the Board of Directors' review committee. I am grateful to all of these persons for their careful reading of the manuscript and for the many helpful suggestions which were offered. Thanks go also to Dorothy Suchman, who was my research assistant; to Johanna Stern, who was helpful in many ways in assembling the data; and to Irving Forman, who drew the charts. Joan Tron's high standard of editorial craftsmanship contributed immeasurably to the final product. Finally, I wish to express my gratitude to my wife for her patience and confidence over the extended period of preparation of this study.

THOMAS M. STANBACK, Jr.

XVII

76626-61-pt. I-2

CHAPTER I

INTRODUCTION AND SUMMARY

Changes in inventory investment are widely regarded as a major indicator of cyclical instability in the American economy, and have been closely followed in recent years by economists, businessmen, and the press. This study deals principally with the behavior of manufacturers' stocks—the principal component of nonfarm inventories—since the end of World War II. Available aggregates for all manufacturers, durable-goods manufacturers, and nondurable-goods manufacturers, as well as certain commodity series, are used. Attention is concentrated on the period 1947–56, although certain parts of the analysis have been extended to the end of 1958 in order to include the 1957–58 business cycle contraction. Comparisons are also made with prewar behavior when information is available.

A major objective has been to test the relevance for the postwar years of the principal findings and hypotheses of Moses Abramovitz as set forth for the interwar years in his "Inventories and Business Cycles, With Special Reference to Manufacturers' Inventories."¹ Since his earlier investigation has influenced the approach and organization of this study, it is well to comment here on Abramovitz' work. His most important contributions were the determination of typical patterns of timing and the relative importance of movements in manufacturers' inventories during the interwar business cycles, the development of an explanation for the patterns he found, and the conclusion that the overall movement in manufacturers' stocks should, for purposes of analysis, be regarded as a complex of somewhat dissimilar movements of a number of different types of stocks.

In spite of severe limitations necessitated $b\bar{y}$ principal reliance on annual data, Abramovitz was able to estimate the timing of stocks and inventory investment at business cycle turns, and to establish the general magnitude of movements in expansions and contractions.

In studying the behavior of the various components, he analyzed stocks successively at each stage of fabrication (i.e., purchased materials, goods in process, finished goods), breaking down these classes still further in an attempt to establish the nature and significance of the patterns of movement. Essentially he tried to determine the size of each type of stock and the factors which influenced its fluctuations, examining such commodity series as were available in order to check his hypotheses and shed additional light on cyclical characteristics.

The organization of the present study is similar to Abramovitz'. Chapter 2 examines the importance of nonfarm and manufacturers' inventory investment in the postwar and prewar periods. Chapter 3 analyzes the behavior of total stocks held by all manufacturers, durable-goods and nondurable-goods manufacturers, and by individual industries. In chapters 4, 5, and 6 the behavior of stocks by stage of fabrication is treated in the sequence of purchased materials, finished

1

¹ New York, National Bureau of Economic Research, 1950.

goods, and goods in process. Chapter 7 summarizes the timing characteristics and describes the behavior of manufacturers' inventories as a composite of these three major types of stocks. Chapter 8 examines the manner in which inventory investment movements contribute to cyclical instability. Finally, an appendix is provided containing notes on sources and methods of processing the data, as well as the principal time series used.

For the convenience of the reader the conclusions in chapters 2 through 8 are summarized below:

Changes in nonfarm inventory investment, as indicated in chapter 2, have contributed significantly to business cycle contractions. Measured on the basis of annual data, they have accounted for at least 42 percent of the decline in national product in every business cycle contraction of the interwar and postwar periods except that of 1929-32. With this exception, there is no evidence that the role of movements in nonfarm inventory investment during business cycle contractions has changed significantly since the prewar period.

Movements in nonfarm inventory investment have contributed to changes in national product to a lesser degree during expansions than during contractions—a tendency which has increased since World War II. This is due to the greater length of postwar expansion phases, as well as to the earlier occurrence of the peaks in inventory investment.

Movements in manufacturers' inventory investment have played a somewhat larger role since the war, accounting for 83 percent of those in nonfarm inventory investment, compared with 56 percent in the prewar period. Although turns in manufacturers' inventory investment have led business cycle turns since the war, investment movements have conformed closely to general business contractions, the sharpest declines occurring between cyclical peaks and troughs.

Chapter 3 shows that total manufacturers' stocks have lagged behind business cycle turns from 1 to 8 months, in contrast to Abramovitz' estimate of a 6-to-12-month prewar lag. Manufacturers' inventory investment has moved in well-defined cyclical patterns and has demonstrated a high degree of conformity to business cycles. Timing of the total inventory investment series relative to business cycles varied from roughly coincident to a lead of 14 months.

The durable-goods inventory and investment series have been cyclically more sensitive than those of nondurable goods, conforming more closely to business cycle movements and moving with greater amplitude. The explanation appears to be that the production and sale of these goods are subject to greater amplitude of movement and that fluctuations in these industries tend to occur at about the same time. In addition, of stocks held at each stage of fabrication, durables appear to be more responsive to cyclical influences.

Manufacturers' inventories are significantly smaller relative to sales than before World War II. Moreover, composition has changed from approximately 40 percent purchased materials, 20 percent goods in process, and 40 percent finished goods before the war—to an estimated 38, 29, and 33 percent in 1952–53. The principal reason for this change is the increased importance of the durable-goods industries which carry larger goods-in-process stocks.

The purchased-materials stocks discussed in chapter 4 conform well to cyclical movements, turning earlier at sales and business cycle

INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION 3

peaks than at troughs. Turns tend to coincide with business cycle peaks and to lag at troughs. On the other hand, investment leads all turns. In these movements the durables series show a higher conformity and greater amplitude.

Abramovitz' theory, although it goes far to explain the general behavior of purchased-materials inventories and investment, does not provide for the occurrence of certain observed timing in the inventory and investment series. Moreover, it fails to consider the possible role of unfilled order backlogs, availability of materials, and price behavior of purchased goods in influencing inventory behavior. In an effort to provide a fuller explanation, these influences are examined. It is noted that the amplitude of purchased-materials investment movement appears to be related to the size and direction of movement of unfilled order backlogs and that the timing of investment tends to agree with certain measures of availability of materials. Price changes though significant do not appear to play a vital causal role.

Finally, inventory objectives and purchasing policy of firms vary cyclically, moving in general agreement with changes in availability of materials and realized investment.

Postwar finished-goods behavior (ch. 5) was studied by analyzing two sets of inventory data. The first was composed of data for 25 commodities, representing the largest component of finished goods: staple made-to-stock commodities. The second was composed of Department of Commerce industry data. Among the commodity series, finished stocks are found to move in a strongly inverted fashion relative to sales or output, but with a well-developed tendency to turn and come into phase as the duration of the activity phase increases. The Department of Commerce industry data reveal a consistent tendency for stocks to lag behind sales and reference turns, but movements are rarely of the completely inverted sort observed in the commodity series. It is concluded that the staple made-to-stock inventories contribute a strong tendency toward inverted behavior, but that other categories offset inverted tendencies to a significant degree.

Study of inventory investment data for the 25 commodities reveals that investment moves in an inverted fashion relative to rates of change in sales, but that peaks (troughs) in inventory investment lead or turn coincidentally with troughs (peaks) in rates of change in activity. Since peaks and troughs in the latter tend to occur well before the end of each business cycle phase, inverted movements in finished-goods inventory investment may be expected to terminate relatively early in the business cycle phase. Analysis of finished-goods inventory investment in the Department of Commerce series shows this to be the case. Total manufacturers' finished-goods investment turns roughly coincidently with business cycle turns or moves in an inverted fashion (i.e., lags) for only a few months in the early part of a business cycle phase.

Problems of deflation render conclusions regarding goods-in-process behavior less dependable than those for purchased materials and finished goods, but certain characteristics may be noted (ch. 6). The deflated inventory series conform to business cycles with virtually coincident timing at four of the six reference cycle turns, and inventory investment leads all business cycle turns. Analysis of timing sequence among the industry series indicates that, for both inventory and investment, goods in process lead or turn coincidently with purchased materials and lead the finished-goods series.

Although these findings are generally consistent with Abramovitz' theory, irregularities in timing appear, particularly among the individual industry investment series, which cannot be explained by his Moreover, durable-goods investment appears to be influanalysis. enced by levels of unfilled orders in a manner similar to purchased (1) The Three factors seem to contribute to this behavior. materials. composition of postwar goods in process is different from that observed by Abramovitz. Owing to the increased role of durables, there is a much larger proportion of these stocks held by industries engaging in discontinuous processes—roughly 50 percent compared with his estimate of 36 percent. (2) Goods-in-process stocks in discontinuous-process industries are likely to be held in substantial quantities between stages. These stocks may be expected to rise and fall in a manner similar to that of purchased materials. (3) Goods held within stages need not fluctuate in as close conformity to changes in output as Abramovitz maintained.

In chapter 7, analysis of amplitude of investment movements reveals that inventory change has contributed significantly to cyclical instability at each stage of fabrication, the three components combining to form a total manufacturing inventory complex which is highly sensitive to cyclical forces. A summary of timing measures indicates that purchased-materials and goods-in-process inventories and investment tend to have similar timing characteristics, leading turns in the comparable finished-goods series. But the last lags behind the other investment series by no more than 6 months at four of the six business cycle turns, and shows almost no countercyclical tendencies.

A brief comparison of Abramovitz' estimates of prewar timing with postwar findings is presented. Where timing differences are noted, possible reasons for the differences are discussed.

Chapter 8 attempts to answer two fundamental questions: How do changes in inventory investment contribute to the cumulative forces of business cycle expansion and contraction? Is it possible that movements in inventory investment may spark the upswing or bring about the downturn, thereby constituting an underlying cause as well as an aggravating force?

In an examination of Lloyd Metzler's theory of the inventory cycle, it is noted that, in addition to the forces he discusses, inventory movements may be influenced by cyclical changes in the availability of goods and in the purchasing policies of firms. Inventory objectives may vary to a considerable degree as the expansion progresses and may be sharply revised when goods become readily available. Moreover, in making purchasing decisions, firms consider not only inventories on hand but also unfilled purchase orders. In the theory presented here it is held that changes in inventory objectives may contribute to cyclical instability and, under certain conditions, cause inventory investment to reverse, thereby contributing to a turn in the cyclical tide.

CHAPTER 2

THE ROLE OF INVENTORY INVESTMENT IN BUSINESS CYCLES: A PRELIMINARY VIEW

In the present chapter it will be shown that during both the prewar and postwar periods, changes in nonfarm inventory investment have made a major contribution to changes in national product during contractions, and a significant but less important contribution during expansions. Typically, the impact during expansions is greatest in its early stages. Increases in inventory investment since World War II have comprised a significantly smaller part of rises in national product than they did before the war. Of great importance for the present study is the observation that movements in manufacturers' inventory investment during both expansion and contraction phases of the business cycle account for a major part of the changes in total nonfarm inventory investment.

INVENTORY INVESTMENT DURING CONTRACTIONS

Table 1 and charts 1 and 2 present evidence of the importance of nonfarm inventory investment movements in both prewar and postwar business cycles. Measured on the basis of annual data, the changes are found to have accounted for at least 42 percent of the decline in national product in every business cycle contraction except that of 1929-32.¹ Quarterly data, available for the postwar period only, indicate that changes in investment comprised 195, 62, and 35 percent of the declines in national product from the peak to the trough quarters of the recessions of 1948-49, 1953-54, and 1957-58.

¹ Inventory investment (i.e., the changes in stocks), like manufacturers' investment in durable equipment, is a component of gross national product. Accordingly, it is the change in inventory investment that must be compared with changes in GNP.

6 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

	EATAI	1910119					
Business cycle phase	Change in GNP		Change in nonfarm inventory invest- ment		Col. 3÷ col. 1	Col. 4÷ col. 2	
	Annual data (1)	Quarterly data (2)	Annual data (3)	Quarterly data (4)	(percent) (5)	(percent) (6)	
Prewar (billions of 1929 dollars): 1919-20	+\$1.4 +14.6 +8.2 +7.6 +28.7	(1) (1) (1) (1) (1) (1)	+\$0. 4 +2. 4 +1. 5 +2. 3 +5. 7	8.8.9 8.8.9 8.9	29 16 18 30 20	() () () () () () () () () () () () () (
Total, 1919-37 Postwar (billions of 1954 dollars): 1946-48 1949-53 1954-57 1958-60	60. 5 +10. 4 +76. 3 +45. 5 +39. 5	(1) +\$80.2 +48.9 +48.2	$12.3 \\ -6.1 \\ +3.7 \\ +2.8 \\ +6.2$	(¹) +\$9.5 +4.0 +9.0	20 58 -5 -6 16	(1) (1) 12 8 19	
Total, 1949-60 ³ Average (prewar and postwar) ²	161.3	177.3	12.7	22.5	8 14	(¹) ¹³	
	CONTRA	CTIONS					
Prewar (billions of 1929 dollars): 1920-21 1923-24 1926-27 1920-27 1929-32 1937-38	$ \begin{array}{c} -\$3.6\\ +1.5\\ +1.0\\ -32.0\\ -3.1 \end{array} $	(1) (1) (1) (1) (1)	-\$2.9 -3.2 -1.3 -5.3 -3.6	(1) (1) (1) (1) (1)	81 213 130 17 116	(1) (1) (1) (1) (1)	
Total, 1920-38 Postwar (billions of 1954 dollars): 1948-49 1955-54 1957-58	36.2 4 5.9 7.3	(1) -\$4.3 -11.1 -15.8	16.3 5.6 3.2 3.1	(1) -\$8.4 -6.8 -5.6	45 1, 400 54 42	(1) 195 62 35	
Total, 1948–58 Average (prewar and postwar)	13.6	-31.2	-11.9	20.8	88 66	(1) 67	
	FULL (YCLES					
Prewar (billions of 1929 dollars): 1919-21 1921-24 1924-27 1927-32 1932-38	\$5.0 13.1 7.2 39.6 31.8	(1) (1) (1) (1) (1) (1) (1)	\$3.3 5.6 2.8 7.6 9.3	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	66 43 39 19 29	(1) (1) (1) (1) (1)	
Total, 1919-38 Postwar (billions of 1954 dollars): 1949-54 1954-58.	96.7 82.2 52.8	(1) \$91.3 64.7	28.6 6.9 5.9	(1) \$16.3 9.6	30 8 11	(¹) 9 15	
Total, 1949–58 Average (prewar and postwar)	135.0	156.0	12.8	25. 9	9 20	(1) 17	

TABLE 1.—Changes in gross national product and nonfarm inventory investment, 1919-60

EXP	A N	SIC)NS
		~~~	

Not available.
 Expansion of 1946-48 is omitted.

.

Note.-Quarterly comparisons are based on business cycle reference dates.

Source: Prewar GNP data, compiled from Simon Kuznets, "National Product Since 1869" (NBER, 1946), table I-15; inventory data from Moses Abramovitz, "Inventories and Business Cycles" (NBER 1950), p. 568. Postwar GNP and nonfarm inventory investment changes based on data in Survey of Current Business 1954 National Income Supplement and July 1961 issue.

CHART 1



GROSS NATIONAL PRODUCT, FINAL PURCHASES, AND INVENTORY INVESTMENT, ANNUAL TOTALS, 1919-60

[Billions of dollars]

➡ Final purchases equal GNP less total inventory investment. Shaded areas represent business contractions; unshaded areas, expansions.

Source: Prewar data for gross national product and final purchases, from tables I-15 and I-11, Simon Kuznets, "National Product Since 1869," NBER 1946; for inventory investment, from Moses Abramovitz, "Inventories and Business Cycles," NBER 1950. Postwar data from Department of Commerce.

#### CHART 2

GROSS NATIONAL PRODUCT, FINAL PURCHASES, AND INVENTORY INVESTMENT, QUARTERLY AT ANNUAL RATES, 1947-61



Final purchases equal GNP less total inventory investment.
 Shaded areas represent business contractions; unshaded areas, expansions.
 Source: Department of Commerce.

It is difficult with certainty to rank the various contractions according to the relative importance of inventory investment movements, but comparison is facilitated by making use of final purchases (gross national product less nonfarm and farm inventory investment) rather than national product data. In table 2 changes in nonfarm inventory investment are compared with changes in final purchases; phases are ranked according to what appears to be the relative contribution of the former. It will be noted that in terms of annual data, final purchases rose during the contraction of 1948-49 and during all the prewar contractions except those of 1929-32 and 1937-38. Declines in inventory investment may, therefore, be regarded as the principal cause of contraction. The contractions of 1923-24 and 1948-49 belong at the top of the list in this respect, but it is debatable which should be accorded first place.

8

		Change in nonfarm inventory investment		Change in final pur- chases		Col. 1÷	Col. 2÷
Rank	Contraction	Annual data	Quarterly data ²	Annual data	Quarterly data ²	(percent)	(percent)
		(1)	(2)	(3)	(4)	(5)	(6)
1 2 3 4 5 6 7 8	1923-24 1948-49 1920-21 1926-27 1953-54 1957-58 1929-32	$\begin{array}{r} -3.2 \\ -5.6 \\ -2.9 \\ -1.3 \\ -3.6 \\ -3.2 \\ -3.1 \\ -5.3 \end{array}$	(1) -8.4 (3) -6.8 -5.6 (4)	+5.2 +7.6 +1.8 2 -3.8 -4.2 -26.4	(*) +5.8 (*) (*) -5.9 -10.1 (*)	-61 -74 -483 -72 180 84 74 20	(*) (*) (*) (*) (*) (*) (*)

TABLE 2.—Ranking of eight contractions according to magnitude of change in nonfarm inventory investment relative to change in final purchases, 1920-58 1

Prewar data in 1929 prices. Postwar data in 1954 prices. *Quarterly comparisons are based on quarterly reference cycle dates. Not available.

Source: For nonfarm inventory investment data see table 1. Prewar data for final purchases from Simon Kuznets, "National Product Since 1869," NBER 1946, tables I-15 and I-11. Postwar final purchases computed from GNP and nonfarm inventory investment data, see table 1.

In general, inventory investment has contributed in large measure to recessions both before and since the war, with no well-defined difference apparent in its importance in the two periods. Only in the contraction of 1929-32 was its role a relatively minor one.

## INVENTORY INVESTMENT DURING EXPANSIONS

Table 1 reveals that changes in nonfarm inventory investment, though important, are significantly less so for expansions than for contractions. In all expansions, prewar and postwar, changes in nonfarm inventory investment average 14 percent of changes in national product, whereas the comparable figure for contractions is 66 percent.

Moses Abramovitz noted this difference in the relative importance of inventory investment in the prewar data, and gave as a possible explanation the fact that expansions are longer than contractions.² He observed that, ceteris paribus, the total change in the level of output may be expected to be greater the longer the phase, even though the rate of change may remain the same or decline. On the other hand, inventory investment is simply the rate of change in inventories in a given time period, and it is influenced by the rate of growth (or decline) of output in that period. Consequently it is not to be expected that the change in the levels of inventory investment in the terminal periods will be greater the longer the phase. It follows, therefore, that the share of change in inventory investment relative to the change in national product during a phase may not be expected to be as large for expansions as for the shorter contractions.

This is a cogent argument, and no doubt partially explains the phenomenon.³ It is not the entire explanation however, for the generalization also holds for year-to-year changes within business cycle phases. In the interwar and postwar periods in only 2 years of

^{*} Moses Abramovitz, "Inventories and Business Cycles, with Special Reference to Manufacturers Inventories," New York, National Bureau of Economic Research, 1950, p. 484. * It may also help to explain why the share of inventory investment in the long and severe contraction of

¹⁹²⁹⁻³² was so small,

significant expansion (1927-28 and 1928-29) the change in nonfarm inventory investment constitutes as large a proportion of that in gross national product as in the year of sharpest disinvestment during the subsequent recession (table 3). Possible reasons for this important characteristic will be examined in chapter 8.

Another finding (table 3) is that increases in inventory investment relative to the increase in GNP tend to be larger in the early than in the later stages of expansions. This was true in three of the four prewar expansions which were of sufficient duration (i.e., 2 years or more) to permit observation and in three of the four postwar expansions.⁴

Years of-		Change in GNP during—		Change in nonfarm inventory investment during		Col. 3÷ col. 1 (percent)	Col. 4÷ col. 2 (percent)
Expansion	Contraction	Expansion (1)	Contraction (2)	Expansion (3)	Contraction (4)	(5)	(6)
		Pre	war (billions	of 1929 dolla	rs)		
1919-23         1921-22         1922-23         1925-26         1925-28         1928-29         1928-29         1932-33         1933-34         1935-36         1936-37	1929-21 1923-24 1926-27 1926-27 1930-31 1931-32 1937-38	$\begin{array}{c} +1.4 \\ +4.7 \\ +9.9 \\ +2.6 \\ +0.6 \\ +1.1 \\ +6.5 \\ \\ +4.6 \\ +10.4 \\ +6.5 \\ +6.7 \\ +4.6 \\ +10.4 \\ +6.5 \\ \end{array}$	3.6 +1.5 +1.0 +1.0 7.1 -11.3 -13.6 3.1	$\begin{array}{c} +0.4 \\ +2.2 \\ +2.2 \\ +1.1 \\ +.4 \\ +.6 \\ +1.7 \\$		29 4 22 42 7 54 26 	81 
		Pos	l stwar (billion	s of 1954 doll	ars)		
1946-47 1947-48 1950-51 1951-52 1952-53 1954-55 1956-57 1958-59 1958-60	 1048-49  1953-54 1957-58	$\begin{array}{r} -0.2 \\ +10.8 \\ +25.4 \\ +23.7 \\ +11.7 \\ +15.5 \\ +29.6 \\ +8.2 \\ +7.7 \\ +7.1 \\ +12.4 \end{array}$		$\begin{array}{r} -7.7 \\ +1.6 \\ +9.1 \\ +2.5 \\ -6.8 \\ -1.1 \\ +7.5 \\5 \\ -4.2 \\ +8.0 \\ -1.8 \end{array}$	5.6 	385 15 36 11 -58 -7 25 -6 -55 -29 -14	1, 400 

 
 TABLE 3.—Annual changes in gross national product and nonfarm inventory investment compared, 1919-38, 1945-60

Source: See table 1 for sources.

Finally, it will be noted (table 1) that changes in nonfarm inventory investment played a much larger part in prewar expansions (20 percent of total changes in gross national product) than in the postwar expansions (8 percent). Two possible explanations may be advanced. First, as already noted, postwar expansions have been longer than

⁴ It will be noted in table 6 that for manufacturers' inventory investment the generalization held in each of the four prewar expansions for which comparison was possible and in three of the four postwar expansions.

prewar expansions. This added phase length would tend to reduce the role of changes in inventory investment. Second, peaks in inventory investment have displayed longer leads in postwar than in prewar expansions. Under such conditions, measurement of change in inventory investment from business cycle trough to peak understates the role of inventory investment during the postwar expansions.

In order to take account of differences in length of phase and of the earlier occurrence of peaks in inventory investment during the postwar period, changes in final purchases and inventory investment have been computed for both complete expansions and the first 2 years of expansions (table 4). Computing annual changes separately for the first 2 years of expansion compensates for differences in the length of phase. It also measures the change from the trough to the approximate investment peak in the two postwar expansions which were more than 2 years in duration.

 TABLE 4.—Changes in final purchases and nonfarm inventory investment, eight

 expansions, 1919-60

	Length of	Chang			
Business cycle expansion	expansion (years)	Final purchases	Nonfarm inventory investment	Col. 3÷col. 2 (percent)	
	(1)	(2)	(3)	(4)	
		Billions of	1929 dollars		
1919-20 1921-23 1924-26 1927-29 1932-37	1 2 2 5 (2)	$\begin{array}{c} 0 \\ +11.8 \\ +6.1 \\ +5.6 \\ +22.9 \ (+6.3) \end{array}$	+0.4 +2.4 +1.5 +2.3 +5.7 (+3.0)	20 25 41 24 (48)	
		Billions of	1954 dollars		
1946-48 1949-53 1954-57 1958-60	2 4 (2) 3 (2) 2	+15.2 +72.2 (+35.8) +42.3 (+31.7) +34.0	$\begin{array}{c} -6.1 \\ +3.7 \\ +2.8 \\ +6.2 \end{array} (+7.0)$	-40 5 (32) 7 (22) 18	

Parenthetic entries are for 1st 2 years of expansion. Source: See table 1 for sources.

During the initial 2-year interval the ratio of inventory investment change to change in final purchases was substantially larger than during the full expansions; it appears that the smaller role of changes in inventory investment during the postwar period is associated with the greater phase length and early occurrence of the investment peak. Among the three expansions which were longer than 2 years, the greatest change in inventory investment in the earlier years occurred in the prewar expansion of 1932–35. The next greatest occurred in the 1949–51 period which included the abnormal inventory accumulations of the early months of the Korean war. During the first 2 years of the 1954–57 expansion the relative magnitude of changes in inventory investment appears to have been roughly comparable to that of the 1921–23 and 1924–26 expansions. During the most recent expansion it was somewhat less.

### MANUFACTURERS' INVENTORY INVESTMENT

Changes in manufacturers' inventory investment have been particularly significant during business cycles, especially in contractions. During the recessions of 1948–49, 1953–54, and 1957–58, they accounted for 79, 56, and 25 percent of the change in gross national product from the peak to the trough business cycle quarters (table 5). When the postwar and prewar annual data are compared, it is found that changes in this investment have contributed significantly to business cycles in both periods.

Manufacturers' inventory investment exhibits the same characteristics as were observed for nonfarm investment:

1. The contribution to instability has been greatest during contractions (table 5).

2. During expansions change in investment tends to reach a maximum during the first half of the expansion (table 6).

3. Investment has played a larger role in prewar than in postwar expansions (table 5).

This is not surprising, for movements in the two series have been similar, though not always identical, in both the prewar and postwar periods (charts 1 and 2). During the first two postwar expansions, both nonfarm and manufacturers' inventory investment established two peaks, a major one early in the phase and a lesser one toward the end. In both instances the earlier movement was essentially influenced by war. The sharp rise in investment from 1945 to mid-1946 represented necessary restocking of inventories which had been seriously depleted by the war, and the sharp upward movement from mid-1950 to mid-1951 was strongly influenced by the speculative psychology which prevailed during the early stages of the Korean war. In the 1954-57 expansion there was only a single upward movement in the two series, with peaks occurring during the first half of 1956.

#### INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION 13

	EXPA	NSIONS					
Dustrass syste share	Change	in GNP	Change facturers inve	in manu- s inventory stment	Col. 3 divided by col. 1	Col. divid by co	4 led
Business cycle phase	Annual data	Quarterly data	Annual data	Quarterly data	(percent)	(perc	ent)
	(1)	(2)	(3)	(4)	(5)	(6)	
Prewar (billions of 1929 dollars): 1919–20 1921–23 1924–26 1927–29 1927–29 1932–37	+\$1.4 +14.6 +8.2 +7.6 +28.7	8 8 8 8 8 8	+\$0.4 +1.4 +.6 +1.4 +3.3	99999	29 10 7 18 11	8888 8988	
Total, 1919–37 Postwar (billions of 1954 dollars):	60.5	(1)	7.1	(1)	12	(1)	
1946-48. 1949-53. 1954-57. 1958-60.	+10.6 +76.3 +45.5 +39.5	(1) +\$80.2 +48.9 +48.2	$ \begin{array}{c} (^{1}) \\ +3.0 \\ +2.2 \\ +3.7 \end{array} $	(1) +\$5.6 +2.7 +6.2	( ¹ ) 4 5 9	(4)	7 6 13
Total, 1949-60 ³ Average (prewar and postwar) ³	161. 3	177.3	8.9	14.5	6 9	(1)	8
	CONTRA	OTIONS	- <u></u>	- <u>L</u>	·		
• Prewar (billions of 1929 dollars): 1920-21	$ \begin{array}{c} -\$3.6\\+1.5\\+1.0\\-32.0\\-3.1\end{array} $	9888	$ \begin{array}{c c} -\$1.6 \\ -1.6 \\7 \\ -2.4 \\ -2.7 \end{array} $		44 -107 -70 8 87	() () () () () () () () () () () () () (	
Total, 1920–38. Postwar (hillions of 1954 dollars):	-36.2	(1)	-9.0	(1)	25	(1)	
1948–49 1953–54 1957–58	4 -5.9 -7.3	$-4.3 \\ -11.1 \\ -15.8$	-3.0 -3.8 -1.6	-3.4 -6.2 -4.0	750 64 22		79 56 25
Total, 1948–58 Average (prewar and postwar)	13.6	-31.2	-8.4	-13.6	62 44	( ¹ )	44
	FULL C	YCLES			<b>.</b>		_
Prewar (billions of 1929 dollars): 1919-21 1921-24 1924-27 1927-32 1932-38	\$5. 0 13. 1 7. 2 39. 6 31. 8	SEEE	\$2.0 3.0 1.3 3.8 6.0	89868	40 30 18 10 19	99999	
Total Postwar (billions of 1954 dollars):	96.7		16.1	(1)	17	(4)	
1949-54	82. 2 52. 8	\$91.3 64.7	6. 8 3. 8	\$11.8 6.7	87		13 10
Total Average (prewar and postwar)	135. 0	156.0	10. 6	18.5	8 12	(1)	12

# TABLE 5.—Changes in gross national product and manufacturers' inventory invest-ment, 1919-60

•

¹ Not available. ² Expansion of 1946–48 omitted.

Source: See table 1 for sources of prewar data and postwar GNP data. Deflated manufacturers' inven-tory investment changes based on material from Department of Commerce.

#### 14 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

sof	Change i dur	in GNP ring	Change i facturers investme	n manu- inventory nt during	Col. 3 divided by col. 1	Col. 4 divided by col. 2
Contraction	Expan- sion (1)	Contrac- tion (2)	Expan- sion (3)	Contrac- tion (4)	(percent) (5)	(percent) (6)
		Pre	war (billio	ons of 1929 (	iollars)	
1920-21 1923-24 1926-27 1920-30 1930-31 1931-32 1931-32 1937-38	$\begin{array}{r} +1.4 \\ +4.7 \\ +9.9 \\ +5.6 \\ +5.6 \\ +1.1 \\ +6.5 \\ +1.5 \\ +6.7 \\ +4.6 \\ +10.4 \\ +6.5 \end{array}$		+0.4 +.9 +.54 +.6 +.902 +1.2 +.4 +1.0 +.7		29 19 5 7 54 14 4 18 9 9 10 11	
		Postv	war (billion	ns of 1954 d	ollars)	1
1945-46 1948-49 1953-54 1957-58	$\begin{array}{r} -0.2 \\ +10.8 \\ +25.4 \\ +23.7 \\ +11.7 \\ +15.5 \\ +29.6 \\ +8.2 \\ +7.7 \\ +27.1 \\ +12.4 \end{array}$	31.5 4 	$ \begin{array}{c} +0.7 \\ +4.5 \\ -5.8 \\8 \\ +5.2 \\ +1.0 \\ -4.0 \\ -1.5 \\ \end{array} $		6 	 750  64 
	s of Contraction 1920-21	s of Change during the during the state of	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

TABLE 6.—Annual	changes in gross	national	product	and	manufacturers'	inventory
	investmen	t compare	d. 1919-	38.	1945-60	-

Source: See table 5 for sources.

The similarity in patterns of movement in the two inventory investment series is largely due to the fact that manufacturers' inventory investment is the major component of nonfarm inventory investment. Table 7 shows that manufacturers' stocks have comprised between 46 and 56 percent of nonfarm inventories in the prewar and postwar periods. Moreover, in six of the eight cycles, movements in manufacturers' inventory investment have been larger than would be expected on the basis of the proportion of total stocks held (in one case, they were equal). During the single cycle in which they were smaller, the 1924-27 episode, atypical countermovements in durable-goods inventory investment occurred, serving to dilute the intensity of movements in total manufacturers' inventory investment (see table 10, ch. 3).

terific Provide Lange and the second s									
	acturers' ories as	Chan	ge in ma i	anufa nves	cture tmen	ers' in t	ventory	Change in gross	
Cycle	percent o inven	f nonfarm tories ¹	As perc farm in	As percent of change in non- farm inventory investment *					product as percent of average
	Abramo- vitz- Kuznets	Depart- ment of Com- merce	Expan- sion	Cont	trac- n	Tc cy	otal cle	turing inven- tories ¹ contrac- tion	national product contrac- tion
Prewar: 1919-21	46 48 49 50 53 50		100 58 40 61 58 58 3 91 (4) 81 (59 79 (68	55 50 54 45 75 55 55 54 119 52	(40) (91) (71)	61 54 46 50 65 56 8 73 99 64	(1) (72) (70)	16 14 6 17 19 12 8 8 8	5.4 2.0 1.2 37.0 3.9 12.0 .01 1.7
1958-60 Weighted average \$		55	60 (69 70 (64	) 71	(65)	 83	(70)		1.8
				1					i

TABLE 7.—Analysis of relative size of cyclical changes in manufacturers' inventory investment, 1919-60

 Cycle averages of inventories were used in these comparisons.
 Figures in parentheses are based on quarterly data.
 Measured from 1945 to 1948 in order to compare trough and peak inventory investment. Not available,

Averages exclude 1946-48 expansion; include 3 expansions, 3 contractions, 2 full cycles.

Source: Prewar figures computed from data in Abramovitz, Inventories and Business Cycles, pp. 476, 8. Postwar figures from material provided by Department of Commerce. All data are deflated. 568

The continuing importance of postwar movements in nonfarm and manufacturers' inventory investment is especially interesting in view of the fact that stocks are smaller relative to the volume of sales and output than they were before the war. For the years 1919 to 1939 Abramovitz constructed indexes of year-end ratios of deflated manufacturing stocks to output.⁵ Similar ratios have been computed here for the years 1947 through 1954. The 1947-54 average compares as follows with decade averages for the twenties and the thirties: 1000 00

1920-29	103
1020 20	110
1930-39	119
1947-54	77
1011 0111111111111111111111111111111111	

Comparable data are not available for distributive and other firms but the fact that such stocks have also been reduced sharply relative to sales is attested to by their decreased share of total nonfarm inventories (table 7). It is concluded therefore, that although manufacturing and other nonfarm stocks are relatively smaller today than before the war, their movements have played an equivalent role in postwar business cycle contractions.

It should be kept in mind, however, that postwar recessions have been comparatively mild. Changes in inventory investment have been large relative to changes in gross national product (tables 1 and 5); but movements in national product during contractions have not been large relative to average levels of national product, and postwar changes in manufacturers' inventory investment have not been large in relation to the average size of stocks (table 7).

Abramovitz, "Inventories and Business Cycles," p. 570. 76626-61-pt. I----3

#### CHAPTER 3

## MANUFACTURERS' INVENTORIES AND INVENTORY INVESTMENT

Analysis of the Department of Commerce total stock series in the present chapter yields several important findings. Both total manufacturers' stocks and inventory investment are shown to have turned somewhat earlier in relation to business cycle turns since 1946 than Abramovitz indicated was characteristic of the prewar period. Nevertheless, movements in quarter-to-quarter change in inventories (i.e., inventory investment) have well-defined cyclical patterns in most of the series and show a high degree of conformity to both industry sales cycles and business cycles. Durable-goods series were cyclically more sensitive than those of nondurable goods, showing a higher conformity to business cycles and moving with greater amplitude. Stocks in durable-goods industries have been relatively larger since the war, and their increased importance has altered the composition of total stocks.

#### TIMING COMPARISONS

During the 1946-58 period, total stocks lagged behind reference turns by 1 to 8 months (table 8 and chart 3), in contrast to Abramovitz' estimate of a 6-to-12-month prewar lag.¹ Total investment timing varied from roughly coincident to a lead of 14 months (table 9 and chart 4), compared to Abramovitz' coincident timing estimate.

17

¹ It will be noted that timing of the two most recent turns in table 8 is based upon book value data.

#### INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION 18

#### TABLE 8.—Timing and conformity of manufacturers' total inventory at reference and activity turns; comprehensive series

·	Le	ad () or l	ag (+) in 1	nonths—B	usiness cyc	eles
Industry	Peak Novem- ber 1948	Trough October 1949	Peak July 1953	Trough August 1954	Peak July 1957	Trough April 1958
		T	iming at re	ference tur	ns	
Total, manufacturing Durable goods industries, total Nondurable goods industries, total	+4 -14 (3)	(*) +6 +4 (*)	$^{+2}_{+2}_{+2}$	+1 +9 +1	+2 +3 +3 +1	+8 +8 +5
		Timi	ng at activ	ity (sales)	turns	
Total, manufacturing Durable goods industries, total Nondurable goods industries, total	+6 -15 ( ³ )	(*) (*)	+2 +2 (ª)	(*) ⁻¹	+8 +9 +7	+9 +8 +6
B. SUMMARY OF	TIMIN	3 (16 IND	USTRY &	ERIES) 4		
			At busir	iess cycle	At busir	less cycle

A. TIMING MEASURES 1

	At busin	ess cycle aks	At business cycle troughs		
	Number	Percent	Number	Percent	
Leads, more than 3 months Rough coincidences Leads, 3 months or less. Coincidences. Lags, 3 months or less. Lags, more than 3 months.	6 12 (8) (0) (4) 9	22 45 (30) (0) (15) 32	2 10 (3) (0) (7) 15	7 37 (11) (0) (26) 56	

#### C. SUMMARY OF CONFORMITY

	16 industry series		7 dura ind	able goods lustries	9 nondurable goods industries		
	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean	
Comparisons with reference turns: Number of comparisons Matching inventory turns Percentage of matching turns Comparisons with output turns: Number of comparisons Matching inventory turns Percentage of matching turns	96 65 68 75 54 72	64 54 84 53 48 91	42 30 71 38 30 79	28 28 100 28 28 28 100	54 35 65 37 24 65	36 26 72 25 20 80	

Inventory series have been deflated to 1956; 1957-58 turns based on undeflated data. Korean cycle reference turns are not shown, since there were no corresponding inventory turns.
No corresponding inventory turn.
Inventory turn occurs, but no sales turn.
Analysis includes only deflated 1943-58 data.

Source: Based on material from Department of Commerce.



#### CHART 3 INVENTORIES AND SALES: TOTAL MANUFACTURING, DURABLE- AND NONDURABLE-GOODS INDUSTRIES, 1945-61

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated inventory cycles; circles, of undeflated cycles. All sales data are undeflated.

Source: Department of Commerce. Data deflated by the author.

It should be noted, however, that the prewar estimates were based on year-end data and hence are necessarily less precise. Abramovitz estimated from the annual data the range within which inventory and inventory investment turns would lead or lag behind reference turns on a monthly basis. Using his method, I have estimated the range of timing for total manufacturers' deflated inventories from

#### 20 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

postwar annual data. These estimates may be compared with actual timing as follows:

Reference turn	Timing	
	Estimated range	Actual
Peak November 1948 Trough October 1949 Peak July 1953 Trough August 1954	+3 to +12 0 to +9. +3 to +12 +3 to +12	+4 +6 +2 +1

The consensus of Abramovitz' findings for the prewar period was for longer lags than these. Two of his turns showed a minimum lag of 12 months. I agree with Abramovitz that lags were typically somewhat longer before the war, but it seems likely that the range was greater than he indicated. Evidence of this is to be found in Department of Commerce estimates of the book value of manufacturers total stocks (quarterly, 1926–38, monthly thereafter) and in National Industrial Conference Board estimates (monthly, 1929–44). Prewar timing measures for these series are as follows:

	Timing	at peak	Timing at trough		
Date of reference turn	Department of Commerce	NICB	Department of Commerce	NICB	
October 1926	-3½		-716		
August 1929 March 1933 Mar 1937	+71/2	+5½	+3½	+21/2	
June 1938	T*72		+121/2	+121/2	

During the postwar period a tendency for the inventory series to turn earlier at peaks than troughs, as compared with both reference and activity series may be observed (table 8). This finding is less well established in the comprehensive series than in the individual industry series and is offered only as a tentative observation.²

Table 8 shows the timing of movements in the comprehensive series. In two sets of comparisons (the turns which were related to the recessions of 1948-49 and 1957-58) stocks show shorter lags at peaks than at troughs; in the remaining comparison they do not (see also chart 3). The departure from the observed tendency is not so damaging as might at first appear, for the trough in stocks has been marked at the technically lowest point of an almost flat-bottomed recession phase which extends 10 months after the date of the trough selected. In the industry data, the evidence is more impressive.

¹ In this chapter and the chapters which follow, inventory and inventory investment turns are compared both with turns in the comparable sales or output series and with business cycle reference turns. This is a necessary, if somewhat repetitious, procedure; inventory behavior in a given industry or group of industries must be presumed to be functionally related to economic activity as represented by sales or output (typically, the two move together closely). It is essential, therefore to relate movements in stocks or investment to activity. At the same time, in order to generalize regarding behavior of inventories during business cycles and to compare the timing of one inventory series with another, it is essential to make direct timing comparson with reference turns. In addition to the NBER business cycle reference turns, use has been made of the two subcycles and the criteria involved in establishing subcycle chronology, see Ruth P. Mack, "Notes on Subcycles in Theory and Practice," American Economic Review, May 1957, pp. 161-174.

#### INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION 21

The summary of timing comparisons for the 16 series (table 8) shows that 47 percent of the peaks in the inventory series lagged behind reference peaks, whereas 82 percent of the inventory troughs lagged behind reference troughs.

The tendency for shorter lags at peaks holds for both of the two cycles for which industry data were studied (1948–49 and 1953–54). During the first postwar business cycle, 54 percent of inventory peaks lagged behind the reference peak, and 73 percent of inventory troughs lagged behind the reference trough. For the second cycle the figures were 42 and 92 percent, respectively.

We shall see, however, that the generalization does not extend to inventory investment which turns relatively early at activity and reference peaks and troughs. Some instances of earlier timing at peaks than at troughs may be noted, but there is no consistent pattern.

The tendency of quarterly inventory investment series (quarter-toquarter change in inventories) to move in well-conforming cyclical patterns was found in most of the individual industry series for total stocks and, as we shall see in the following chapters, by stage of fabrication as well.

The relatively high degree of conformity of the total inventory investment series to business and output cycles is clearly evident from the summary of conformity measures given in table 9. Industry investment turns could be consistently related with corresponding reference turns in 86 percent of possible comparisons, and with output turns in 87 percent.³ This conformity is of roughly the same magnitude as that noted for total stocks series in table 8 (68 and 72 percent, respectively).

³ A principal method of measurement used in this study was to record the number of turns in inventory or investment series that could be "matched" with reference or activity turns, using NBER rules for matching. Peaks in a series are said to match reference (activity) peaks when there is no intervening trough in either the reference (activity) cycle or in the series in question.

			Lead (—	) or lag (	+) in m	onths		
	Busine	ss cycle	Korea	n cycle		Busines	s cycles	
Industry	Peak (refer- ence, Novem- ber 1948)	Trough (refer- ence, October 1949)	Peak (Febru- ary 1951)	Trough (June 1952)	Peak (July 1953)	Trough (Au- gust 1954)	Peak (July 1957)	Trough (April 1958)
			Timin	ıg <b>"</b> at refer	ence tu	ns		
Total manufacturing Durable-goods industries, total	$-6 \\ -6$	$-2 \\ -2 \\ -2$	$^{+3}_{+6}$	-1 -1	$-2 \\ -2$	0 -3	$-14 \\ -20$	+1
total	-6	+7	+3	(2)	(2)	-9	-14	+4
			Timing a	at activity	7 (sales)	turns	**	
Total manufacturing Durable-goods industries, total	4 7	-4 -2	(3) (3)	(3) (3)	$-2 \\ -2$	$-2 \\ -5$	-8 -14	+2 +1
total	-1	+4	+4	(2)	(4)	(8)	-8	+5

# TABLE 9.—Timing and conformity of manufacturers' total inventory investment at reference and activity turns; comprehensive series

A. TIMING MEASURES 1

B. SUMMARY OF TIMING (16 INDUSTRY SERIES) 6

	At busin pea	less cycle aks	At business cycle troughs	
	Number	Percent	Number	Percent
Leads, more than 3 months Rough coincidences Leads, 3 months or less. Coincidences Lags, 3 months or less.	$12 \\ 11 \\ (8) \\ (2) \\ (1)$		8 15 (11) (0) (4)	30 56 (41) (0) (15)

#### C. SUMMARY OF CONFORMITY .

	16 industry series		7 durable-goods industries		9 nondurable-goods industries	
	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean
Comparisons with reference						
Number of comparisons	96	64	42	28	54	36
turns	83	55	40	28	43	27
Comparisons with output	86	86	95	100	80	75
Number of comparisons Matching investment	75	53	38	28	37	25
turns	65	49	38	28	27	21
turns	87	92	100	100	73	84

Inventory series have been deflated to 1956; 1957-58 turns based on undeflated data.
Sales turn occurs but no well defined corresponding inventory investment turn.
No sales turn occurs.
No sales or inventory turn occurs.
Inventory turn occurs, but no sales turn.
Analysis includes only deflated 1948-56 data.

Source: Based on material from Department of Commerce.

CHART 4



INVENTORY INVESTMENT: TOTAL MANUFACTURING, DURABLE- AND NONDURABLE-Goods Industries, 1945-61

#### CYCLICAL SENSITIVITY OF DURABLE GOODS

Conformity measures point clearly to the greater cyclical sensitivity of durables. Turns in the durable-goods inventory series (table 8) could be matched with reference turns (except Korean) in 100 percent of comparisons; among the nondurables, only 72 percent. Similar figures for inventory investment (table 9) were 100 percent for durables and 75 percent for nondurables.

Amplitudes of fluctuations in inventory investment may be compared for durables and nondurables by means of a simple tabulation of the change from peak to trough and from trough to peak in investment cycles. Since total stocks of durable-goods industries on the average are not greatly in excess of nondurable (about 1.2 times as large) the results may be compared directly. The following tabulation (in billions of 1954 dollars at annual rates) shows that the change in level of inventory investment was far greater for durables than for

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated cycles; circles, of undeflated cycles. Source: Department of Commerce. Data deflated by the author.
nondurables in all but one of the postwar phases for which deflated data were available.

			Du	rables					Nondurables		
P-T	Ι	49 to	III	49	-2.8	II	48 to	II	49	-2.9	9
T–P	III	49 to	$\mathbf{III}$	51	+10.3	II	49 to	II	51	+4.8	3
P-T	III	51 to	$\mathbf{H}$	52	-9.5	II	51 to	$\mathbf{III}$	54	-4.4	4
T-P	II	52 to	$\mathbf{H}$	53	+3.9	III	54 to	III	56	+2.0	6
P-T	II	53 to	II	54	-6.7	$\mathbf{III}$	56 to	II	58	-2.9	9
T-P	II	54 to	I	56	+7.9	II	58 to	IV	59	+2.0	6
P-T	Ι	56 to	II	58	-7.3						
T-P	II	58 to	I	60	-10.4						

This was not the result of one or two dominating series, but a highly representative performance. During the 1949 recession six of the seven durables inventory series showed declines of 10 percent or more from their peaks, whereas only five of the nine nondurables declined by this amount. During the 1953-54 recession five of the seven durables series declined 10 percent or more, but only one of the nondurables did so.⁴

Another characteristic of durables inventory movements is that the series move together; this makes for little muting of the impact through dissimilar and counteracting changes. This characteristic is clearly in evidence when diffusion indexes of percentage of series increasing were constructed for the durables and nondurables series. Among the durables the tendency for stocks to move together was indicated by very high scores (indicating that most series were rising) or very low scores (indicating that few series were rising), whereas among the nondurables the tendency for stocks to move in different directions was indicated by scores close to 50 percent. During the period extending from second quarter 1948 to second quarter 1955, scores were above 80 percent or below 20 percent more than half of the time for the durables and less than a sixth of the time for the nondurables.

There appear to be two types of causation for these well-established cyclical tendencies among the durables. In the first, the cyclical behavior appears to derive from greater fluctuations and more closely coordinated changes in activity among the several durables industries. In the second, the explanation lies in the composition of stocks; i.e., the inventories held at different stages of fabrication are of the types which are more responsive to cyclical forces.

The first point is readily established by applying the same test to output series that was used to show amplitude of movements in inventories. During the 1949 contraction, all seven durables output series declined by more than 10 percent of their cycle bases, whereas only three of the nine nondurables series declined by such amounts. During the 1953-54 recession, six of the seven durable output series declined by more than 10 percent, but only four of the nine nondurables declined by as much as this amount. In addition, diffusion indexes for output of durables showed a greater swing than nondurables during these two periods, indicating a closer intercorrelation in the movements of durables.⁵

The second type of causation will become apparent as the behavior of stocks at each stage of fabrication is analyzed, but the general argument may be stated here. Durables inventories are composed in a

⁴ The declines were measured as percentages of the average level of stocks during each cycle. ⁴ G. H. Moore, Business Cycle Indicators, Princeton for NBER, 1961, vol. I, pp. 233-234.

larger degree than nondurables of the types of stocks which move sensitively when manufacturing activity increases or decreases. Procurement of durable purchased stocks is largely demand oriented; it is not limited by supply conditions, as may be the case with certain nondurable agricultural raw materials. Durable goods-in-process stocks are held between stages (rather than within stages) to a much greater extent. It is these between-stage stocks which may be varied to the greatest extent with changes in demand conditions (chapter 6). Finally, durable goods are typically manufactured to order rather than to stock; this tends to cause finished stocks to respond more sensitively to cyclical forces.

By making use of Abramovitz' data, it is possible to determine the relative size of durable-goods manufacturers' stocks before the war and to learn whether they displayed as high a degree of responsiveness in that period. Table 10 shows the average proportion of total stocks made up by durables for each cycle from 1919 to 1958. In addition, it shows the proportion of the change in manufacturers' inventory investment in each phase which was caused by changes in durables investment. It can be concluded that durables stocks have constituted a larger share of total stocks in recent years, and have contributed in a greater measure to economic instability. The characteristic of high cyclical responsiveness, however, was well established in the earlier period.

# Composition of Manufacturers' Stocks

We have seen that total manufacturers' stocks are smaller relative to output than they were before the war. Census data and Department of Commerce statistics indicate, in addition, that their composition has been significantly altered. Abramovitz estimated the prewar composition of stocks as 40 percent purchased materials, 20 percent goods in process, and 40 percent finished goods. The Annual Survey of Manufactures shows the 1952–53 proportion to be approximately 38 percent purchased materials, 29 percent goods in process, and 33 percent finished goods.

The proportions differ considerably for the durables and the nondurables groups. In the period 1952-54 the composition of durable and nondurable manufacturers' stocks was approximately: ⁶

	Purchased	Goods in	Finished
	materials	process	goods
Durables	30	40	30
Nondurables	44	14	42
Total	36	28	36

The major difference in composition of durables and nondurables is, of course, the large stock of goods in process held by the durables group. These in-process inventories account for the large stocks-tosales ratios of the durables group (approximately 2.1 to 1 compared to 1.6 to 1 for nondurables). The stocks-to-sales ratios for purchased

⁶ The composition of total stocks given here was obtained from Department of Commerce data and is slightly different from that in the Annual Survey of Manufactures. The difference is due principally to the fact that the former are reported on a firm basis and include some finished stocks held by company distributing organizations, whereas census data are reported by individual manufacturing plants.

materials and for finished goods do not differ significantly between the durables and nondurables groups.⁷

The change in composition of manufacturers' stocks from prewar to postwar appears to be explained largely by the shift toward durables. Using the prewar ratios of nondurables to durables inventories (table 10) which is about 2 to 1, the prewar composition (if composition within durables and nondurables had been the same as postwar) would be 39 percent purchased materials, 23 percent goods in process, and 37 percent finished goods. These percentages are quite close to the actual prewar figures (40-20-20) as estimated by Abramovitz.

The significance of the behavior of stocks at each stage of fabrication is determined by the size of each category and by the amplitude and cyclical pattern of its movement. Judged on the basis of size it is clear that each category may be considered to be of substantial importance. Analysis of cyclical conformity, timing, and amplitude will be presented in the chapters which follow.

TABLE 10.-Analysis of percentage of change in total manufacturers' inventory investment accounted for by inventory investment of durables manufacturers, 1919-61

Cycle	Durables inventory as percentage of total inventory	Percentage of change in manu- facturers' inventory invest- ment accounted for by dura- bles inventory investment ¹			
	(cycle average)	Expansion	Contraction		
Prewar: 1919-20-21	38 37 43 36 35 52 54 54 ( ³ )	76 83 (*) 28 39 97 (*) 97 (84) 95 (93) 78 (74)	46 40 (3) 52 40 50 (68) 92 (82) 112 (88) (4)		

¹Figures in parentheses are based on quarterly data. ²Durables inventory investment out of phase with total manufacturers' inventory investment and reference cycle. * Not available.

Source: Prewar figures computed from deflated data in Abramovitz, "Inventories and Business Oycles." pp. 564-565. Postwar figures are in 1954 dollars; data from Department of Commerce.

⁷ The changed composition of goods in process is treated in detail in chapter 6.

### CHAPTER 4

# PURCHASED-MATERIALS INVENTORIES AND INVENTORY INVESTMENT

Abramovitz began his investigation of purchased-materials behavior by adopting a preliminary hypothesis that "stocks of raw materials generally grow and decline with manufacturing activity." 1

Although he recognized that movements in prices may play a role, this hypothesis provided the principal basis for his inquiry. Purchased materials were seen to move in response to changing levels of output, but with delays occasioned principally by lack of immediate availability of the materials in question. The analysis based upon this hypothesis consisted principally of estimating the length of the average period of delay in securing materials. This estimate was based on a special study made during the preparation of the 1929 Census of Manufacturers, which classified materials used by manufacturers according to their origins.² On the basis of this study Abramovitz concluded that "most raw materials stocks held by manufacturers are likely to vary positively with a short lag" and that total manufacturers' stocks of purchased materials might be expected to conform closely to business cycles, "lagging 3 months or somewhat more at turning points." ³ Moreover, movements in purchased-materials inventory investment were expected to be a lagged reflection of movements in the rate of change in output. Evidence at hand did not permit him to estimate how long the lag might be. He simply noted that "a lag of investment in raw materials behind the rate of change in output of even 6 months would still leave this class of investment leading business cycle turns by This, of course, is presumably subject to considerseveral months. able irregularity, especially near business peaks, corresponding to the irregularity in the timing of turns in the rate of change in output." 4 In an analysis of the behavior of 10 raw-materials inventory series, however, he found support for the tentative conclusions reached in his estimate of typical timing based on availability of materials.⁵

No study such as that used by Abramovitz is available for the postwar period, and there is no basis for an accurate comparison of

 ¹ Abramovitz, "Inventories and Business Cycles With Special Reference to Manufacturers Inventories," New York, NBER 1950, p. 178. His definition of raw materials as "goods purchased by manufacturers but not yet manipulated by their owners," is used in this study as a description of "purchased materials," It should be noted that purchased materials defined in this way include not only raw or semiprocessed materials, but also tuly fabricated component parts, such as carburetors purchased by automobile mann-facturers or small motors purchased by the producers of electric fans.
 ³ Abramovitz divided these stocks into three groups accounting to source. The first group included to be available from suppliers on relatively short notice accounted for approximately two-thirds of all purchased materials, used by manufacturers in 1920. A second group, comprising about 10 percent of the raw materials purchased by manufacturers, was composed of imports. These stocks are not easily adjusted to changes in requirements. The third group, accounting for approximately a quarter of manufacturers' raw materials, included products (principally agricultural) whose availability is subject to short-term movements that are independent of current demand. Abramovit anotit here and typically held in large quantities by middlemen (e.g., cotton stored by cotton merchants). Ibid., pp. 184-187, 314.
 ⁴ Ibid., p. 487.
 ⁴ Ibid., p. 487.

postwar and interwar composition of stocks. Fortunately, however, there is recent direct evidence relating to the availability of purchased materials. The National Association of Purchasing Agents has published monthly, since April 1955, statistics on purchasing leadtimes as reported by their members. Each respondent reports monthly the basis of its purchasing policy for principal purchased materials within one of the following categories: Hand-to-mouth buying, 30-day requirements, 60-day requirements, 90-day requirements, 120-day requirements, or more.

These NAPA statistics indicate that, in the main, purchased materials are readily available, but to a greater or lesser degree depending on the stage of the business cycle. This is apparent in the following data which show the percentage of purchasing agents reporting leadtimes of 60 days or less for production (purchased) materials in business cycle peak and trough months (see also chart 16):

	Percent
Peak, July 1957 1	75
Trough, April 1958	91
Peak. May 1960	79

¹ June data were used because no July data were published.

Clearly, Abramovitz' assumption that most purchased materials are available within a relatively short span of time is valid for the postwar period. But has this condition led to the type of inventory and investment behavior which he concluded would be the logical outcome of such ready availability? In the sections that follow we shall examine the behavior of purchased-materials inventory and inventory investment, note certain characteristics not envisioned by Abramovitz and inquire into the reasons for the observed behavior.

### BEHAVIOR OF INVENTORIES

# TIMING AND CONFORMITY OF INVENTORIES: COMPARISONS WITH SALES TURNS

The timing of purchased materials to sales turns, as shown in table 11, varies widely from turn to turn in the individual industry series and among the series at each peak and trough. Nevertheless, the conformity to sales turns is generally high and approximately the same for durables and nondurables industries (about 80 percent of sales turns can be matched by inventory turns in each category).

### TABLE 11.-Timing and conformity of manufacturers' purchased-materials inventories at sales turns

	Lead (-) or lag (+) in months, in zones associated with reference turns							
Industry	1948 peak	1949 trough	1953 peak	1954 trough	1957 peak	1958 trough		
Total manufacturing         Durable-goods industries, total         Nondurable-goods industries, total         Machinery (including electrical)         Transportation equipment (including motor vehicles)         Stone, clay, and glass         Pool         Paper         Chemical         Petroleum and coal         Rubber	$-1 \\ -17 \\ +4 \\ -6 \\ -17 \\ -24 \\ -2 \\ -14 \\ +2 \\ +6 \\ -13 \\ +7 \\ +7 \\ +7 \\ +7 \\ +7 \\ +7 \\ +7 \\ +$	$\begin{array}{r} +4 \\ +4 \\ +7 \\ +7 \\ +7 \\ +7 \\ +2 \\ +1 \\ +1 \\ +1 \\ +1 \\ +1 \\ +1 \\ +7 \\ +7$	$())^{+1} ()^{+1} ()^{-1} + 1 \\ ()^{-1} + 1 \\ + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-1} + 1 \\ ()^{-$	(*) + 6 + 8 + 8 + 10 + 10 + 5 + 12 + 13 + 7 + 7 + 15 + 12 + 12 + 12 + 12 + 12 + 12 + 12	+10 +2 +7	+6 +4 +11		

#### A. TIMING MEASURES !

#### B. SUMMARY OF TIMING (9 INDUSTRY SERIES) 1

	At sales peaks	At sales troughs
Leads, more than 3 months Rough coincidences. Leads, 3 months or less Coincidences. Lags, 3 months or less. Lags, more than 3 months.	5 6 (2) (1) (3) 2	0 3 (0) (0) (3) 13
1		

#### C. SUMMARY OF CONFORMITY

	9 indust	ry series	4 durab indu	le-goods stries	5 nondurable-goods industries	
	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean
Number of comparisons	44 33 75	34 20 88	20 16 80	16 15 94	22 17 77	18 15 83

Inventory series have been deflated to 1956. 1957-58 turns based on undeflated data. Timing of inventory turns for "Korean" cycle is not shown since there were only 3 inventory turns that could be matched with sales turns related to this episode. See table 12 for timing comparisons of inventory turns with "Korean" reference turns.
 No sales or inventory turns occurred. Most recent inventory turn to occur was peak in M ay 1952 associated with Korean war cycle.
 Inventory turn occurs, but no sales turn.
 Sales turn occurs but no inventory turn, proceeding inventory peak was related to Korean war cycle.

A sales turn occurs but no inventory turn; preceding inventory peak was related to Korean war cycle.
 Inventory peaks occurred in series as follows: Nondurable-goods industries, November 1951; stone, clay, glass, March 1952; food and beverages, May 1951; chemical, June 1952; petroleum and coal, June 1952.

Source: Based on material from Department of Commerce.

In spite of diversity, there is a well-established characteristic in the timing of both the comprehensive and industry purchased-materials series during the 1948-54 period: turns in these series occur earlier relative to sales at peaks than they do at succeeding troughs. This is true for 24 of the 26 turns in the industry series that could be matched with a specific sales cycle, and for all turns in the deflated comprehensive series.

In addition, turns frequently lead at peaks, on some occasions by very substantial intervals. Lags at troughs are sometimes quite long, well over a year for certain of the industry series. Abramovitz made no provision for the occurrence of substantial leads in stocks relative to sales or output or for substantial lags at troughs. Purchasedmaterials stocks were presumed to be so clearly associated with output that only lags due to delays in procurement could occur.

Nevertheless, these timing characteristics find support in certain of his prewar data. The timing for three purchased materials which he believed behaved typically is as follows: ⁶

	At peaks in activity				At troughs in activity			
	Number of—			Aver-	Number of			Aver-
	Leads	Coin- ci- dences	Lags	lead (-) or lag (+) in months	d or (+) Leads Coin- ci- dences Lags	lead (-) or lag (+) in months		
Raw cotton at mills Raw silk at mills Raw cattle hides at tanners	3 1 2	0 0 1	5 3 2	+1.5 +0.5 +1.2	0 1 1	1 0 0	6 2 3	+5.1 +1.0 +6.2
3 commodities	6	1	10		2	1	11	

When timing comparisons are made at the 1957 peak and the 1958 trough the results are more ambiguous. The durables and nondurables series show shorter lags at peaks than at troughs, but the total manufacturing series shows a 10-month lag at the peak in the sales series and a 6-month lag at its subsequent trough. This paradoxical difference in the timing patterns of the aggregate and its two components is due, not to differences in the timing of durables and nondurables sales turns (which are virtually coincident), but to the fact that the two component inventory series do not decline significantly for a number of months following their peaks; in November 1957 they actually show simultaneous increases sufficient to cause the aggregate to register a peak in that month.

It is difficult to draw conclusions regarding this recent episode. The fact that the data have not been deflated and that prices did rise slightly at this time makes it not unlikely that appropriate deflation would have caused the series to show earlier turns at peaks. A second source of difficulty is the unusual behavior of manufacturing activity during this cycle. Output and sales reached peaks approximately half a year prior to the peak in general business activity, but showed no sharp decline until the business cycle turn. Under such conditions there was not, perhaps, the usual incentive to reduce stocks.

conditions there was not, perhaps, the usual incentive to reduce stocks. Nevertheless, on the basis of all the evidence, it seems justifiable to regard as significant the tendency for purchased materials to turn earlier at sales peaks than at troughs. Moreover, it is apparent that purchased materials may turn coincidentally with sales turns or even earlier and that lags at troughs may be of considerable duration.

[•] As indicated above, Abramovitz worked with a limited sample of 10 commodity series in investigating the behavior of purchased-materials stocks. Of these 10 series, only 8 were available in monthly or quarterly form; only the 3 shown appeared "to be supplied to manufacturers under conditions that afford the degree of responsiveness of supply to changes in demand that characterizes most raw materials purchased by manufacturers * * *," ibid., pp. 190, 393.

The 1957-58 experience does not invalidate these findings; it merely indicates that purchased materials inventory turns may lag at sales peaks as well as at troughs.

### TIMING AND CONFORMITY OF INVENTORIES: COMPARISONS WITH REFERENCE TURNS

Timing and conformity of purchased-materials turns to business cycle turns (table 12 and chart 5) reveal tendencies similar to those pointed out above. Purchased-materials series typically turn earlier at business cycle peaks than at troughs. Among the industry series this earlier timing is manifested in leads or roughly coincident timing at peaks, and in lags at troughs. Thirteen peaks could be matched with business cycle reference peaks. Five of these turns led by more than 3 months, seven were roughly coincident, and only one lagged more than 3 months. (Rubber products lagged behind the reference peak of 1948 by 8 months.) Some of the leads were quite long, as much as 18 months. Eighteen troughs could be matched with business cycle troughs, of which 16 lagged more than 3 months and 2 were roughly coincident. Lags ranged from 3 to 16 months.

Using business cycle turns as a basis for comparison, the conformity rating of the durables is higher than that of nondurables. Purchasedmaterials stocks of durables show a somewhat higher conformity to sales turns associated with business cycle turns than the nondurable. In addition, the durable sales turns, themselves, conform better to business cycles; i.e., there are fewer "skipped" phases. The net result is that the durables purchased materials show a slightly higher conformity to business cycles, the four durables series matching business cycle turns in 15 of 16 possible comparisons, whereas nondurables matched in only 16 of 20.

## TABLE 12.—Timing and conformity of manufacturers' purchased-materials inventories at reference turns

	Lead (-) or lag (+) in months										
Industry	Business cycle		Korean cycle		Business cycles						
	Peak Novem- ber1948	Trough Octo- ber1949	Peak Febru- ary1951	Trough June 1952	Peak July 1953	Trough August 1954	Peak July 1957	Trough April 1958			
Total manufacturing Durable-goods industries, total. Nondurable-goods industries, total. Primary metals. Machinery (including electrical). Transportation equipment (includ- ing motor vehicles). Stone, clay, and glass. Food and beverages. Paper. Chemical. Petroleum and coal. Rubber.	$ \begin{array}{r} -3 \\ -16 \\ -1 \\ -5 \\ -16 \\ -18 \\ -3 \\ -16 \\ 0 \\ +1 \\ -17 \\ +8 \end{array} $	$ \begin{array}{r} +6 \\ +4 \\ +8 \\ +7 \\ +5 \\ +3 \\ +4 \\ +9 \\ +10 \\ +6 \\ +16 \\ +10 \\ \end{array} $		+4+2 (3) + 3 + 1 (3) (3) (3) (3) (3) (3) (3) (3) (3) (3)	$ \begin{array}{c} +1 \\ +1 \\ -1 \\ 0 \\ +1 \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) \\ (3) $	+ 8 + 10 + 7 + 10 + 7 + 10 + 7 + 14 + 6 + 9 + 14 + 13 + 14 + 13 + 13 + 14 + 13 + 13	+4 -4 +1	+5 +4 +10			

#### A. TIMING MEASURES 1

### B. SUMMARY OF TIMING (9 INDUSTRY SERIES)

	At business cycle peaks	At business cycle troughs
Leads, more than 3 months.	5	0
Rough coincidences.	7	2
Leads, 3 months or less.	(2)	(0)
Coincidences.	(3)	(0)
Lags, 3 months or less.	(2)	(2)
Lags, more than 3 months.	1	16

#### C. SUMMARY OF CONFORMITY

	9 indust	ry series	y series 4 durabl		5 nondurable- goods industries	
	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean
Number of comparisons Matching inventory investment turns Percentage of matching turns	54 40 74	36 31 86	24 20 83	16 15 94	30 20 67	20 16 80

¹ Inventory series have been deflated to 1956. 1957–58 turns based on undeflated data. ³ No matching inventory turn.

Source: Based on material from Department of Commerce.



CHART 5 Purchased-Materials Inventories and Sales: Total Manufacturing, Durable- and Nondurable-Goods Industries, 1946–58

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated inventory cycles; circles, of undeflated cycles. All sales data are undeflated.

Source: Department of Commerce. Data deflated by the author.

When measured against Korean reference dates, turns in the comprehensive as well as the industry series show very long lags at peaks, quite in contrast to the leading tendency demonstrated at business cycle peaks. It should be noted that the Korean cycle was unique in that manufacturers reported exceptionally high levels of order backlogs at the beginning of 1951 and that there was a marked difference in the behavior of durables and nondurables. These matters will be discussed in a subsequent section.

## BEHAVIOR OF INVENTORY INVESTMENT

The total manufacturers' purchased-materials investment series shows four cyclical movements from 1945 to 1954, two occurring in the course of each business cycle. (See Chart 6.) In each pair of investment cycles the first is the major movement. The second involves a relatively small increase prior to the sharp declines in both inventory investment and inventories proper which conform closely to the two business cycle recessions.

Differences between the behavior of durables and nondurables occurred during these two secondary movements. For the durables series, there was no significant resurgence of inventory investment immediately following the major movement which reached its peak in the third quarter of 1946, but there was a well-developed upswing (reaching its peak in the second quarter of 1953) following the major investment movement of 1949-50. On the other hand, the nondurables showed a much stronger investment cycle following the initial major postwar cycle, but no cyclical movement following the period of intensive inventory accumulation which reached its maximum level in the final quarter of 1950.

## TIMING AND CONFORMITY OF INVESTMENT: COMPARISONS WITH SALES TURNS

The timing and conformity analysis presented in table 13 for the comprehensive and industry series not only substantiates the preceding observations but also provides detail on the essential timing characteristics and relative cyclical sensitivity of the durables and nondurables series.

Among the durables, two of the four industry series (machinery and transportation equipment) showed inventory investment expansion movements in 1948, which were merely declines in the rates of disinvestment, but all showed investment expansions conforming to the final stage of sales expansions that culminated in 1953. Among the nondurables the tendency was clearly different: all industry series showed inventory investment cycles in 1948, but none (except the rubber-goods series) conformed to sales movements in the latter half of the succeeding expansion.

Industry	Lead (-) or lag (+) in months, in zones associated with reference turns-Korean war								
	1948 peak	1949 trough	1951 peak	1952 trough	1953 peak	1954 trough	1957 peak	1958 trough	
Total manufacturing Durable-goods industries, total Nondurable-goods industries, total Primary metals Machinery (including electrical) Transportation equipment (includ- ing motor vehicles) Stone, clar, and glass Food and beverages Paper Ohemical Petroleum and coal Rubber	$ \begin{array}{r} -4 \\ -7 \\ -6 \\ 4 \\ -7 \\ 4 \\ -9 \\ -5 \\ -4 \\ +2 \\ -1 \\ +1 \\ \end{array} $	$ \begin{array}{r} -7 \\ -5 \\ -14 \\ -5 \\ -7 \\ -7 \\ -4 \\ +4 \\ +7 \\ -2 \\ +10 \\ \end{array} $	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	(1) (2) (3) +2 (4) (4) (4) (4) (4) (4) (4) (4) (4) (4)	$ \begin{array}{c} -2 \\ -2 \\ -2 \\ -2 \\ -7 \\ -2 \\ +4 \\ (4) \\ (5) \\ (5) \\ (5) \\ (5) \\ -2 \end{array} $	$\begin{array}{c} -11 \\ -2 \\ (^{3}) \\ 0 \\ -11 \\ -11 \\ +7 \\ (^{5}) \\ (^{3}) \\ (^{3}) \\ +11 \\ -5 \end{array}$		+2 +1 +5	

 
 TABLE 13.—Timing and conformity of manufacturers' purchased-materials inventory investment to sales turns

A. TIMING MEASURES ¹

See footnotes at end of table, p. 35.

**TABLE 13.**—*Timing and conformity of manufacturers' purchased-materials inventory* investment to sales turns-Continued

	9 indust	ry series	4 durab indu:	le-goods strie <b>s</b>	5 nondurable-goods industries		
Timing and conformity comparisons	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean	
Number of comparisons	42 37 88	34 30 88 13 10 (6) (2) (2) 7	20 20 100	16 16 100 11 3 (2) (1) (0) 2	22 17 78	18 14 78 2 7 (4) (1) (2) 5	

### B. SUMMARY OF TIMING AND CONFORMITY

¹ Inventory investment series have been deflated to 1956; 1957-58 turns based on undeflated data.

Inventory investment turn occurs, but there is no sales turn.
 No turn occurs in either sales or inventory investment.
 Peak was submerged—a decline in disinvestment.
 Sales turn occurs but no matching inventory investment turn.

Source: Based on material from Department of Commerce.

#### CHART 6

#### PURCHASED-MATERIALS INVENTORY INVESTMENT, NEW ORDERS AND CHANGE IN INDEX OF INDUSTRIAL PRODUCTION, 1946-58 Α. Total Manufacturing





. . .

. .

...

CHART 6-Continued



#### CHART 6—Continued

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated investment cycles; circles, of undeflated cycles. All new orders data are undeflated.

Source: Investment and new orders data from Department of Commerce. Data defiated by the author. Industrial production series from Federal Reserve Board.

In general, conformity was higher among the durables: All 20 sales turns were matched by stock turns, whereas among the nondurables only 17 of the 22 sales turns could be matched. Moreover, the tendency to lead was much better established for durables than for nondurables. Only 2 of the 16 turns in the durables series which occurred at major sales turns (turns associated with business cycle turns) show a lag, whereas among the nondurables a lag appears in 7 of the 14 turns.

### TIMING AND CONFORMITY OF INVESTMENT: COMPARISONS WITH REFERENCE TURNS

Comparison of inventory investment movements to business cycle turns provides additional evidence that this investment is sensitive to business cycle influences and, further, that the durables group is significantly more sensitive to these forces than the nondurables. Turns in purchased-materials inventory proper could be matched with reference turns (including the Korean subcycle reference turns) in 40 of 54 possible comparisons (table 12). Table 14 shows that inventory investment turns could be matched in 44 of the 54 comparisons.

The durables show a perfect conformity score (inventory investment turns could be matched in each of the 24 comparisons). The nondurables could be matched in only 20 of 30 possible comparisons. Further, as was true in the comparisons with sales turns, the durables inventory investment showed a stronger tendency to lead: 15 of the 16 durables investment turns matched with business cycle reference turns led or turned roughly, coincidently, 10 of the 14 nondurables investment turns matched in the same way showed similar timing.

 
 TABLE 14.—Timing and conformity of manufacturers' purchased-materials inventory investment at reference turns

	Lead (-) or lag (+) in months											
Industry	Busine	ss cycle	Korea	n cycle	Business cycles							
	Peak Novem- ber 1948	Trough October 1949	Peak Febru- ary 1951	Trough June 1952	Peak July 1953	Trough August 1954	Peak July 1957	Trough April 1958				
Total manufacturing	-6	-5	-3	+2	-2	-9	-23	+1				
total	-6	-5	-3	-1	-2	0	-23	+1				
Nondurable-goods indus- tries, total	-6 (3)	-11 -5	-3 -6	( ² ) +2	( ³ ) —2	-6 0	-2	+4				
trical)	4-6	-5	-3	-1	-8	9	•					
Transportation equipment (including motor vehicles) Stone, clay, and glass Food and beverages Paper Chemical Petroleum and coal Rubber	$     \begin{array}{r}         4 \ 0 \\         -6 \\         -8 \\         -2 \\         -3 \\         0 \\         -3     \end{array} $	-5 -2 +7 +1 +4 +1 +13	$ \begin{array}{r} +3 \\ +3 \\ -3 \\ +6 \\ +15 \\ (3) \end{array} $	$ \overset{-1}{}^{+5}_{-10} \overset{+}{}^{+5}_{(2)} \overset{(2)}{(2)} $	$^{-2}_{+4}$ $^{(2)}_{(2)}$ $^{(2)}_{(3)}$ $^{(2)}_{-2}$	$ \begin{array}{c} -9\\ 0\\ (^2)\\ (^2)\\ +12\\ -3 \end{array} $						

#### A. TIMING MEASURES 1

B. SUMMARY OF TIMING AND CONFORMITY

	9 indust	ry series	4 durab indu	le-goods stries	5 nondurable-goods industries		
Timing and conformity comparisons	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean	
Number of comparisons Matching inventory turns Percentage of matching turns Leads more than 3 months Rough coincidences: Leads, 3 months or less Coincidences. Lags, 3 months or less Lags, more than 3 months	54 44 88	36 30 83 11 14 (8) (4) (2) 5	24 24 100 	16 16 100 9 6 (3) (3) (0) 1	30 20 67 	20 14 70 2 8 (5) (1) (2) 4	

¹ Inventory series have been deflated to 1956. 1957-58 turns based on undeflated data.

* No matching inventory investment turn.

³ This turn cannot be marked because data begin in early 1948. The data show, however, that there was a well-defined peak at least 6 months prior to reference peak. ⁴ Peak was "submerged"—a decline in disinvestment.

Source: Based on material from Department of Commerce.

#### CYCLICAL AMPLITUDE OF INVENTORY INVESTMENT

In addition to their higher conformity to business cycles, the durables series also move with greater amplitude than the nondurables. This characteristic is clearly shown for inventory investment in table When measured in relative terms, amplitude of total change 15. (peak to trough or trough to peak) is found to be larger for durables in every phase. Relative changes computed on a per-month basis alter the results in only one phase: the abrupt contraction in non-durables from the second to the fourth quarters of 1948 is larger than for durables.

Purchased-materials stocks in durables industries are somewhat smaller than in nondurables, averaging around 40 percent of total purchased materials for the period as a whole, and this serves to reduce somewhat the cyclical impact of the sensitive durables inventory When the magnitude of change in durables investment movements. and nondurables inventory investment is measured in dollars rather than relative to inventory levels (table 15), the latter half of 1948 is once again the only phase in which the contraction in nondurables inventory investment is found to be larger than in durables.

Industry	Peak- trough (1948-49)	Trough- peak (1949-50)	Peak- trough (1950-52)	Trough- peak (1952-53)	Peak- trough (1953-54)	Trough- peak (1953-55)	Peak- trough (1955-58)
			Total o	change (rel	ative) 1		
Total manufacturing Durable goods Nondurable goods	-7.15 -8.60 -8.27	+10.28 +14.94 +9.28	-7.82 -12.12	+2. 43 +7. 23	-4.57 -9.69 3 -3.93	+7. 44 +12. 72 +6. 38	-7.20 -14.78 -5.00
		Ch	ange per n	ionth (rela	tive) ¹		
Total manufacturing Durable goods Nondurable goods	-0.60 72 -1.38	+0.57 +.83 +.39	-0.37 67	+0.81 +.60	-0.38 65 244	+0.35 +.61 +.42	-0.34 70 13
		·	Total char	nge (million	is of dollars	5) 3	
Total manufacturing Durable goods Nondurable goods	843 415 588	+1,220 +700 +680	-1,017 -626	+336 +406	-627 -513 <b>3</b> -320	+1, 143 +921 +507	-1, 136 -1, 089 -415
		Cha	nge per me	onth (milli	ons of dolla	urs) ³	
Total manufacturing Durable goods Nondurable goods	-70 -35 -98	+68 +39 +28	-48 -35	+112 +34	-52 -34 3-36	+54 +44 +34	-54 -52 -11

**TABLE 15.**—Amplitude of change in purchased-materials inventory investment cycles. by phase, 1948-58

 ¹ Expressed as percentage of the mean level of inventories during the phase. Mean level is an average of the beginning and ending inventory levels of the terminal quarters of the phase.
 ¹ To permit comparison with change in durable goods, purchased-materials investment change in non-durable goods has been measured from the second quarter of 1953 to the first quarter of 1954.
 ¹ All 1948-54 data have been deflated (1947 dollars). Measures for the two most recent phases are based on undefleted those valued deter. undeflated (book value) data.

Source: Based on Department of Commerce data.

### TIMING AND CONFORMITY OF INVESTMENT: COMPARISONS WITH RATES OF CHANGE IN OUTPUT

Abramovitz' analysis led him to expect that rates of change in purchased materials would lag behind rates of change in output. Timing analysis of the postwar data tends to bear out this expecta-tion. In table 16 all turns in rates of change in output for total manufactures and for durables manufacturers could be matched by turns in purchased-materials investment, typically with a lag of investment behind output. Timing was irregular, however, especially for the nondurables category. It ranged from coincident to a 12month lag for total manufacturers, from a 3-to 9-month lag for durables, and from a 6-month lead to a 24-month lag for nondurables. The industry series shows the relationship to be closest where expected, among the durables (particularly the total machinery series), and to be poor or nonexistent among the five available nondurable series. The latter, by the nature of conditions under which materials are procured, would not be expected to show a close timing relation between rates of change in output and inventory investment. On the other hand, some of the lags are much too long to be explained in terms of delays in procurement and, for some of the series, irregularities in timing would seem to be too great to find a place in the Also there are leads in the investment turns which are theory. contradictory to Abramovitz' hypothesis.

FACTORS INFLUENCING THE BEHAVIOR OF PURCHASED MATERIALS

Thus far we have seen that Abramovitz' theory, although it goes far to explain the general behavior of purchased-materials inventories and investment, does not provide for the occurrence of certain observed timing relative to turns in sales, the business cycle as a whole, and rates of change in output.

		Lead (-) or lag (+), in months										
Industry	1949 troughs	1950 peaks	1951 troughs	1952 peaks	1953 troughs	1955 peaks	1958 troughs	Average timing	in phase, mid-1948 through 1958 i			
Total manufacturing Durable-goods industries Nandurable goods industries	$+3 \\ +3$	+3 +6	+12 +9	+6 +6	 0 +9	$^{+3}_{+3}$	+3 +3	+4.3 +5.6	86 82			
ries Primary metals	$-6 \\ -6$	+3 +6	(*) +3	(*) +6	+3 +6	+24 (*)	+6 (²)	(ª) +3.0	(3)			
electrical) Transportation equip	0	+3	+9	0	0	(1)	(7)	+2.4				
vehicles) Stone, clay, and glass Paper Chemicals Rubber	-6 +6 +9 +9 +9 (2)	+9 +12 +12 +12 ( ² )	(*) +15 (*) (*)	+6 ( ³ ) ( ³ ) ( ³ ) +6	(3) (4) (4) (4) (4) (4) (4) (4) (4) (4) (4	(2) (2) (2) (2) (2)	*(2) (2) (3) (3) (3)	+4. 2 (3) (3) (4) (3)				

TABLE	16Timing	of	purch	hased	-mat	erial	ls :	inventory	investment	to	rates	of	change
	-		_		ir	<i>i</i> 01	utŗ	out				•	•

¹ Undeflated investment data used for 1956-58. Months in phase computed after adjustment for average ¹ No timing comparisons were possible.
 ² No timing comparisons were possible.
 ³ Data lacked sufficient comparability for computation.

Source: Inventory investment computations based on material from Department of Commerce deflated by the author; output computations based on Federal Reserve Board Index of Industrial Production (Manufacturing). Food and beverages and petroleum and coal industries omitted because no timing comparisons were possible.

We may now ask if there are not other significant factors omitted by Abramovitz. Victor Zarnowitz has established that the durablegoods industries produce principally to order and that backlogs of unfilled orders vary in size in the course of the business cycle.⁷ In addition, the National Association of Purchasing Agents data show that "lead times" for procurement of materials vary cyclically. Abramovitz' contention that inventory objectives are related to output levels makes no provision for the possible effect that changes in the size of unfilled order backlogs have upon the willingness of selling firms to hold purchased materials. Neither does his theory provide for the impact of altered conditions of availability upon the desired inventory levels of purchasing firms.

It is reasonable to suppose that both these phenomena are impor-The necessity of carrying stocks arises principally from the tant. need to avoid the penalties suffered when interruptions occur in production or in the delivery of merchandise by a firm to its customers. Production stoppages, slowdowns, or the inability to supply merchandise in the required type or amount within a time period considered convenient and reasonable by the purchaser entail additional production costs and result in lost sales. They may also damage customer relations thereby reducing the market share of the firm. Consequently, management acts to avoid such risks by carrying stocks in a quantity adequate to provide sufficient protection against delays in delivery by suppliers as well as protection against increases in its own sales requirements. Just what constitutes "sufficient" protection is a matter for individual managers to decide. There are costs and risks incident to carrying additional stocks (e.g., costs of storage and insurance, risk of obsolescence, and adverse price movements) which must be set against the protection they provide. On the basis of all these factors, it would be logical to assume that increased backlogs of unfilled orders (or other assurances of steady or rising future output) or deteriorating availability of materials add to the desirability of carrying additional purchased materials stocks and that decreased backlogs or increased availability reduce the need for stocks.

In the sections which follow these two factors, the level of unfilled orders and availability of materials, along with a third—price movements—are considered in detail.

### UNFILLED ORDER BACKLOGS

Zarnowitz has noted that unfilled orders typically lead at peaks in activity, but turn roughly coincidently with activity troughs.⁸

After a major trading movement the large backlog of orders which will have accumulated can provide the basis for expanding shipments for some months, even while the backlog is declining. On the other hand, such a lead is not likely to occur during the latter part of a recession, since any increase would immediately tend to increase the level of production and, where inventories of finished stocks are held or the period of production is short, the level of shipments.

It follows that during late expansion, declining backlogs of orders constitute an invitation to trim any purchased-materials stocks that

⁷ Victor Zarnowitz, "The Timing of Manufacturers' Orders During Business Cycles," in Business Cycle Indicators, Geoffrey H. Moore, editor, Princeton for NBER, 1961, vol. I, pp. 426, 451. * Ibid., p. 451.

may have accumulated excessively during the preceding speculative period. Of course, declining order backlogs seldom signal the oncoming recession as clearly as they may appear in retrospect to have done, but they serve at least to put the manufacturer on his guard. Once made, the decision to curtail stocks is implemented with relative ease. Existing purchase orders may be canceled or, more probably, delivery postponed. Seen in this light changing levels of unfilled orders appear as a reasonable explanation of the observed early turns in purchasedmaterials stocks at business cycle peaks.

In late recession, however, there will be no comparable influence to exert pressure for larger stocks. Although new orders may turn up before the end of the recession phase, there will be little incentive to press for a higher level of purchased-material stocks until unfilled orders have begun to accumulate. When the decision to add to stocks is finally made, there still remains the delay occasioned by processing the orders and making delivery.

Although these observations may account for the observed lagging tendency of purchased-materials stock turns at troughs, they do not explain the duration of some of these lags. This may occur for any, or all, of the following reasons. (1) Since postwar recessions have been a year or less in duration the reduction of stocks which were at excessive levels during expansion may not have been completed. (2) The manufacturer may be expected to be less aggressive in raising purchased-materials stock levels in early expansion than he is in lowering these stocks in early contraction, since he is not under the same pressure to do so. The beginning of recession is accompanied by demands to conserve working capital, by a rapid increase in availability of goods (this point is discussed at length below), and by expectations of more favorable terms of purchase. In the early months after the trough, however, the manufacturer has no assurance that the incipient recovery will not collapse; prices are not as yet firming appreciably, and materials can still be procured with relatively little delay. (3) The manufacturer may fail to foresee the extent of the pickup as he places his purchase order and an undesired liquidation of inventories occurs; for this reason he underestimates. There is no evidence to indicate which of these is most important, but it is not unreasonable to suppose that all have played a part.

It will be observed in chart 7 that upward movements in purchased-materials inventory investment are substantial only when the increases in new orders are sufficiently in excess of the increases in shipments to bring about rapid accumulations of order backlogs, or when large order backlogs are not significantly reduced by current shipments at the time that a new surge in orders occurs. This generalization sheds light on the great bulge in total purchased-materials investment which occurred in 1946 and the small amplitude (relative to the very large increases in new business) of the second investment cycle which immediately preceded the 1948–49 recession. Similarly, it is helpful in explaining the huge movement in purchased-materials investment in 1950 and the relatively small movement which occurred just prior to the business cycle peak of August 1953. In each of these cases the major movements occurred under conditions of large and The succeeding smaller movements, however, ocrising backlogs. curred under conditions of declining backlogs even though the upward movements of new orders were quite sizable.



### CHART 7

MANUFACTURERS' UNFILLED ORDERS, RATIOS OF UNFILLED ORDERS TO SALES, New Orders, and Purchased-Materials Investment, 1946-58

See chart notes on p. 45.



[•] Ratio computed for midmonth in quarter. See chart notes on p. 45.



### CHART 7-Continued

• Ratio computed for midmonth in quarter.

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated investment cycles; circles, of undeflated cycles. All other data are undeflated.

Source: Department of Commerce. Data deflated by the author.

In addition, an explanation is provided for the previously observed differences in the behavior of durables and nondurables purchasedmaterials investment preceding the business cycle peaks of 1948 and 1953. It will be recalled that in the former of these two episodes, durables purchased-materials stocks were declining for months prior to the peak in shipments, with an inventory cycle evident after mid-1947 only in the rate at which disinvestment occurred; but nondurables showed a well-developed second postwar cycle in purchasedmaterials investment from third quarter 1947 to fourth quarter 1948. In 1952-53, the opposite was the case; nondurables declined after fourth quarter 1950, with only small irregular movements in inventory investment noted, whereas durables displayed a well-established cycle from second quarter 1952 to fourth quarter 1953.

The explanation is to be found in the differences in order backlogs during these two periods (chart 7). In the months immediately following the war, durables and nondurables experienced sharp in-creases in orders which were sufficient to carry unfilled orders to record peacetime heights. Under these conditions manufacturers increased their stocks of purchased materials sharply. Following this initial wave of demand there was a second wave (i.e., expansion of new orders) which occurred in both durables and nondurables (chart 6), but with quite dissimilar developments. In the nondurables the resurgence in demand in the last half of 1947 resulted in a sustained level of unfilled orders at close to record heights through the first quarter of 1948 (chart 7) and was the occasion for a second investment cycle in purchased materials.9 In the durables, the second wave of demand which began in late 1947 and continued through most of 1948, though relatively much more substantial than that in nondurables, was easily accommodated as manufacturers reconverted from wartime operation. The result is striking: durables unfilled orders fell almost continuously from the beginning of 1947 until the end of the recession, and there was only a submerged second cycle in the deflated purchased-materials investment series.

Beginning in 1949, both categories experienced upward movements continuing through the two great surges in buying in the third quarter of 1950 and first quarter of 1951, prompted by the Korean war; but after early 1951, movements in the two categories were dissimilar. When it was discovered that there were not to be important consumer shortages, the demand for a variety of nondurables declined and a well-defined recession in the related nondurables industries occurred. New orders fell sharply until 1951, when sales reached a trough. The recovery from this recession, which reached a peak in 1953, was comparable in amplitude of movement to that of 1947-48. The volume of goods demanded was not sufficient, however, to increase the level of unfilled orders, and sales which had been sustained previously by order backlogs gradually declined in spite of the increase in the volume of new business. Under these conditions deflated purchased materials investment declined throughout the latter part of the business cycle expansion.

In the durables category, defense orders sustained shipments and output to the degree that no well-marked recession developed, although there was a sharp downward movement in new orders from first to

⁹ It will be noted that a peak in unfilled orders was reached in November 1947, but order backlogs fell very little until after March 1948.

third quarter 1951. The backlog of unfilled orders rose continuously until September 1952 and remained at approximately this level until April 1953. Under these conditions the upsurge in new orders which lasted until May 1953 brought a concomitant increase in purchasedmaterials investment.

The relation between the level of unfilled orders and the vigor of inventory investment suggests that the capacity level may play a significant role in inventory investment and in the strength of recovery movements in the manufacturing sector. During a period characterized by extensive industrial overcapacity it is not to be expected that there will be a substantial accumulation of unfilled orders.¹⁰ Under such conditions firms would tend to refrain from any substantial purchased-materials investment. This attitude, if generalized, would work significantly against the development of a full-fledged recovery.

### AVAILABILITY OF MATERIALS

Fortunately, there are data on cyclical variations in availability of materials used by industrial firms. These data are found in vendor performance statistics collected by the Purchasing Agents Association of Chicago, and in quarter-to-quarter changes in total manufacturers' unfilled orders (Department of Commerce).

The vendor performance series is derived from monthly reports. submitted by member agents, in which each respondent states whether vendors' deliveries are slower, about the same, or faster than in the preceding month.¹¹ These reports have been consolidated into a diffusion index by combining for each month the percentage of agents reporting increased delays in delivery with one-half of the percentage reporting no change.

The index presumably records the approximate rate of change in availability of materials. When the value of the index is greater than 50, a rising series indicates that availability of materials is declining. at an accelerating rate and a declining series indicates continued deterioration, but at a decreasing rate. Conversely, when the index value is less than 50, a declining series indicates accelerated improvement in supply conditions, and a rising series, that the rate of improvement is slackening.

The series for manufacturers' unfilled orders provides essentially the same type of information. Unfilled orders are the accumulated difference between new orders received (less cancellations) and shipments. In a given period, the change in the level of unfilled orders will be the difference between the volume of received orders and the volume of If orders exceed shipments, unfilled orders will rise, and shipments. the larger backlog of unfilled orders will often make it necessary for the firm to quote longer delays to its customers. If, on the other hand, the firm is able to reduce the backlog, it may be able to reduce delays in delivery.

Accordingly, the rates of change in unfilled orders should provide a measure of the acceleration of deterioration, or of improvement in

¹⁹ An upward movement in orders would result in some accumulation of unfilled order backlogs because there is a necessary period which must elapse in the production process. This period prevents the outflow (shipments) from equaling the inflow (orders) as long as the inflow is rising. The unfilled orders which thus accumulate, however, are likely to be quite small in comparison to the backlogs which would arise with similar demand conditions and significant constraints upon increases in output. ¹¹/This series should not be regarded as merely an index of regional supply conditions, for firms in the highly diversified Chicago industrial area purchase from suppliers located throughout the United States.

supply conditions, in the same manner as does the vendor performance series. Such differences between the behavior of the two series as may appear, should stem from the difference in coverage,¹² or from the fact that one is a value series, and thus affected by price changes, whereas the other is a diffusion index unaffected by price.

It is interesting that these are the same unfilled orders data which were examined in relation to the inventory objectives of selling firms. But the logic is apparent: high or rising unfilled order backlogs bring assurance of a high level of operation to the seller, but to the buyer they bring delays in delivery and problems of procurement. As a result, both seller and buyer find justification for high levels of purchased-materials stocks. The converse holds for low or rapidly falling backlogs.

When the vendor performance and unfilled orders series are compared they are found to move in a similar manner. The troughs associated with the three business cycle recessions are coincident on two occasions and differ by only 3 months on the other. The major peaks in each of the 1950-51 and 1955 series are 6 months apart on one occasion, and 3 months on the other (chart 8 and table 17).

In addition to these major movements, the vendor performance series records very clearly two brief upswings, one beginning in 1947, the other beginning in 1952. These movements are less well defined in the unfilled orders series.

Another basis for comparing the series is the date at which they fall below the zero line, indicating the beginning of overall improvement in supply conditions, or rise above it, indicating the start of overall deterioration. The two series show close timing agreement in the dates at which supply conditions begin to become worse (chart 8). At two of these comparisons (third quarter 1949 and third quarter 1958), the turns are coincident; at the remaining one (late 1954), timing varies by 3 months. In comparisons of points at which supply conditions improve, the series agree less well. On one occasion (late 1946-early 1947), timing varies by 6 months; in the other comparisons, the unfilled orders series lags by 18 months (1951-52) and by 12 months (1955-56). A possible explanation for these substantial lags is that the total unfilled orders series, in contrast to the diffusion index in which each firm has equal weight, reflects in a larger measure the sluggish movements of order backlogs related to Government defense contracts (the extended lag following May 1951) and producers' durable equipment (the lag in 1956).

¹² The vendor performance series is based upon a sample whose coverage cannot be measured precisely, although it is restricted to the performance of manufacturers who supply industrial concerns. The unfilled orders series is well defined as to coverage but provides information on all manufacturers, regardless of their role as suppliers.

.

TABLE 17.—Timing of vendor performance series to quarterly change in unfilled orders and unfilled-orders-to-shipments ratios, all manufacture         1947-58
---------------------------------------------------------------------------------------------------------------------------------------------------------------

	Trough	Peak	Trough	Peak	Trough	Peak	Trough	Peak	Trough
Turns in vendor performance ¹ . Turns in quarter-to-quarter	May 1947	August 1948	May 1949	August 1950	February 1952.	August 1952	November 1953.	August 1955	February 195
change: • Total manufacturers un- filled orders. Unfilled-orders-to-ship- ments ratios.	0	0	0	+6 +6	3 3	—3 0	0 +3	+3 +12	-: -(

¹ Percent of purchasing agents (Chicago area) reporting increased delays in delivery plus one-half of those reporting no change.

Source: Vendor performance series compiled from statistics collected by Purchasing Agents Association of Chicago; other series based on material from Department of Commerce.



CHART 8

VENDOR PERFORMANCE AND CHANGE IN UNFILLED ORDERS AND RATIOS OF UNFILLED ORDERS TO SALES, ALL MANUFACTURERS, 1946-58

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of specific cycles. Data are quaterly. For description of vendor performance series, see table 17. Change in unfilled orders data are end-of-quarter to end-of-quarter. Changes in unfilled orders-to-sales ratios are from midpuarter month to midquarter month.

Source: Vendor performance data from Purchasing Agents Association of Chicago; other data from De-partment of Commerce.

Another method by which the unfilled orders data may be made to reflect supply conditions is to express them as a ratio to current shipments. Such a ratio should measure the average delay (in months), assuming that the current rate of shipments is a reliable measure of current ability to supply goods.

There are two possible objections to this measure, however. First. production requires a significant interval. The current rate of shipments is therefore influenced by previously existing demand conditions and does not necessarily represent accurately the current ability of manufacturers to fill orders. Second, these data measure shipments by manufacturers who produce to stock out of finished goods as well as by manufacturers who produce to order. Accordingly, the total shipments series may be a faulty standard of current ability to make shipment on order.

Quarter-to-quarter changes in ratios of unfilled orders to shipments are presented in chart 8, and appropriate timing measures in table 17. Timing of this series is substantially the same as that for the rates of change in unfilled orders except in one comparison, the peak in 1956.

The turns in the data proper show less close agreement, but there is no substantial difference except at the most recent turn.

The behavior of these series demonstrates certain cyclical characteristics which are apparent in each of the three postwar business cycles:

1. Availability of materials reaches its maximum at approximately the trough of the business cycle and begins to deteriorate with the beginning of recovery, or very shortly thereafter. In no instance does either the vendor performance series or the unfilled orders ratesof-change series rise above the zero line prior to the reference trough. Supply conditions deteriorate at an accelerating rate during early expansion and then at a diminishing rate.

2. By mid-expansion, supply conditions begin to improve, although the high levels of unfilled orders in the three observed expansions attest to substantial delays. The improvement may proceed at varying rates, but it is continuous throughout the latter half of the expansion.

3. In the final months of expansion and during early recession, availability improves at an accelerating rate. The maximum rate of improvement is attained before the end of the recession.

### PRICE MOVEMENTS

Although it is not within the scope of this study to analyze the behavior of prices, it seems reasonable to suppose that there exists a more than incidental relationship between cyclical movements in prices of purchased materials and changes in the rates of their accumulation. During periods of rising demand, tightening supply, and rapid inventory accumulation, price increases are likely to occur. Under certain conditions such price rises may, in turn, set off inventory buying which will give rise to still further pressure upon prices; under other conditions the interaction with inventory movements might be expected to be small.

Recent work by Victor Zarnowitz sheds new light on the conditions which may cause cyclical price movements. Zarnowitz has advanced and tested the hypothesis that "in industries in which a high or substantial share of output is produced to order, price change is a positive function of the change in the [unfilled order] backlog volume."¹³ He finds, as set forth in a recent progress report, that for major industries which have a relatively high degree of competition, high coefficients of correlation between price change and backlog change may be noted. But "in industries with heavily predominant elements of imperfect competition or oligopoly the association between price changes and backlog changes, while still positive, is rather weak."¹⁴ Zarnowitz concludes that "while the response of prices is thus found to be in the direction consistent with the hypothesis, its extent is often small, e.g., the elasticities of price change with respect to backlog change are mostly low (less than unity for all the major manufacturing industries examined with the exception of textile-mill products)."¹⁵

These findings suggest two points which are relevant to the present study: (1) For many industries, especially in the durable goods sector where inventory investment movements are most volatile, the influence of price changes on inventory behavior may be of rela-

^{19 40}th Annual Report of the National Bureau of Economic Research, New York, 1960, pp. 40-41. 14 Ibid. 19 Ibid.

tively little importance; and (2) the relationship between price movements and changes in unfilled order backlogs is consistent with and lends support to the hypothesis that the amplitude of movement in purchased-materials inventory investment is conditioned by the level and rate of change of unfilled orders.

The conditions of "excess demand" which, Zarnowitz finds, accompany price fluctuations are the same as those which prompt a high level of inventory accumulation, because (a) the price changes which are observed and anticipated by purchasing firms stimulate inventory accumulation, (b) the delivery problems which accompany such market conditions make additional purchased-materials stocks desirable, and (c) inventory accumulation itself influences price movements (i.e., there is a reciprocal relationship).

### INVENTORY OBJECTIVES AND THE PURCHASING PROCESS

In the foregoing section it was shown that the availability of materials fluctuates cyclically. The next task is to examine series which provide evidence of the manner in which inventory objectives and purchasing policy are affected.

### MOVEMENTS IN NEW ORDERS AND PURCHASED-MATERIALS INVESTMENT

There is a strong tendency for purchased-materials investment to move in a manner similar to new orders. For the comprehensive data, this is evident at most turns during the entire period. Among three of the industry series ¹⁶ one (machinery) shows similar movements throughout the period, a second (transportation equipment), for part of the period, and a third (primary metals) displays a considerable degree of irregularity.

It will be noted (table 18 and chart 6) that the important departures from good conformity occurred during periods when unusual conditions existed. The inventory investment series for total manufactures, durables, and transportation equipment all show lags of 9 months behind new orders at the 1952 investment troughs. This appears to be due principally to the exceptionally large backlog of unfilled orders among the durables which did not diminish significantly until after Under these conditions purchased-materials investment mid-1952. declined very sluggishly, the contraction finally terminating in a sharp downward movement of both stocks and investment in the second quarter of 1952. The abruptness of this decline may be at least partially attributed to the extensive steel strike which occurred at that time and which reduced the flow of this raw material for almost 2 months.

¹⁶ Of the nine industry classifications for which stage of fabrication inventory data are available, it was possible to procure new orders figures for only three: primary metals, total machinery, and total transportation equipment.

	Lead (-) or lag (+) in months in zones associated with reference turns										
Lndustry		Percent of months in phase									
	1948 peak	1949 trough	1951 peak	1952 trough	1953 peak	1954 trough	Mid- 1948 through 1955	Mid- 1948 through 1958			
Total manufacturing. Durable-goods industries, total Nondurable-goods industries, total. Primary metals 3 Machinery (including electrical) Transportation equipment (includ- ing motor vehicles)		0 -3 -3 -3 0 +3	-3 -3 -3 -6 -3 0	+9 +9 0 +3 0 +9	+3 +3 +6 0 -6 +12	0 0 +3 +6 +3 +3 +3	79 80 83 60 87 63	76 866 79 (*) (*)			

TABLE 18.—Timing of purchased-materials inventory investment to new orders 1

¹ Inventory investment series are quarterly data. Computations were based on undeflated inventory data to permit comparisons with new orders material for which no deflation was possible. As a matter of caution a comparison was made based on deflated inventory data. The results were approximately the same as those shown. New orders data are quarterly totals. ³ This series showed 2 extra cycles not found in new orders,

Not available.

Source: Based on material from Department of Commerce.

In addition to this departure from close agreement, the transportation-equipment investment series also shows a 12-month lag behind new orders at its last (second quarter 1953) peak. Here again the explanation would appear to be found in the behavior of order backlogs, for unfilled orders continued to rise in the transportation equipment industry after they had turned down elsewhere. New orders for transportation equipment turned down after the second quarter of 1952, but remained above the level of shipments until the end of the first quarter 1953.

Although all turns in new orders for primary metals could be matched from 1949 on, there were two extra cycles in the investment series with the result that overall conformity was low. The poor conformity of the primary metals series cannot be explained on the same grounds as given for the transportation equipment series, but no close conformity should be expected. Producers of primary metals utilize basic ores and other raw materials which are procured only with considerable delay. It is not to be anticipated that investment in these stocks will move as sensitively as the inventory investment of machinery manufacturers, for example, whose purchased-materials stocks are drawn to a considerable degree from standardized, readily available metals and metal parts.

I have extended the analysis to the 1956-58 period for total manufacturers, total durables, and total nondurables. Chart 6 shows the related new orders and inventory investment series moving together in what appears to be the same close relationship noted for the earlier Table 18 substantiates this: the percentages of months in period. phase are not significantly altered by adding the data for these three more recent calendar years.

But why does this close association between orders and purchasedmaterials investment exist? If the new orders data are looked upon as a purchase orders series, their close relationship to purchasedmaterials inventory investment are revealed: ¹⁷ (a) When manufacturers are attempting to increase stocks at increasing rates they will tend to increase purchase orders above previous levels; (b) when manufacturers are attempting to decrease the rate of accumulation of purchased stocks, but either desire or permit them to continue to rise, they will tend to place smaller orders than previously; (c) during periods when manufacturers are attempting to reduce stocks outright, they will reduce their orders still more; and (d) when manufacturers are allowing their stocks to continue to fall but are attempting to decrease the rate of liquidation they will increase the size of their orders.

Thus it is clear that the pattern of new orders placed will be strongly influenced by action taken to adjust inventories. Moreover, the effect should occur in a roughly instantaneous fashion. In late expansion when manufacturers wish to reduce the rate of inventory accumulation or decrease stocks outright, they will reduce orders placed and request delays of goods currently scheduled for delivery. It may not be within the power of these ordering firms to halt immediately the increase in purchased materials, for they are likely to be committed to their suppliers for a sizable quantity of goods scheduled for future delivery. It would be surprising, however, if they could not sharply alter the rate of accumulation on very short notice. Similarly, in late recession manufacturers who wish to reduce the rate of decline in purchased-materials stocks or to increase them outright will immediately increase orders. Again, the manutacturer may not be able, nor may he desire to halt instantaneously the direction of movement in purchased stocks, but he should be able to alter sharply the rate of liquidation.

### CYCLICAL CHANGES IN PURCHASING POLICY

The data relating to cyclical behavior of buying policy originates in reports to the National Association of Purchasing Agents regarding current purchasing policy. Firms are reported as purchasing predominantly on a hand-to-mouth basis, on the basis of 30-day requirements, 60-day requirements, 90-day requirements, 6-month requirements, 1-year requirements or more. The data have been consolidated into two categories: percentage of firms purchasing on the basis of current requirements (30 days or less) and percentage purchasing on the basis of more than current requirements (60 days or more). The latter series is presented in chart 9.

Although this time series is not long, it provides considerable information on the cyclical variations which occur in purchasing policy. The first data became available in January 1950, only 3 months after the reference trough in late 1949. We are able to observe, therefore, almost the entire expansion of 1949-53, the complete expansion of 1954-57 and the first year of the most recent expansion, as well as the recessions of 1953-54 and 1957-58. Of particular

¹¹ The Department of Commerce new orders series concern orders received but there are two reasons why such data may reflect new orders placed as well. (1) To the extent that the orders reported as received originate with other firms within the same industry they are, of course, authentic purchase orders data for that industry. The larger the industry group, the more likely is this to be the case. (2) To the extent that orders received by manufacturers give rise to purchase orders placed by reporting firms. Evidence that firms tend to place purchase orders synchronously with the receipt of new orders has been presented by Ruth Mack and Victor Zarnowitz. See Ruth P. Mack, "Consumption and Business Fluctuations," New York, V1956 and Ruth P. Mack and Victor Zarnowitz. March 1958, 18-49.

#### CHART 9





 Purchasing policy series is percent of purchasing agents reporting 60 days or more delay in delivery. January 1950 to April 1955 data cover all purchases; subsequent data, production materials only.
 For description of series see table 17.

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of specific cycles.

Source: Purchasing policy series compiled from National Association of Purchasing Agents data. Vendor performance series compiled from Purchasing Agents Association of Chicago data. Other series compiled from Department of Commerce data.

interest is the behavior of the purchasing-policy series in early expansion and during recession. In each of the three expansions, observable in whole or in part, the early months are characterized by very sharp increases in the percentage of firms purchasing on the basis of more than current requirements. Between January and June of 1950 (the remainder of the year is excluded because of the crisis buying which accompanied the outbreak of Korean hostilities) the percentage of firms purchasing on this long-range basis rose from 53 to 69 percent. In the period June 1954 to December 1955, it rose from 41 to 80 percent,¹⁸ and from April 1958 to April 1959, it rose from 43 to 68 percent.

It will also be noted that during the two complete expansions observed, the percentage of firms purchasing in the 60-day-or-longer range reached a maximum by midphase. The range of purchase began to shorten well before the reference peak, but the sharpest drop occurred during the period extending approximately from the reference peak month to the date of the reference trough. In the period from May 1953 to May 1954, the series fell from 64 to 40 percent; and in the period from May 1957 to February 1958, it fell from 71 to 41 percent.

Taken together the series representing purchasing policy, supply conditions, new orders, and purchased-materials investment provide valuable insight into the processes by which inventory objectives fluctuate in the course of the business cycle.¹⁹

As already observed, movements in the new-orders series appear, to a significant extent, to reflect buyers' efforts to adjust stocks; i.e., a rising volume of orders would seem to reflect attempts of buyers to decrease the rate of disinvestment in purchased goods or to increase the rate of investment; and a declining volume of orders, attempts to reduce the rate of investment in purchased goods or to increase the rate of disinvestment. Of course, the new-orders data are comprised principally of purchases for production or for resale rather than for inventory accumulation; in general, their rise and fall must be largely in response to changes in final demand. Nevertheless, inventory investment demand is superimposed upon this demand, and the data give evidence that the former influences significantly the timing of turns in the aggregate order series. This was inferred in the previous section from the similarity in the timing of turns in new orders and in purchased-materials investment. It is now supported further by the close agreement between timing of both of these series and the purchasing-policy series.

¹⁹ The data are not perfectly comparable throughout, however. Prior to May 1955 agents submitted reports in which all types of purchases were combined. From May 1955 forward, reports were made on the basis of three categories: production materials, maintenance and materials repair, and capital expenditures. Only data covering production materials buying are shown for this latter period. ¹⁹ This discussion is based on charf 9 and table 19, which concern vendor-performance, purchasing-policy, or descent production materials in which concern vendor-performance, purchasing-policy, or descent period.

orders, and purchased-materials investment series.

TABLE 19.—Timing comparisons: Vendor performance, purchased-materials investment, total manufacturers new orders, purchasing policy series, 1946-58

Series compared	Peak, August 1950	Trough, May 1952	Peak, February 1953	Trough, May 1954	Peak, August 1956	Trough, May 1958
Vendor performance	0	-3	+6	-6	12	-3
ment	+3	0	+3	-6	-12	0
orders	+6	9	0	-6	+3	-3

A. TIMING TO PURCHASING POLICY SERIES

#### B. TIMING TO PURCHASED MATERIALS INVESTMENT SERIES

Series compared	Trough,	Peak,	Trough,	Peak,	Trough,	Peak,	Trough,	Peak,	Trough,
	August	May	May	Novem-	May	May	Novem-	August	May
	1947	1948	1949	ber 1950	1952	1953	ber 1953	1955	1958
Vendor performance Purchasing policy Total manufacturers' new orders	-3 (1) (2)	+3 (1) +3	(1) 0	-3 -3 +3	-3 0 -9	-9 -3 -3	0 +6 0	0 +12 +15	3 0 3

¹ Not available. ² No turn.

Sources: Vendor performance data compiled from statistics collected by Purchasing Agents Association of Chicago; purchasing policy series compiled from data published by the National Association of Pur-chasing Agents. Other series based on material from Department of Commerce.

In general, during periods in which total new orders are rising, the purchasing-policy series is also rising (i.e., the typical purchase range is lengthening), thereby indicating that the volume of orders placed during these periods of expanding trade is rising in relation to current operating requirements. It is at this time that realized disinvestment is declining sharply (in the very early months of expansion) or realized investment is rising (as the expansion progresses).

During periods in which total new orders are falling, the purchasingpolicy series, declines, revealing that the volume of purchase orders is falling in relation to current operating requirements (i.e., the typical purchasing range is shortening). During such periods realized investment is found to be decreasing (or realized disinvestment increasing).

The inference that the rise and fall in orders reflects, to a significant degree, the efforts of firms to alter the rate of accumulation or deaccumulation of stocks, is supported by the fact that orders lead turns in business cycles. Since turns in manufacturers' shipments, as well as turns in retailers' sales, show no similar tendency to lead cycle turns, it may be presumed that the early turns in orders do not reflect a reversal in end-use demand, but rather an attempt by businessmen to adjust their stocks or their purchase commitments.

In addition, the role played by supply conditions in cyclical variations in purchasing policy is an important one. To be sure, it is not possible to distinguish conclusively between cause and effect here: supply conditions deteriorate because purchasing policy alters and orders increase; purchasing policy may change and orders may increase in turn, because supply conditions deteriorate. Yet, the similarity of movement in the vendor-performance and purchasing-policy series indicates some degree of causation in the role played by supply conditions, particularly in the light of the functions exercised by purchasing executives.²⁰ It is difficult to imagine that the changes in vendor performance observed in the data do not prompt executives to revise their inventory target levels. Indeed, if purchasing agents were not influenced by such considerations, one of their major associations would not collect and publish the data on vendor performance examined here.

The influence of supply conditions may operate in two ways during expansions: (1) Deterioration in supply conditions makes it more difficult to achieve inventory objectives, so realized inventory investment will be somewhat less than that desired; and (2) deterioration in supply conditions influences the inventory objective itself.

The second is by no means a simple concept. If it were, we might expect to find that the purchasing range would be rising as long as availability is deteriorating; that is, as long as the vendor-performance series lies above the zero line. In fact, however, the turn in the purchasing-policy series occurs at approximately the same time as that in the vendor-performance rate-of-change series.

One can only speculate regarding the explanation of this intriguing relationship. One suggestion is that purchasing agents may view the declining rate of deterioration in vendor performance as a signal that delivery conditions will begin to improve in the near future, and thus, via a change in anticipations, the range of purchase is shortened.

### SUMMARY AND CONCLUSIONS

Although there are variations in the behavior of individual industry series, certain characteristics of purchased-materials behavior appear to be clearly established. Purchased-materials stocks conform well to cyclical movements, turning earlier at sales and business cycle peaks than at troughs. Turns in the comprehensive series tend to coincide with business cycle peaks and lag at business cycle troughs. This timing characteristic does not carry over to inventory investment behavior: the comprehensive investment series show leads at all business cycle turns. In these movements the durables series show a higher conformity and a greater amplitude of movement.

The well-established tendency of inventory investment turns to lead reference turns during the period under study is of major significance for business cycle analysis. In late expansion the decline in the level of inventory investment has constituted a source of declining demand for the factors of production, and during late recession the decline in the rate of disinvestment has provided a source of new strength in the economy. To the extent that these observations are typical, purchased-materials investment may be regarded as one of those highly strategic forces which serve to turn the cyclical tide.

Abramovitz' theory does not provide for the occurrence of certain observed timing in the inventory and investment series. Moreover, it fails to consider the possible role of unfilled-order backlogs, availability of materials, and price behavior of purchased goods in influencing inventory behavior. In an effort to provide a fuller explanation, these influences were examined.

¹¹Note that the two series are reported by different purchasing agent groups, so that they are, at least statistically, independent.

The amplitude of movements in purchased-materials investment was found to be related to the size and direction of movement of manufacturers' unfilled-order backlogs. Series reflecting availability of materials were constructed from the vendor-performance data of the Purchasing Agents Association of Chicago and from unfilled-orders data. From the behavior of these series it was observed that—

(1) Availability of materials reaches its maximum at approximately the trough of the business cycle and begins to deteriorate with the beginning of recovery, or shortly thereafter. Supply conditions deteriorate at an accelerating rate during early expansion and then at a diminishing rate.

(2) By midexpansion, supply conditions begin to improve although there are still substantial delays in delivery.

(3) In the final months of expansion and during early recession availability improves at an accelerating rate. The maximum rate of improvement is attained before the end of the recession.

There is evidence that industrial prices move in response to the same conditions of demand and supply that influence purchasedmaterials inventory investment. It is reasonable to suppose that they not only respond to such forces but also play a causal role. However, for many industries, especially in the durable-goods sector, the influence may be relatively small.

In a final section, the behavior of inventory objectives and purchasing policy was observed in relation to the availability of materials and realized inventory investment. Data relating to leadtime of purchase show that the cyclical behavior of purchasing policy is welldeveloped; the series move in a manner similar to that of the series reflecting availability of materials. Moreover, the timing of the purchasing-policy series agrees rather closely with that of manufacturers' new orders and purchased-materials inventory investment. Taken together these series tell of a sensitive process by which rising inventory objectives and deteriorating supply conditions are matched by rising orders and an ever-increasing rate of inventory accumulation in the earlier stages of expansion. Reversal begins in late expansion, but the sharpest declines in inventory objectives, orders, and realized inventory investment, as well as the sharpest improvement in supply conditions, occur during recession.
## CHAPTER 5

## FINISHED-GOODS INVENTORIES AND INVENTORY INVESTMENT

In analyzing the behavior of finished goods, Abramovitz distinguished two major types: finished goods made to order and finished goods made to stock. The second category was subclassified according to whether the production cycle of the commodity was governed by demand or by conditions of raw materials supply, and according to whether the finished products were perishable items or staples. This classification may be outlined as follows:

- I. Goods made to order.
- 11. Goods made to stock:
  - A. Goods whose production cycles are governed by demand: 1. Perishables.
    - 2. Nonperishable staples.
  - B. Goods whose production cycles are governed by the supply of raw materials:
    - 1. Perishables.
    - 2. Nonperishable staples.

Abramovitz drew no conclusions regarding the timing of finishedgoods inventories as a whole, but instead dealt with the behavior of the component classifications. Goods made to order he saw as tied to the level of output.¹ Of those goods whose production cycle is governed by demand, nonperishable staples would tend to move inversely to the rate of manufacturing and to business cycles; perishable goods, even if produced to stock, would be expected to vary with the volume of business, moving with, perhaps, a moderate lag. Goods whose output is governed by supply would not be expected to be cyclically sensitive. Their behavior would be influenced by the conditions of supply and the degree of perishability of the product.

Knowledge of the relative importance of these several components may be gained by examining the composition of finished stocks, comparing Abramovitz' estimates for the prewar period with postwar estimates. Such an investigation is made in this chapter; it is followed by an analysis of inventory behavior which deals first with the largest component, staple (nonperishable) made-to-stock finished goods whose production is governed by demand, and then with total finished goods. In subsequent sections inventory investment behavior is treated in a similar manner.

## Composition of Stocks

Abramovitz' estimates of the size of the various components was based largely on an analysis of finished goods within each industrial

¹The statements contained in this paragraph are based on Abramovitz, "Inventories and Business Cycles, With Special Reference to Manufacturers Inventories," New York, November 1950, pp. 240-241.

breakdown of the 1939 Census of Manufactures. He used the following classification:

1. Finished goods whose raw materials were procured from the agricultural sector:

(a) perishable.

(b) nonperishable staple.

2. Finished goods whose raw materials were procured from the nonagricultural sector:

(a) perishable.

(b) nonperishable staple.

His investigation was made on the assumption that commodities whose raw materials were procured from the agricultural sector would have production cycles that were supply determined and from the nonagricultural sector, production cycles that were demand deter-The assumption results in an underestimate of the demand mined. determined category, in that there are important industries using agricultural raw materials that are not supply dominated (e.g., cotton The approach does, however, indicate the importance of textiles). nonfarm goods which, by and large, can be reliably classified as demand dominated. Moreover, it makes possible the identification of those industries for which agricultural raw materials are most important, and permits, through an inquiry into the nature of their production processes, the determination of whether their fluctuations are supply or demand influenced.

A duplication of Abramovitz' techniques, making use of 1947 Census of Manufactures data, revealed the following:

1. The staple demand dominated category was by far the largest; it is estimated at roughly 66 percent. Staple finished goods whose raw materials are procured from the nonagricultural sector accounted for 55.3 percent of total finished stocks; some produced from agricultural raw materials, but clearly demand dominated, accounted for 10.3 percent more.²

2. Those industrial categories using agricultural raw materials which may be supply influenced, and whose finished stocks are staple, accounted for about 23 percent of total finished goods. Thev included principally the food, beverage, and tobacco industries which are probably the least sensitive to business cycles.

3. The perishables are a relatively small category. Perishable finished goods whose raw materials are agricultural were estimated to be 9.1 percent; those with nonagricultural raw materials, 2.4 percent.

This classification of census material provides no insight into the size of the made-to-order category, but there is reason to believe that it is relatively small. Abramovitz estimates that it is no more than 15 to 25 percent of total finished goods stocks in spite of the fact that a much larger volume of business is done on a production-to-order basis.³

I estimate that in 1947 the staple finished stocks held in the textile industry accounted for 5.8 percent of total finished goods; in the apparel industry, 3.1 percent, and in the leather goods industry, 1.4 percent. All are demand-oriented industrial activities, although, according to Abramovitz' classification, they make important use of agricultural raw materials.
 I found finished stocks-to-sales ratios for goods made to order to be much lower than similar ratios for goods made to stock, indicating that the magnitude of the former is much smaller than might be implied by the importance of the practice of producing to order (see footnote 7 below). For an analysis of the importance of production to order as a postwar practice in American manufacturing see Victor Zarnowitz, "The Timing of Manufacturers' Orders During Business Cycles," in Business Cycle Indicators, vol. I, Princeton for NBER, p. 426.

## INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION 63

If we accept Abramovitz' rough estimate that 15 to 25 percent of finished stocks are made to order and deduct this from the 66 percent of total finished stocks previously designated as the staple demand dominated category, then staple made-to-stock finished goods in industries whose production cycles are demand dominated comprise from 41 to 51 percent of total finished goods. This estimate, combined with the percentages of perishables and the supply-dominated staples given above, provide an estimated composition of finished goods in 1947 which is compared with Abramovitz' estimates for 1939 in table 20.

Two observations may be helpful in interpreting the subsequent analysis of finished goods behavior. The first is that demand-dominated, nonperishable, made-to-stock finished goods, though important, are less so than before the war. This change in composition should make for somewhat less inverted behavior in the entire category. The second is that the estimate of made-to-order finished goods may well be too low. I have accepted Abramovitz' very tentative figure in the absence of other evidence, but a larger made-to-order category is quite possible in view of the increased importance of durables (produced principally to order) in the postwar period. Of course, a larger made-to-order component would also tend to reduce inverted behavior in the entire finished-goods category.

Components of finished goods stocks	1939 (percent)	1947 (percent)
Goods made to order. Goods made to stock Goods whose production cycles are governed by demand: Perishables Nonperishable staples Goods whose production cycles are governed by supply: Perishables Nonperishable staples	15-25 75-85 ⁽¹⁾ 50-60 ⁽¹⁾ 16	15-25 75-85 2 41-51 9 23

TABLE 20.—Estimated composition of finished goods stocks, 1939 and 1947

¹ Total perishables were estimated to be 9 percent. There was no breakdown.

Source: For 1939 estimates see Abramovitz, "Inventories and Business Cycles," p. 246; 1947 estimates were prepared from 1947 Census of Manufactures data.

In evaluating the significance of composition we must keep in mind that there does not exist in practice the sharp distinction between firms producing to stock and firms producing to order that is implied by the preceding classification. One finds new cases in which the producer sells entirely on an immediate delivery basis out of stock already on hand, and relatively few in which there is no production except to order. There will usually be at least some orders taken for subsequent delivery; and the practice of producing to stock may vary from that of a firm which produces a line of standardized items, selling freely from stock, but studying closely such orders for future delivery as it receives as a guide in planning production—to the firm which produces almost entirely against orders, but allows a relatively small quantity of buffer stocks to fall and rise against the cyclical tide.

Nevertheless, manufacturers do tend to operate predominantly on either a produce-to-stock or produce-to-order basis. Evidence that variability of practice may exist within these well-established classifications is found in the results of tests performed by Victor Zarnowitz,

76626-61-pt. I---6

which revealed that made-to-stock and made-to-order activities could be distinguished by the size of average ratios of stocks to unfilled orders over a period of years. Working with a variety of manufac-turing activities and industry groups which had previously been reliably classified as producing principally to stock or to order, Zarnowitz found that the stock-to-unfilled-orders ratios fell into a well defined dichotomy: ratios for the made-to-order series had average values substantially below the level of 1; ratios for the made-to-stock series, values above 1.⁴ Zarnowitz also notes that, "according to the ratios * * * industries which sell mainly from future output are decidedly dominant in the composite of all durable manufactures. In contrast, production to stock apparently prevails within the aggregate of nondurable goods industries." 5

## THE STAPLE, MADE-TO-STOCK INVENTORY SERIES

Abramovitz studied the behavior of demand-dominated staples made to stock with a sample of 18 nonfarm commodities.⁶ In the present study, 25 series, representing a variety of activities, are analyzed. Among them are postwar extensions of some of Abramovitz' original 18 series. The only available commodities which have been omitted are those which would have given roughly duplicate coverage (e.g., additional lumber series, some other categories of gas heaters, men's hosiery).

Abramovitz' 18 series, as well as the 25 here, may be regarded as representing activities in which production is principally to stock rather than to order, although several of the postwar series would seem to be on the borderline.⁷

## TIMING AND CONFORMITY OF STOCKS DURING ACTIVITY CYCLES

The 25 commodity series display well-developed timing and conformity characteristics. They tend to move with good conformity in a strongly inverted fashion relative to activity (see table 21).⁸ Ninetyone inventory turns may be matched with the 109 activity turns on an inverted basis.9 In a large majority of the comparisons (64 out of 91) the inverted turns in stocks were roughly coincident with activity The remaining comparisons were about equally divided turns. between leads and lags of 4 or more months. Average timing for all comparisons on an inverted basis was approximately coincident.

Zarnowitz, in Business Cycle Indicators, vol. I, p. 426.

⁴Zarnowitz, in Business Cycle Indicators, vol. 1, p. 220. ⁴ Ibid. ⁶ We shall have occasion to refer frequently to this category of finished goods. For the sake of convenience ⁴ it shall be designated henceforth simply as "staples made to stock." ¹ In attempting to apply the Zarnowitz test to the 25 series it was found that only 8 have unfilled orders data available. Of these, seven (gas domestic heating stoves, bathtubs, lavatories, southern pine lumber, oak flooring, paper, and ladies full-fashioned hosiery) are clearly made-to-stock activities with stocks-to-unfilled-orders ratios ranging from 1.39 to 5.15. The other series—paper products—is a borderline case with a ratio of 0.62 with a ratio of 0.62.

with a ratio of 0.62. In an effort to classify the remaining series, stock-to-sales ratios were tested as an alternative basis of classification. The series for which both types of ratios were available were ranked first according to Zarnowitz' average stocks-to-unfilled-orders ratios, and then according to average stocks-to-sales ratios during the period 1948-55. The ranks were identical. Using the stocks-to-sales in lieu of stocks-to-unfilled-orders ratios it was then possible to classify the entire sample. With 2 exceptions (newsprint, 0.29, and wood pulp, 0.10) all of the series had stock-to-sales ratios considerably larger than the border-line paper products (stocks-to-sales ratio, 0.42); 16 had ratios above 1, 20 above 0.70.

considerably larger than the bolder-line paper products the entropy and the paper products the entropy and the paper products the entropy and the paper product the paper product of the secure such data for 20 of the 25 commodities. For the remaining commodites, output data were used. As Abramovitz has shown, the two give very similar results. See Abramovitz, "Inventories and Business Cycles," p. 248. Immediately following the war there occurred a long period of finished-goods inventory buildup. No inverted behavior of stocks took place until the recession of 1948-49. For this reason the comparisons were not begun on an inverted basis until the beginning of the 1948-49 recession period.

# **TABLE 21.**—Timing and conformity of 25 commodities at activity turns, 1947–55

_						
	Stocks	Indicator of activity	Activity turns ¹	Turns matched ¹	Extra turns	Percent months out of phase ³
1. 2. 3. 4. 5. 6. 7. 8. 9. 10. 11. 2. 3. 4. 5. 6. 7. 8. 9. 10. 11. 12. 13. 14. 5. 16. 7. 18. 19. 20. 1. 22. 3. 4. 25.	Gas domestic stoves	Shipments.           do.           cdo.           cotton consumption           atmills. <tr< td=""><td>555553553504522355 555355355 555355355 553355 55335 55335 55335 55335 55335 55335 55335 55335 55335 55335 55335 55355 55555 55555 55555 55555 55555 55555 5555</td><td>5553535535535535535533333043312555325 335533553333304433125553325 333304433125553255332553325533255332553325533</td><td>2 2 2 5 1 3</td><td>80 74 87 84 90 90 90 90 90 90 90 61 65 65 65 61 61 75 75 65 65 61 90 61 90 61 90 61 91 91 91 91 91 91 91 91 91 91 91 91 91</td></tr<>	555553553504522355 555355355 555355355 553355 55335 55335 55335 55335 55335 55335 55335 55335 55335 55335 55335 55355 55555 55555 55555 55555 55555 55555 5555	5553535535535535535533333043312555325 335533553333304433125553325 333304433125553255332553325533255332553325533	2 2 2 5 1 3	80 74 87 84 90 90 90 90 90 90 90 61 65 65 65 61 61 75 75 65 65 61 90 61 90 61 90 61 91 91 91 91 91 91 91 91 91 91 91 91 91
<i></i>	£ 140 103 01 1001103		ľ	Ů		. ~

A. TIMING AND CONFORMITY MEASURES

#### B. SUMMARY OF TIMING AND CONFORMITY 1

	Number
Number of activity turns	109 91
Leads, 4 or more months	14 64 (29)
Coincidences	(11) (31)
Lags, 4 or more months.	13 +.08

Only turns beginning with the trough in activity corresponding to the trough of the 1948-49 recession have been used in this analysis. Conformity and timing measures are based on inverted timing.
First major peak to final major trough.
No activity turns.
No 1948-49 cycle.

Source: Line 1: Facts for Industry, series M51E, Department of Commerce; lines 2, 3: Facts for Indus-try series M51N; lines 4, 5: data from National Electrical Manufacturers Association, with restrictions on publishing; lines 6, 7, 8: Facts for Industry series M51H; lines 0-15, 17-19, 21, 22: Survey of Current Busi-ness, various issues and supplements; line 16: National Association of Hosiery Manufacturers, annual re-ports; line 20: Textile Organon, various issues; line 23: Facts for Industry, series FF1, 14A, line 24: various issues of Textile H-Lights, American Cotton Manufacturers Association (data for cotton consumption at mills from Survey of Current Business); line 25: data from National Federation of Textile Manufacturers, with restrictions on publication.

Abramovitz' investigations established for the prewar period the hypothesis that staples made to stock would tend to conform to activity cycles in an inverted fashion, but that the tendency for stocks to move in the opposite direction from sales and output would weaken as the phase increased in duration. This length-of-phase hypothesis was demonstrated by an analysis of timing in inventory turns compared, on an inverted basis, to the timing of turns in corresponding indicators of manufacturing activity. When leads and lags in the inverted stock series were classified according to the length of their comparable activity phases, it was found that the percentage of leads

#### 66 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

in the stock turns increased as the duration of the phase increased; the average timing changed from a lag for phases of 12 months and under, to a lead which became progressively larger as the activity phase lengthened beyond 12 months.¹⁰

When the same test was performed for the postwar data, the hypothesis was sustained both for expansions and contractions, with some indications that the postwar turns occurred slightly earlier. Sufficient observations for analysis could be made for only two categories of phase duration, 12 months and under, and 13 to 24 months. For these two categories leads of 1 or more months occurred in 30 and 50 percent of total timing comparisons, in contrast to 19 and 43 percent for similar prewar timing computations. The average postwar timing (inverted) was a lag of 1.2 months for activity phases of 12 months and under, and a lead of 1.05 months for those of 13 to 24 months, compared with average prewar lags of 6.1 months and 0.7 month, respectively.¹¹

## TOTAL FINISHED GOODS INVENTORIES

For the period from 1945 to 1958, it is clear (see chart 10) that total finished stocks conform positively to sales cycles, lagging by a number of months at most turns. During the relatively short sales contraction phases lags in the inventory series make for essentially inverted timing, but during the longer expansion phases the two series move in the same direction in most months. It is for this reason that the timing comparisons in table 22 are made on a positive, rather than an inverted basis.

## TIMING AND CONFORMITY OF STOCKS: COMPARISONS WITH SALES AND REFERENCE TURNS

Timing comparisons for finished goods inventory turns are shown in table 22 for the comprehensive and the nine-industry series, along with measures indicating the degree to which cyclical movements in finished goods conform to cycles in the sales series. The total manufacturing and durables series conformed positively to major turns in sales activity in 1948-49, 1953-54, and 1957-58, but displayed long lags at five of the six turns (table 22 and chart 10). It is interesting that the nondurables inventory series shows a longer lag than the durables series at only one of the four comparisons; the lag is actually shorter on two occasions. Zarnowitz' finding that the practice of producing to stock is much more prevalent among nondurables would lead us to expect significantly longer lags in this series at all turns. There is no direct evidence to explain the observed behavior. It is possible that it may reflect the fact that nondurables sales do not fluctuate as vigorously as those of durables. If inverted movements in staples made to stock are regarded as largely due to unanticipated changes in sales, it is to be expected that the duration of inverted behavior (i.e., the duration of the lag of inventories behind sales) would be influenced by the extent to which sales fluctuate.

¹⁰ Ibid., pp. 249-256. ¹¹ Ibid., p. 253.



#### CHART 10

FINISHED-GOODS INVENTORIES AND SALES: TOTAL MANUFACTURING, DURABLE-AND NONDURABLE-GOODS INDUSTRIES, 1946-58

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated inventory cycles; circles, of undeflated cycles. All sales data are undeflated.

Source: Department of Commerce. Data deflated by the author-

 TABLE 22.—Timing and conformity of manufacturers' finished goods inventories at sales turns

Industry .		Lead (-) or lag (+) of inventories to sales, months, in zones associated with business cy reference turns					
		1949 trough	1953 peak	1954 trough	1957 peak	1958 trough	
Total, manufacturing. Durable goods industries, total. Nondurable goods industries, total. Primary metals. Transportation equipment (including motor vehicles). Stone, clay, and glass Food and beverages. Paper. Chemical. Petroleum and coal. Rubber.	+9 +7 +11 +7 +8 -21 +10 \$+20 \$+20 \$+20 \$+20 +14 +7 +13	+9 +11 +8 *20 +9 +9 +9 +9 *21 *20 *15	$ \begin{array}{c} +8 \\ +5 \\ (3) \\ +5 \\ +3 \\ (2) \\ -3 \\ (4) \\ (2) \\ +12 \\ +2 \end{array} $	$\begin{array}{c} -1 \\ -1 \\ (1) \\ +13 \\ +15 \\ +5 \\ (2) \\ +2 \\ (4) \\ (3) \\ +6 \\ +7 \end{array}$	+12 +12 +12 	+9 +9 +6 	

#### A. TIMING MEASURES 1

B, SUMMARI OF TIMING AND CONFORMITI						
Timing and conformity comparisons	9-industry series	4 durable goods in- dustries	5 nondurable goods in- dustries			
Number of comparisons at business cycle turns Matching inventory turns Percentage of matching inventory turns Leads more than 3 months Rough coincidences Leads 3 months or less Coincidences Lags months or less Lags more than 3 months	34 30 88 1 4 (1) (0) (3) 25	16 14 88 1 1 (0) (0) (1) 12	18 16 89 0 3 (1) (0) (0) 13			

SUMMARY OF STAINS AND CONFORMERY

¹ Inventory series have been deflated to 1956, 1957-58 turns based on undeflated data. Timing comparisons are based on a positive relationship between cyclical movements. Timing of inventory turns for "Korean" cycle is not shown since there were no inventory that could be matched with sales turns related to this episode. See table 23 for timing comparisons of inventory turns with Korean reference turns.

Sales turn occurs, but no matching inventory turn.
 Timing relationship is completely inverted: turn is coincident with subsequent sales turn.

4 No turn occurs in either sales or inventories.

Source: Based on material from Department of Commerce.

For the Korean period it is difficult to measure the timing of manufacturing and total durables inventory movements because the related sales series show only a period of retardation rather than actual decline.¹² If, however, sales turns were marked from diffusion index series showing accumulated net increases (total manufactures sales series: peak February 1951, trough February 1952; total durables sales series: peak February 1951, trough November 1951), we would note inverted inventory movements throughout the contraction periods.

When we turn to the industry measures, the major impression is the consistent tendency for finished-stocks series to lag behind sales turns; lags are noted at 28 of the 30 possible comparisons. The duration of these inverted movements varies considerably, however, particularly among the nondurables. The five nondurables series

¹³ Sales failed to contract because expanding defense shipments in key durable-goods industries offset declines elsewhere in the economy. If diffusion indexes based on durable-goods industry group-sales volume (quarterly) are constructed, the behavior of the aggregate sales series appears in a different light. The number of industries with expanding sales fails sharply from 90 percent in the first quarter of 1951 to 46 percent in the second, and remains well below the 50-percent level through the first quarter of 1952. As would be expected, the contraction is more pronounced in the nondurables group, but the diffusion indexes of the durables seles series show a contraction to below 50 percent during the second and third quarters of 1951 and no marked expansion until the first quarter of 1952.

available for study give a picture of erratic behavior, with long lags at the first major turns, and short lags, roughly coincident turns, or nonconforming movements at the last major turns.

Timing comparisons with reference turns are not significantly different from those with sales turns (table 23). Relatively earlier timing at business cycle peaks than troughs is noted for 22 of the 30 comparisons of industry series with business cycle reference turns, but the tendency is not well established in the comprehensive series.

It may be concluded from this analysis that total finished-goods stocks tend to lag behind sales and reference turns with considerable The completely inverted timing noted in the variation in timing. staple made-to-stock series is not observed in the total finished-stock series. Lacking additional evidence, we must conclude that the staple made-to-stock inventories contribute a strong tendency toward inverted behavior, but that the other categories of finished goods move more readily with sales, thereby offsetting inverted tendencies to a very significant extent. The net result is the strongly lagged timing which has been observed.

TABLE 23.—Timing and conformity of manufacturers' finished-goods inventories at reference turns

	Lead (-) or lag (+) in months							
Industry .	Business cycle		Korean cycle		Business cycles			
	Peak, Novem- ber 1948	Trough, Octo- ber 1949	Peak, Febru- ary 1951	Trough, June 1952	Peak, July 1953	Trough, August 1954	Peak, July 1957	Trough, April 1958
Total manufacturing. Durable goods industries, total. Primary metals. Machinery (including electrical). Transportation equipment (includ- ing motor vehicles) Stone, clay, and glass. Food and beverages. Paper. Chemical. Petroleum and coal. Rubber.	$\begin{array}{c} +7 \\ +8 \\ +6 \\ +8 \\ +10 \\ -12 \\ +9 \\ +14 \\ +18 \\ +9 \\ +7 \\ +8 \end{array}$	$+11 \\ +11 \\ +11 \\ +20 \\ +11 \\ +11 \\ +12 \\ +18 \\ +16 \\ +11 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +11 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 \\ +18 $	+14  (2)  +14  (3)  +12  (2)  +13  (3)  (4)  (5)  (5)  (6)  (7)  (7)  (7)  (8)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)  (7)	+4 (3) +10 (2) (3) +15 (3) (3) (3) (3) (3) (3) (3) (3) (3) (3)	+8 +5 (i) +9 +4 +3 (i) +6 (i) +6 (i) +2 +2 +2 +2 +2 +2 +2 +2 +2 +2 +2 +2 +2	$\begin{array}{c} +1 \\ +1 \\ (^{2}) \\ +13 \\ +17 \\ +7 \\ (^{2}) \\ +4 \\ (^{3}) \\ (^{2}) \\ (^{2}) \\ +7 \\ +9 \end{array}$	+6 +6 +6 	+8 +9 +5

A. TIMING MEASURES 1

B. SUMMARY OF TIMING AND CONFORMITY

	9 industry series		4 durab indu	le-goods stries	5 nondurable-goods industries	
Timing and conformity comparisons	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean
Number of comparisons Matching inventory turns Percentage of matching inventory turns Leads more than 3 months. Rough coincidences.	54 34 63	36 30 83 1 2	24 16 67	16 14 88 1	30 18 60	20 16 80 0
Leads 3 months or less Coincidences Lags 3 months or less Lags more than 3 months		(0) (0) (2) 27		(0) (0) (1) 12		(0) (0) (1) 15

¹ Inventory series have been defiated to 1956; 1957-58 turns based on undefiated data. ³ Sales turn occurs but no matching inventory turn.

Source: Based on material from Department of Commerce.

## THE STAPLE, MADE-TO-STOCK INVENTORY INVESTMENT SERIES

In studying finished-goods investment, once again we turn first to the important staple made-to-stock category before proceeding to the analysis of total finished-goods investment behavior as revealed in the Department of Commerce data.

In his study, Abramovitz used the NBER technique of dividing each activity cycle into nine separate stages. Levels of stocks were measured at each stage and inventory investment was computed by measuring the change per month between stages with eight intervals occurring in each cycle (four during expansion, four during contraction).

For the postwar period changes in these stocks have been computed on a strictly chronological basis (for each calendar quarter). The timing of each peak or trough (i.e., the lag from the beginning of the activity phase to the turn in the investment series) has been computed as a percentage of the duration of the activity phase during which it occurs. To provide a framework as similar to Abramovitz' as possible, the resulting timing measures have been arranged so that each expansion and contraction in activity is divided into four parts called quartile intervals. To time inventory investment, frequency distributions have been prepared, showing the proportion of inventory investment peaks occurring during each quartile interval of activitycontraction phases, the proportion of troughs occurring during each quartile interval of activity-expansion phases, and the percentage of all turns occurring in the first, second, third, and fourth quartile intervals of their respective activity phases (table 24).

TABLE 24.—Distribution of timing	of inventory investment turns (inverted) by quartile
intervals of activity	cycles, 25 finished goods, 1947-55

Investment turns	Percent	Mean length of			
	lst quartile interval	2d quartile interval	3d quartile interval	4th quartile interval	phase (months)
Investment peaks, all activity contractions. Cumulated percentage. Investment troughs, all activity expansions. Cumulated percentage. Investment peaks or troughs, all activity phases. Cumulated percentage. Investment peaks: 1948-49 contraction phases. 1951-52 contraction phases. 1953-54 contraction phases. Investment troughs: 1949-51 expansion phases	21 (21) 33 (33) 25 (25) 13 19 9 40 52 7	26 (47) 19 (52) 21 (46) 26 19 27 27 0 47	21 (68) 33 (85) 25 (71) 30 14 20 29 40	32 (100) 14 (100) 24 (100) 30 48 13 13 19 7	14 12 11 20 15

NOTE.—The above analysis is based upon 93 observations of timing of finished-goods inventory investment turns. With only 5 exceptions all investment peaks occurred during activity contractions and all investment troughs during activity expansions. The exceptions may not be regarded as significant departures from this typical timing relationship: 4 investment turns occurred from 1 to 3 months prior to the activity phase with which an inverted comparison could be made. These turns were classified as falling within the 1st quartile interval of the subsequent activity phase. The other exception was an investment peak which lagged the corresponding activity trough by 1 month. This turn was classified as falling within the 4th quartile interval of the preceding activity phase.

Source: See table 25.

The measures show that almost half (46 percent) of the investment series reached peaks or troughs by the middle of their respective activity phases (measured on an inverted basis), and well over twothirds (71 percent) during the first three quartile intervals. The results are not greatly different for frequency distributions made separately for expansions and contractions: the expansions showed only a slightly higher percentage (52 percent) of turns occurring during the first two quartiles, and the contractions, a slightly lower percentage (46 percent) during the same period.

Such measures of the relative timing of peaks and troughs are, of course, not completely comparable with medians of average rates per month from stage to stage, but they would seem to bear out Abramovitz' general conclusion that inventory investment in this class of stocks which moves against the tide of activity expansion or contraction levels off or declines in the second half, and certainly by the fourth quarter, of the phase.

Despite tendencies generally consistent with Abramovitz' findings, these measures provide some evidence of earlier turns in activity cycle phases than existed in the prewar period. It will be noted that a fourth of the peaks or troughs in the inverted inventory investment movements occur during the first quartile interval of all activity phases. This is in sharp contrast to the typical patterns shown by Abramovitz, in which the inverted movement of inventory investment has either not yet begun or has scarcely begun during the early stages of the phase.¹³

## INVENTORY INVESTMENT AND RATES OF CHANGE IN ACTIVITY

In the above discussion the timing of made-to-stock inventory investment was related to turns in activity proper. Turning now to the timing of investment and peaks or troughs in activity rates of change, it is reasonable to suppose on a priori grounds that the relationship will be an inverted one. Very abrupt changes in the flow of shipments would be expected to result in abrupt changes in madeto-stock goods and could easily lead to coincident inverted timing of turns in rates of activity and inventory investment change. There are situations, however, in which the inverted turns in invest-

There are situations, however, in which the inverted turns in investment might lag or lead, as well as be coincident with turns in rates of change in shipments. Suppose, for example, that increases or decreases in the level of shipments take place very gradually and that manufacturers are strongly disposed to retain past rates of production. Under these conditions fluctuations in rates of change in shipments might play a role secondary to the increasing discrepancy between monthly shipments and output, with the result that the peak or trough in inventory investment would occur late in the phase (when the discrepancy reached a maximum), very probably lagging behind the peak or trough in rates of change in activity. On the other hand, it could be assumed that manufacturers are attempting to keep as tight a rein on finished-goods inventories as possible. They may allow only such inverted movements in stocks to occur as are absolutely necessary to meet the requirement that output be maintained at a high enough level during recession to avoid the loss of valuable workers through excessive layoffs, and at a low enough level during

[&]quot; Ibid., pp. 413-414.

expansion to minimize costly overtime work. Under such conditions large changes in inventories would occur principally as a result of inability to anticipate changes in shipments. Peaks or troughs in inventory investment might occur coincidently (on an inverted basis) with those in rates of change in shipments if the latter were large and abrupt. But if the efforts to control stock levels were partially successful, it is quite possible that the inverted inventory investment movement would be checked before the movement in shipment rates of change had reached its peak or trough, with the result that peaks or troughs (inverted) in inventory investment would lead those in activity rates of change.

In chart 11 a quarter-by-quarter record of the incidence of peaks and troughs in inventory investment and rates of change in activity is presented for the 25-commodity sample. It will be noted that peaks or troughs in investment tend to occur in clusters and that there is a well-developed inverted relationship between these turns and those in rates of change in activity (which also tend to cluster). Concentrations of peaks or troughs in inventory investment occur either coincidently or lead (on an inverted basis) those in rates of change in activity.

This tendency was also apparent when the commodity series were examined individually. The results of all timing comparisons for the 25 series are summarized below. Investment peaks and troughs led or turned coincidently with those in rates of change in activity for all but 16 percent of these comparisons, leads (41 percent) and coincident turns (43 percent) being about equally divided. The tendency was well established for both expansions and contractions.

	Leads	Coincidences	Lags	Total
Timing at peaks in rate of change in activity:				
Number of investment troughs compared.	15	17	10	42
Percent of investment troughs compared	36	40	24	100
Timing at troughs in rate of change in activity;				
Number of investment peaks compared	28	28	7	63
Percent of investment peaks compared	44.5	44.5	11	100
Timing at all turns in rate of change in activity:				
compared	43	45	17	105
Percent of investment turns (inverted)	10			
compared	41	43	16	100

No directly comparable study has been made for the interwar period but Abramovitz' stage-by-stage analysis of rates of change in stocks and in activity during activity cycles provides a basis for examining this timing relationship during the earlier period. Abramovitz has presented summary statements for median rates of change for all expansions and contractions observed regardless of duration, and also for expansions or contractions of 12 months and under, 13 to 24 months, 25 to 36 months, and over 36 months.

Most relevant to this discussion are his profiles for activity phases of 12 months and under, and 13 to 24 months (our postwar data show almost no activity phases longer than 2 years in duration). Patterns of rates of change for these two periods of activity phase duration show quite clearly that turns in inventory investment (i.e., CHART 11



FINISHED-GOODS INVENTORY INVESTMENT AND CHANGE IN MANUFACTURING ACTIVITY: 25 COMMODITIES

median rates of change in stocks) occurred coincidently with or lagged behind (on an inverted basis) turns in rates of change in activity. There were no leads: ¹⁴

	Peak: Rate of change in activity	Trough: Inventory investment
Expansions: Phases 12 months or less Phases 13 to 24 months	Stages IV-V Stages I-II	Stages IV–V. Stages III–IV.
	Trough: Rate of change in activity	Peak: Inventory investment
Contractions: Phases 12 months or less Phases 13 to 24 months	Stages VII-VIII Stages VI-VII	Stages VIII–IX. Stages VI–VII.

Here, once again, is evidence that inventory investment has turned earlier in the postwar period. Abramovitz' prewar data show that timing of inverted investment turns was coincident or lagging. The postwar data for this class of stock show a marked tendency for peaks

Shaded areas represent business contractions; unshaded areas, expansions. Source: Compiled from seasonally adjusted industry data. For details on commodity series see table 21.

¹⁴ This summary of timing is based on charts 78 and 79, ibid., pp. 413, 414. According to the NBER techniques used by Abramovitz, stages I and LX are the terminal troughs of a cycle, stage V is the peak, and stages II to IV and VI to VIII are intermediate stages of the expansion and contraction phases.

## 74 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

and troughs in investment to lead or turn coincidently with troughs and peaks in rates of change in activity.

## RELATION BETWEEN STAPLES MADE TO STOCK AND TOTAL MANUFAC-TURERS' FINISHED GOODS

The discussion of inventory investment thus far has dealt in some detail with staples made to stock. It has already been shown that this is the largest finished-goods category (comprising perhaps as much as one-half of finished stocks). Moreover, it is probable that cyclical fluctuations in these stocks display the greatest relative amplitude. Staple made-to-stock inventories act as buffers, allowing the firm to operate without a complete synchronization of output and shipments. Stocks are permitted to rise or fall as changes in demand bring disturbances in this balance and as, subsequently, such im-balances are corrected. In contrast, made-to-order stocks are merely awaiting shipment. In most instances changes in their level can be little more than a reflection of ease or difficulty in making shipment or of changes in customers' preferences as to actual date of delivery. Finally, perishable goods, by definition, can be neither large nor cyclically sensitive, and supply-dominated stocks can hardly be expected to fluctuate cyclically. In view of the evident size and significance of staple made-to-stock inventories, the question now arises as to whether there might be a generalized investment pattern among the made-to-stock commodity series which would closely resemble the pattern in the aggregative Department of Commerce series. In order to test this hypothesis a diffusion index of the commodity series was constructed by computing quarterly the percent of series in which inventory investment increased (i.e., increases in stocks were larger, or decreases smaller, than in the preceding period) and accumulating the net percentage increases (percentage of series expanding minus 50) from quarter to quarter. Such a diffusion index approximates the aggregate behavior of the series included.

The resulting series, graphed in chart 12, shows a striking resemblance to the Department of Commerce series for total manufacturers' finished-goods investment series. The two display almost identical movements, coinciding in their turning points at five of the six turns and differing by only one quarter at the remaining turn.¹⁶

This presents a curious picture: there is ample evidence that the 25 commodity series for which the diffusion index was constructed represent activities in which goods are predominantly produced to stock; yet inventory investment for these series behaved like that for aggregate and durables manufacturing in which a large portion of output is produced to order.

¹³ A strong upward trend may be noted in the diffusion index series, which is in sharp contrast to the trend-free behavior of the series for manufacturers' finished goods. The trend in the former series is not surprising in view of the type of data being observed. In preparing these series, quarterly computations were made of the net percentage of inventory investment series rising and the results accumulated. The average extent of participation in expansion and contraction periods is roughly the same (i.e., the typical quarterly amplitude of movement appears to be about the same for expansions and contractions), but the longer duration of the expansion periods gives the cumulative series an upward trend. Such a trend is impossible, of course, in the series showing quarter-to-quarter change in total manufacturers' finished goods, since disinvestment must be shown below the zero line.

#### CHART 12

FINISHED-GOODS INVESTMENT: TOTAL MANUFACTURING AND CUMULATED NET PERCENTAGE OF INVESTMENT SERIES EXPANDING (25 COMMODITIES)



Dots identify peaks and troughs of specific cycles.

Two explanations of this agreement between the series are possible. The first is that total inventory investment was dominated by changes in the volume of finished goods made to stock; goods produced to order were quickly shipped and their variations played a negligible part in determining overall investment. The second is that inventory investment for made-to-order and made-to-stock finished goods share the same pattern.

The latter explanation is unacceptable; to adopt it is to say that inventories of goods produced to order show their sharpest rate of increase after shipments have begun to fall off and their sharpest rate of decrease after shipments have begun to rise. The statistical evidence is meager, but it does not point toward such behavior: Abramovitz found one made-to-order series, steel sheets, in which inventory investment moved with rates of change in sales, leading turns in sales proper. I have located one postwar series, steel barrels and drums, and here movements were irregular and of small amplitude.

The conclusion is that finished goods made to stock have much the greater amplitude of movement and that they dominate finished goods inventory investment. In short, the pattern and timing of inventory investment in our 25 made-to-stock series agree with that of total manufacturers' finished-goods investment because the movements that we see in the latter are predominantly reflections of movements in the former.

Source: Total manufacturers' finished-goods investment series compiled from Department of Commerce data deflated by the author. For details on commodity series see table 21.

When the timing of inventories proper was compared by a similar means, however, the resemblance was less close. Turns in the cumulative diffusion index of the 25 commodity inventory series coincided with the Department of Commerce manufacturers' series in three of the five comparisons, but lagged by 3 and 6 months at the remaining comparisons.¹⁶ It would seem that the influence of the other categories of finished goods is demonstrated here. As stated previously, it is to be expected that these other stocks will move more promptly with cyclical changes in activity than do the staple made-to-stock inventories, and that they will cause the aggregate finished-goods inventory movements to turn somewhat earlier. The stamp of the made-to-stock group remains, however, dominating the patterns in rates of change in the aggregate finished-goods series.

## TOTAL FINISHED-GOODS INVENTORY INVESTMENT

### INVESTMENT AND RATES OF CHANGE IN SALES

We may now ask whether total finished-goods inventory investment, which moves with similar timing to its staple made-to-stock component during business cycles, has a like tendency to move in an inverted fashion with turns in rates of change in sales.

Chart 13 and table 25 show that, for the period beginning in mid-1947, there is a strong tendency for total finished-goods investment, as well as its durable and nondurable components, to move in an inverted manner in relation to rates of change in sales, but to turn and come into phase at least several months before the turn in the latter series. There are two instances in which the lead is a year or more in duration. The first occurs in third quarter 1949, when the trough in investment leads by 12 months the peak in rate of change in sales.¹⁷ The second occurs during the 1954–58 cycle, when the peak in inventory investment leads the trough in rate of change in sales by 18 months in the total manufacturers' and nondurables series and by 27 months in the investment and sales series are inverted during roughly the first year and a half of the expansion. It is only during the latter part of the expansion that we note departure from the earlier pattern.

¹⁰ If the timing of the Department of Commerce and the commodity series are compared with their respective activity series, the relatively greater inverted tendency of the commodity series becomes even more apparent, for the commodity activity series tended to turn relatively early in the business cycle phase (see footnote 20).

more apparent, for the commodity activity series tended to turn relatively early in the business cycle phase (see footnote 20). If There was a second trough in investment, of even greater magnitude, in third quarter 1950, accompanying the outbreak of the Korean war. This movement was of only one quarter duration, however, and the trough has been marked at the earlier date. On the other hand, it is apparent that the peak in rate of change in sales, marked as third quarter 1950, could not have occurred earlier than second quarter 1950 had there been no such outbreak.



.

** . . . .

CHART 13 FINISHED-GOODS INVENTORY INVESTMENT AND QUARTER-TO-QUARTER CHANGE IN Sales, 1946–58



Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of defiated inventory cycles; circles, of undefiated cycles. All sales data are undefiated.

Source: Department of Commerce. Data deflated by the author.

78

Date of turns in rate of change in sales		Timing of finished-goods investment (inverted)					
Quarter	Year		Total manu- facturing	Durable goods	Nondurable goods		
I III III IV IV IV I I	49 50 51 52 53 55 58	(Trough) ¹ (Peak)	0 -12 0 -6 -3 -9 -21	$ \begin{array}{r} -12 \\ -12 \\ 0 \\ -3 \\ -9 \\ -15 \\ \end{array} $	9 12 +3 6 3 3 18		

TABLE 25.—Timing of finished-goods investment to rate of change in sales

For durable goods, date was 4th quarter 1949.
 For nondurable goods, date was 2d quarter 1951.
 For nondurable goods, date was 4th quarter 1957.

Source: Based on material from Department of Commerce.

#### TIMING OF INVESTMENT DURING BUSINESS CYCLES

The postwar tendency of turns in inventory investment (inverted) to lead or coincide with turns in rates of change in activity provides a basis for generalization regarding the typical timing of investment during business cycles.

There is considerable evidence available relating to the timing characteristics of rates of change in activity. Abramovitz studied the behavior of rates of change in activity during reference cycles for 57 prewar series and noted well-established tendencies: "The rate of growth in output reaches a high point considerably before the end of expansion, a trough considerably before the end of contraction." 18 During the postwar period this same tendency has been quite apparent in rates of change in total manufacturing sales, as may be seen from the following chronology of turns: ¹⁹

	Reference turns (quarterly basis)	Quarter-to-quarter change, total manu- facturing sales
Expansion	IV 1945 to IV 1948.	Peak IV 1947.
Contraction	IV 1948 to IV 1949.	Trough I 1949.
Expansion	IV 1949 to I 1951 1	Peak III 1950.
Contraction	IV 1949 to I 1952 1	Trough III 1951.
Expansion	II 1952 1 to II 1952 1	Peak IV 1952.
Contraction	II 1952 to II 1953.	Trough IV 1953.
Expansion	II 1954 to III 1957.	Peak I 1955.
Contraction	II 1954 to III 1957.	Trough I 1958.

¹ Based on subcycle reference turns. See note 2, ch. 3.

It may be concluded, therefore, that since peaks and troughs in inventory investment tend to lead (on an inverted basis) these earlyoccurring turning points in rates of change in sales, their timing will cause inventory investment to move with the tide of business during

76626-61-pt. I-_7

¹⁸ Abramcvitz, "Inventories and Business Cycles," p. 378. ¹⁹ This tendency is also applicable to rates of change in output. During the postwar period quarter-to-quarter changes in total manufacturers' sales and output have had approximately the same patterns of movement with coincident or near coincident timing. Such data are not available for the prewar period, but Abramovitz studied shipments and output for 10 nonfarm (made-to-stock) commodities and demon-strated the "very close connection between the rates of production and shipment of manufactured staples." Ibid., pp. 248, 250-251.

most, if not all, business cycle phases.²⁰ Accordingly, timing comparisons are made with sales and reference turns on a positive basis in the sections which follow.

## TIMING AND CONFORMITY OF INVESTMENT: COMPARISON WITH SALES TURNS

A somewhat higher conformity was noted for investment than for inventories proper with no significant difference in the conformity of the durables and nondurables series. Of the 42 sales turns in the industry sales series, including turns related to the Korean cycle, 41 (all 20 of the durables and all but 1 of the 22 nondurables) could be matched by investment turns.²¹ Timing varied, showing no consistent tendency to lead or lag. Approximately half of the turns in the individual-industry series (21 of 41) occurred roughly coincidently with sales turns; among the others, lags of more than 3 months (14) were more prominent than leads of more than 3 months (6). The comprehensive series bear out this timing tendency. For total manufactures, timing ranges from a lead of 8 months to a lag of 6 months during the 1948–58 period (see also chart 11).

TABLE	26.— <i>Timing</i>	and	<i>conformity</i>	of	manufacturers'	finished-goods	inventory
	•		investmer	it a	t sales turns		

Lead $(-)$ or lag $(+)$ in months, in zones associated with reference turns								
	1949 t	Kore	an war					
1948 peak	Turns prior to mid-1950	Turns after mid-1950	1951 peak	1952 trough	1953 peak	1954 trough	1957 peak	1958 trough
$+5 \\ -1$	$-4 \\ -2$	$^{+8}_{+10}$	(3) (2)	(2) (2)	$^{+1}_{+1}$	-5 -5	$-8 \\ -2$	+2 +4
$\begin{vmatrix} -1 \\ +5 \\ +2 \end{vmatrix}$	( ³ )-1	+7 +10 +8	(3) +2 (2)	(2) +2 (2)	$(^{3})$ +1 -1	(2) -3 -5	-8	+2
$ \begin{array}{c c} -9 \\ +4 \\ -3 \\ +5 \\ +8 \\ +2 \\ +1 \end{array} $	$\begin{vmatrix} -7 \\ (^3) \\ -5 \\ +1 \\ +7 \\ +4 \\ +1 \end{vmatrix}$	$ \begin{pmatrix} (3) \\ +8 \\ +7 \\ +13 \\ +13 \\ (3) \\ +7 \end{pmatrix} $	(3) +3 (3) +5 (2) (4)	(2) +8 (3) +8 +2 (3) (4)	+1 -2 -8 (*) +2 +10 (*) (*)	$ \begin{array}{c} -2 \\ +4 \\ +7 \\ (*) \\ -2 \\ +2 \\ -11 \end{array} $		
	Lead 1948 peak +5 -1 -1 +5 +2 -9 +4 -3 +5 +2 +2 -9 +4 +5 +2 +2 +5 +2 +2 +5 +2 +5 +2 +5 +2 +5 +5 +2 +5 +5 +5 +5 +5 +5 +5 +5 +5 +5	$\begin{array}{c c} \mbox{Lead} (-) \mbox{ or lag} (\\ \hline & 1948 \\ \hline 1948 \\ peak \\ \hline Turns \\ prior to \\ mid-1930 \\ \hline +5 \\ -1 \\ -2 \\ -1 \\ +5 \\ +2 \\ -1 \\ -9 \\ +4 \\ -3 \\ +2 \\ +1 \\ +1 \\ +1 \\ \end{array}$	$\begin{array}{c c} \mbox{Lead} (-) \mbox{ or lag} (+) \mbox{ in mon} \\ \hline 1948 & 1949 \mbox{ trough} \\ \hline 1948 & Turns & Turns \\ prior to & after \\ mid-1930 & mid-1930 \\ \hline +5 & -4 & +8 \\ -1 & -2 & +10 \\ \hline -1 & -5 & +7 \\ +5 & -1 & -5 \\ +2 & -1 & +8 \\ -3 & -5 & +7 \\ +4 & (3) & +10 \\ +2 & -1 & +8 \\ -3 & +5 & +11 \\ +8 & +7 & +13 \\ +2 & +4 \\ +1 & +1 & +7 \end{array}$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $

A. TIMING MEASURES 1

See footnotes at end of table, p. 81.

²⁰ It may be asked why, if there has been such a marked tendency for inverted turns in staple, made-to-stock investment series to occur early in business cycle contractions and expansions, there was not a greater concentration of turns in investment during the early stages of activity phases (see table 24). The answer appears to be that the sample is composed of series for which activity has shown a pronounced tendency to turn prior to business cycle turns, but which displayed roughly the same timing of movements in activity rates of change as the total manufacturing series. This was verified by analysis of diffusion indexes of activity and of rates of change in activity. The former series led total manufacturers' sales but the latter turned roughly coincidently with rates of change in these sales. ³¹ Timing comparisons presented a special problem for the period following the recession of 1945-49. In five of the nine industry series, troughs in inventory investment occurred prior to the second quarter of 1950, but during the third quarter there were second and deeper troughs which were related to the sudden up-surge in sales at the outbreak of the Korean war. These later troughs which were related to the sudden up-sent as accurate a picture of timing as possible, measures for both troughs have been included. (See table 26.)

26.)

	9 indusi	ry series	4 durab indu	le-goods stries	5 nondurable-goods industries		
Timing and conformity comparisons	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean	
Number of comparisons Matching inventory investment Percentage of matching turns Leads more than 3 months ⁴ Rough coincidences Leads 3 months or less Coincidences Lags 3 months or less Lags more than 3 months ⁵	42 41 98 6 21 (7) (0) (14) 14	34 33 97 6 16 (7) (0) (9) 11	20 20 100 3 11 (5) (0) (6) 6	16 16 100 3 8 (5) (0) (3) 5	22 21 95 3 10 (2) (0) (8) 8	18 17 94 3 8 (2) (0) (6) 6	

## B. SUMMARY OF TIMING AND CONFORMITY

¹ Inventory investment series have been deflated at 1956. 1957-58 turns based on undeflated data.

Inventory investment series have been denated at 1000. 1000. Inventory investment turn occurs, but there is no sales turn. Sales turn occurs but no matching inventory investment turn.

⁴ No turn occurs in either sales or inventory investment.

In the 1949 revival zone, investment troughs prior to mid-1950 have been used in the timing measures when 2 troughs occurred.

Source: Based on material from Department of Commerce.

## TIMING AND CONFORMITY OF INVESTMENT: COMPARISON WITH REFERENCE TURNS

It may be concluded from table 27 that finished-goods investment is very sensitive to cyclical forces. Conformity to reference cycles is very high: of 54 possible comparisons 52 turns could be matched with reference turns.

With overall conformity so high it is not surprising that there is little difference between that of durables and nondurables. It will be noted that investment turns could be matched with reference turns on occasions when no sales turns occurred, indicating that investment was responsive to cyclical forces even when these forces were not sufficiently strong to bring about full-fledged movements in the related sales series.²²

There is one observation that deserves special mention: timing of investment turns shows less variation among the several industry series when compared to reference turns than when compared to sales turns. (See the timing measures in table 26 and table 27 for any given peak or trough.) This suggests further evidence of the relations between inventory investment and rates of change in sales. Analysis of the individual series indicates that turns in rates-of-changein-sales movements tended to occur at about the same time in most of the series during the period studied. Some indirect evidence of this may be seen by observing the comprehensive series in chart 13: timing of turns in the durables and nondurables rates-of-change-insales series is coincident in five comparisons, and varies by only 3 months in the other two. Since turns in rates of change in sales occur at about the same time, comparisons of inventory investment with reference turns which were fixed points in time give roughly the same degree of agreement among the revised series as would timing

²⁷ These "extra" turns may be noted in table 26. In most instances they relate to the "Korean" cycle which has already been discussed, but there is one investment cycle, in the paper products industry, which is related to the 1953-54 business cycle peak and trough.

comparisons with rates of change in the respective sales series. Apparently, this relationship is closer than with movements in the series for sales proper.

TABLE 27.—Timing and conformity of manufacturers' finished-goods inventory investment at reference turns

A. TIMING MEASURES

	Lead (-) or lag (+), in months									
Industry	Busi- ness- cycle peak.	Business cycle trough, October 1949		Korean cycle		Business cycles				
	Nov. 1948	Turns prior to mid- 1950	Turns after mid- 1950	Peak, Feb- ruary 1951	Trough, June 1952	Peak, July 1953	Trough, Aug- ust 1954	Peak, July 1957	Trough, April 1948	
Total manufacturing	+3	-2	+10	+6	-1	+1	-3	14	+1	
Durable-goods industries, total	0	-2	+10	+6	+2	+1	-3	-8	+4	
Nondurable-goods indus- tries, total Primary metals	-6 +6	2 (2)	$^{+10}_{+10}$	+6 +6	$^{-1}_{+2}$	$^{+1}_{+1}$	+3 -3	-14	+1	
Machinery (including elec- trical)	+3	+1	+10	+6	+1	-2	-3			
Transportation equipment (including motor vehicles) Stone, clay, and glass Food and beverages Paper Chemical Petroleum and coal Rubber	$\begin{array}{c} 0 \\ +3 \\ -6 \\ +3 \\ +3 \\ +3 \\ -3 \end{array}$	$ \begin{array}{c c}     -5 \\     (2) \\     -2 \\     -2 \\     +4 \\     +7 \\     (2) \end{array} $	+10 +10 +10 +10 +10 +10 +10 +10 +10	+3 +6 +6 +6 +6 +3 +12	$ \begin{array}{c c} +2 \\ +2 \\ +2 \\ +8 \\ +2 \\ -1 \\ (^2) \end{array} $	-2 -2 +1 +7 +1 +10 (2)	$ \begin{array}{c c} -3 \\ -3 \\ +9 \\ +6 \\ -9 \\ +3 \\ -9 \end{array} $			

#### B. SUMMARY OF TIMING AND CONFORMITY

-	9 indust	ry series	4 durable- goods in-	5 nondurable- goods in-		
Timing and conformity comparisons	All turns	All turns ex- cept Korean	dustries, all turns	dustries, all turns		
Number of comparisons. Matching inventory turns. Percentage of matching turns. Leads more than 3 months ³ . Rough coincidences ³ . Leads, 3 months or less. Coincidences. Lags, 3 months or less. Lags, more than 3 months ³ .	54 52 96 4 30 (11) (11) (13) 18	36 35 97 4 21 (10) (10) (10) 10	24 24 100 1 177 (7) (1) (9) 6	30 28 93 3 13 (4) (0) (9) 12		

Inventory series have been deflated to 1956; 1957-58 turns based on undeflated data.
 No matching inventory investment turn.
 In making comparison with the reference trough of October 1949, investment troughs prior to mid-1950 have been used in the timing measures where 2 troughs occurred.

Source: Based on material from Department of Commerce.

## Amplitude of Durables and Nondurables Investment During CYCLES

Measures of amplitude of movements in finished-goods inventory investment reveal that the relative amplitude of the durables series is greater than that of nondurables. This is readily seen in table 28. Total change in relation to size of stocks, is greater for durables than for nondurables in five of the seven phases examined. Nondurable finished-goods stocks, however, are considerably larger than durables stocks, and total investment movements in the nondurables series are larger in five of the seven phases.

Industry	Peak- trough (1948-49)	Peak- trough (1949-50)	Peak- trough (1950-52)	Peak- trough (1952-53)	Peak- trough (1953-54)	Peak- trough (1954-56)	Peak- trough (1956–58)			
· · · · · · · · · · · · · · · · · · ·		Total change (relative) ¹								
Total manufacturing Durable goods Nondurable goods	-7.72 -7.35 -9.37	+15.95 +19.44 +13.55	13. 51 15. 74 12. 40	+4.92 +6.43 +4.88	$ \begin{array}{r} -5.83 \\ -11.23 \\ -3.56 \end{array} $	+6.93 +10.16 +6.74	-7.60 -7.60 -9.00			
	Change per month (relative) 1									
Total manufacturing Durable goods Nondurable goods	-1.29 82 62	+.66 +.81 +.56	$ \begin{array}{c} -1.13 \\ -1.31 \\ -1.38 \end{array} $	+.41 +.71 +.33	65 -1. 25 24	+. 29 +. 34 +. 37	32 36 38			
		Т	otal change	(millions	of dollars)	2	<u> </u>			
Total manufacturing Durable goods Nondurable goods	835 315 561	+1, 321 +908 +923	-1, 703 -815 -920	+666 +360 +379	790 668 278	+1, 190 +856 +602	-1, 436 -700 -882			
Change per month (millions of dollars) ²										
Total manufacturing Durable goods Nondurable goods	-139 -52 -37	+76 +38 +38	$-142 \\ -68 \\ -102$	+55 +40 +25	88 74 18	+49 +29 +33	60 33 37			

#### TABLE 28.—Amplitude of change in finished-goods inventory investment cycles, by phase, 1948-58

¹ Expressed as percent of the mean level of inventories during the phase. Mean level is an average of beginning and ending level of inventories of the terminal quarters of the phase. ³ All 1948-54 data have been deflated (1947 dollars). Measures for the two most recent phases are based on undeflated (book value) data.

Source: Based on material from Department of Commerce.

## SUMMARY

Abramovitz emphasized the fact that finished stocks are of several different types, each possessing quite different cyclical characteristics. Analysis of postwar composition reveals that finished goods whose production cycles are governed by supply comprise less than one-third of the total. The major part, from 41 to 51 percent of the total, is made up of staple made-to-stock inventories. Goods made to order are estimated at 15 to 25 percent of the total. According to Abramovitz, staple made-to-stock goods may be expected to move in an inverted fashion relative to sales, made-to-order stocks will move in positive conformity, and the remainder will be cyclically insensitive or erratic.

Postwar finished-goods behavior was studied by analyzing two sets of inventory data. The first was composed of data for 25 staple, made-to-stock commodities, representing the largest component of finished goods. The second was composed of Department of Commerce industry data for nine industry groups and for comprehensive groupings. Among the commodity series, finished stocks were found to move in a strongly inverted fashion relative to sales or output, but with a well-developed tendency to turn and come into phase as the duration of the activity phase increased. The Department of Commerce industry data revealed a consistent tendency for stocks to lag behind sales and reference turns, but movements were rarely of

#### 84 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

the completely inverted sort observed in the commodity series. The conclusion drawn was that the staple made-to-stock inventories contribute a strong tendency toward inverted behavior, but that other categories offset inverted tendencies to a significant degree.

Study of inventory investment data for the 25 commodities revealed that investment moved in an inverted fashion relative to rates of change in sales, but that peaks (troughs) in inventory investment led or turned coincidently with troughs (peaks) in rates of change in activity. Since peaks and troughs in rates of change in activity tend to occur well before the end of each business cycle phase, the inverted movements in inventory investment may be expected to terminate relatively early in the business cycle phase. Analysis of finishedgoods inventory investment in the Department of Commerce series shows this to be the case. Total manufacturers' investment turns roughly coincidently with business cycle turns or moves in an inverted fashion (i.e., lags) for only a few months in the early part of a business cycle phase.

## CHAPTER 6

# GOODS-IN-PROCESS INVENTORIES AND INVENTORY INVESTMENT

For his study of goods-in-process behavior, Abramovitz once again divided the category into its major components and sought to determine the typical behavior of each. In the present chapter, a discussion of Abramovitz' system of classification and the theory which he developed is followed by an observation of the behavior of goods-in-process stocks and investment as revealed in Department of Commerce data. A third section develops possible reasons for observed differences in actual and hypothesized behavior and presents a revised theory.

Amplitude of movements in purchased materials, goods-in-process, and finished-goods investment are compared in chapter 7. As noted in this chapter, however, significant movements in goods-in-process investment are found only in the durables series. Nondurables in-process investment moves erratically and with little amplitude.

## ABRAMOVITZ' THEORY OF GOODS-IN-PROCESS BEHAVIOR ¹

Abramovitz classified these stocks under two heads, those held within and those held between stages of manufacture. On the basis of an analysis of production processes, and from information in a special Federal Trade Commission survey for 1939, relating to size of goodsin-process stocks held by specific industries, he estimated that 37.7 percent of goods-in-process stocks were held by continuous process industries which characteristically hold them within stages; 36.1 percent by discontinuous process industries with large holdings of in-process stocks between stages; and 2 6.2 percent by industries with a mixture of continuous and discontinuous processes. He pointed out that if inventories in the mixed industries were about equally divided between stocks held within and stocks held between stages, roughly half of all goods in process would be held in each.

Goods-in-process behavior in the continuous process industries was determined by several hypothetical models which showed that the timing of turns in these stocks is dependent upon the timing pattern of inputs and the length of the production period. On the basis of this analysis Abramovitz concludes that: "In such industries stocks of goods in process cannot lag behind production. On the contrary, they The lead, however, cannot exceed an interval are likely to lead. equal to a production period; that is, it cannot exceed the time elapsing between the moment work is begun upon a prospective unit of output in a manufacturing establishment and the time it is ready for delivery."²

Regarding the goods in process held by discontinuous and mixed industries, he points out that "In other [than continuous] manu-

¹ The following account is based upon Abramovitz, "Inventories and Business Cycles, with Special Reference to Manufacturers Inventories," New York, NBER, 1950, pp. 160–177, 380–385. ³ Ibid., p. 380. Abramovitz estimates the average length of the production period in American industry to be 25 days, pp. 171–174.

facturing industries, the relation need not be rigid; yet even here there is a bias in favor of a positive relation between production and goods in process. For only surplus stocks between stages can move inversely to output. The stocks within the various stages of discontinuous industries must still move together with activity in their respective stages. Since activity in these stages is closely bound together, so must output and goods in process within the various stages. Finally, it must be remembered that surplus stocks between stages need not move inversely to output; they only may do so.

"* * * there is, in fact, a very powerful set of forces impelling (all) goods in process as here defined, to move together with output in manufacturing as a whole."³

The analysis of goods-in-process investment follows along similar Emphasis is given to the behavior of in-process stocks within lines. For these goods Abramovitz finds that investment will not stages. lag behind the rate of increase in output and is likely to lead, but not by more than one production period.⁴ He holds that goods "between stages" in discontinouous industries may or may not act in the same fashion, "when they do not, the effect is probably to cause goods in process in the aggregate to respond to changes in activity somewhat more tardily than they otherwise would. Hence investment in goods in process as a whole is likely to lead the rate of growth of production by less than investment in continuous industries does. It may even lag by a short interval. It seems best, therefore, to say merely that investment in goods in process and the rate of growth of output turn at nearly the same time." 5

## **GOODS-IN-PROCESS** INVENTORIES

Before examining the Department of Commerce data on goods-inprocess inventories, it should be noted that the problems of deflation here are much greater than those encountered with purchasedmaterials and finished-goods data. Goods in process are not bought and sold and there is no market price for them. Their value for accounting purposes is determined by cost-accounting techniques which estimate the value of labor and overhead applied to the purchased materials. For the inventory data there are, therefore, no price indexes from which deflators can be constructed, nor any clues to the composition of goods in process (i.e., the proportions which are are in early, middle, and late stages of production) which would permit assumptions as to the value added.

These restrictions reduce the effectiveness of the analysis, but do not preclude learning from the data. Crude deflators can be devised which will permit us to observe whether or not cyclical characteristics found in the undeflated series are sufficiently well stamped upon the data to remain apparent after making an arbitrary allowance for possible price effect. Furthermore, the undeflated data may be examined and, where there are a large number of observations, generalizations may be made concerning well-established characteristics.

² Ibid., p. 165. ⁴ Ibid., p. 387. ⁵ Ibid., p. 387.

Comprehensive goods in process series were deflated by using an average of the purchased-materials and finished-goods deflators. It was not deemed worthwhile to prepare deflators for the individual industry series, but timing and conformity measures of the undeflated series were prepared, as well as a summary of the timing sequence of their turns when related to turns in undeflated purchased materials and finished goods.

## MAJOR PATTERNS OF MOVEMENT IN STOCKS

The undeflated goods-in-process inventory series for total manufacturing shows three well-marked cyclical movements from 1945 to 1958 (chart 14). There is, however, no movement in the series which would indicate conformity to the cyclical forces accompanying the Korean period.

Deflation alters the total manufacturing series significantly in only one phase. The timing of the first cycle peak occurs 14 months earlier in the deflated than in the corresponding undeflated series (table 29). The undeflated-durables series which has the same pattern of movement as total manufacturing is affected in the same way.

In the undeflated nondurables series the cyclical patterns are less clearly defined. The pattern of the first cycle is approximately the same as that of the undeflated total manufacturing series, but movements are smaller. From June 1951 to the second peak in June 1953 a series of irregular movements occurs, and the contraction which follows is of very small amplitude. When the nondurables data are deflated, the resulting series moves so irregularly, prior to mid-1953, that no cycle turns can be marked. The contraction of 1953-54 remains, however, with timing approximately the same as in the undeflated series.



## CHART 14 GOODS-IN-PROCESS INVENTORIES AND SALES: TOTAL MANUFACTURING, DURABLE-AND NONDURABLE-GOODS INDUSTRIES, 1946-58

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated inventory cycles; circles, of undeflated cycles. All sales data are undeflated.

Source: Department of Commerce. Data deflated by the author.

#### INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION 89

# TABLE 29.—Timing and conformity of manufacturers' goods-in-process inventories at output and sales turns

•

	Lea	d (—) or	lag (+	), in mor referen	nths, in ce turn	zones as	sociate	d with
Industry	1948	1949	Kore	an war	1953	1954	1957	1958
	peak	trough	1951 peak	1952 trough	peak	trough	peak	trough
		Tin	aing of	invento	ries to c	output ti	ırns	
Denated inventory data: Total manufacturing Durable-goods industries, total Nondurable-goods industries, total Undeflated inventory data:	-10 -10 ( ² )	+1 +4 (²)	(1) (1) (2)	(1) (1) (2)	$\begin{vmatrix} 0 \\ +1 \\ -1 \end{vmatrix}$	+5 +5 +9		
Total manufacturing. Durable-goods industries, total. Nondurable-goods industries, total Industry series (undeflated):	+4 +5 +5	+1 +4 +6	(1) (1) (2)	(1) (1) (2)	+1 +1 +1 -1	+5 +5 +9	+1 0 +5	+7 +7 +7
Primary metals Machinery (including electrical) Transportation equipment (including	$^{+4}_{-1}$	+7 +7	(1) (1)	(3) (1)	+4 -8	0 +10		
motor vehicles) Stone, clay, and glass Food and beverages. Paper Chemical. Petroleum and coal. Rubber	-6 +3 () +6 -13 +5 ()	+2 +9 (1) +4 +9 +7 +7 +7	(1) (2) (1) + 8 + 8 (2) + 20	(1) (2) (1) +13 -10 (3) (3) (3)	$ \begin{array}{c} 0 \\ +2 \\ (^{1}) \\ (^{3}) \\ +1 \\ (^{2}) \\ (^{2}) \end{array} $	0 + 16 (1) (3) (3) + 7 (3) 0		
		Ti	ming o	f invento	ories to	sales tur	ns	
Deflated inventory data: Total manufacturing Durable-goods industries, total Nondurable-goods industries, total Undeflated inventory data:	12 15 (²)	0 +4 (²)	99 99 99	() () () ()	0 +1 (*)	-1 -1 ( ³ )		
Total manufacturing. Durable-goods industries, total Nondurable-goods industries, total Industry series:	$^{+2}_{0}_{+5}$	$^{0}_{+3}_{-2}$	(1) (1) (1)	(1) (1) (1)	+1 +1 ( ³ )	-1 -1 ( ³ )	$^{+8}_{+7}_{+12}$	+8 +7 +7
Primary metals Machinery (including electrical) Transportation equipment (including	$^{+2}_{-12}$	$^{+7}_{+1}$	(2) (1)	(2) (1)	$^{+2}_{-6}$	$^{-1}_{+5}$		
motor vehicles) Stone, clay, and glass Pood and beverages Paper Chemical Petroleum and coal Rubber	-5 +3 (2) +6 +6 +5 (4)	+2 +3 (2) +2 +9 +4 +3	(1) (2) (1) +8 -5 (3) +8	(1) (2) (1) +7 -11 (3) (1)	$ \begin{array}{c} 0 \\ +3 \\ (^{2}) \\ (^{3}) \\ +2 \\ (^{2}) \\ (^{2}) \\ (^{2}) \end{array} $	-2 + 16 (2) (3) (4) (7) (7) (7) (7) (7) (7) (7) (7) (7) (7		

A. TIMING MEASURES

B. SUMMARY OF CONFORMITY TO OUTPUT

	9 industry series		4 du in	rable-goods dustries	5 nondurable- goods industries		
	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean	
Number of comparisons Matching inventory turns Percentage of matching turns	39 31 79	29 26 90	20 16 80	16 16 100	19 15 79	13 10 77	

No turn occurs in either activity or inventories.
 Activity turn occurs, but no matching inventory turn.
 Inventory turn occurs, but there is no activity turn.
 Not available.

Source: Based on material from Department of Commerce.

## TIMING AND CONFORMITY OF STOCKS: COMPARISON WITH OUTPUT AND SALES TURNS

In table 29 timing and conformity measures are presented for the undeflated and deflated comprehensive series and for the undeflated industry series. Timing comparisons were made with turns in output as well as sales to facilitate subsequent discussion of the relations between stocks and the rate of output.

The movements described above conform to all movements in output and sales for the total manufacturing and durables comprehensive series. Timing is virtually coincident for all turns in the undeflated series prior to the revival in 1958. Deflation alters the timing no more than 1 month, except for the 1948 peak when the turn in stocks occurs slightly over a year earlier.

In the nondurables undeflated comprehensive series the turns conform to the output and sales turns associated with the business cycle, but not to those associated with the Korean cycle. For the two turns which remain in the nondurables series after deflation, the timing is approximately the same.

Among the undeflated industry series, 29 of the 39 turns in output have matching stock turns, but timing is irregular. There appears to be a tendency for stocks to lag behind output, but a number of leads are noted. On the whole the durables conform better and show more consistent timing.⁶

## TIMING OF STOCKS DURING BUSINESS CYCLES

In table 30 timing of both the deflated and undeflated comprehensive series is compared with reference turns. With the exception of the first peak the deflated total manufacturing and durable stocks series turn roughly coincidently with business cycle (reference) turning points. The deflated nondurables series, of course, conforms only to the 1953-54 business cycle turns.

Since the undeflated goods-in-process industry series may not be relied upon to give an accurate picture of timing of cyclical movements, the sequence of turns in these series has been related to comparable turns in the purchased-materials and finished-goods series. The cyclical behavior of these last two types of stocks has already been described, and it is assumed that the patterns of sequence observed in the undeflated series, if well established, will provide an acceptable approximation of those which would be found to obtain if deflation of all the series could be properly carried out.⁷

⁶ There is a tendency for the durables series to show earlier turns relative to output and sales peaks than they do at troughs. The characteristic does not appear, however, in the nondurables comprehensi. e series. ⁷ Inis assumption is, of course, open to the criticism that price effects (upward during expansion, downward during contraction) may tend to find their way into inventory values in a time sequence in which purchased materials show the effect first, followed by goods in process and finished goods. I have not noticed such a tendency in deflating the purchased-materials and finished-goods series, however.

	Lead $(-)$ or lag $(+)$ in months								
Ind <b>ustry</b>	Busine	ss cycle	Business cycles						
	Peak, Novem- ber 1948	Trough, October 1949	Peak, July 1953	Trough, August 1954	Peak, July 1957	Trough, April 1958			
Deflated inventory data: Total manufacturing Durable-goods industries, total Nondurable-goods industries, total Undeflated inventory data: Total manufacturing Durable-goods industries, total	-14 -14 (²) 0	( ² ) ( ² ) +2 +4	0 +1 -1 +1	+1 +1 +1 +1	  +2 +1	  +7			
Nondurable-goods industries, total	+1 0	+1	-1	+1 +1 +1	+1 +6				

TABLE 30.—Timing of manufacturers' goods-in-process inventories at reference turns 1

¹ There were no turns conforming to "Korean" cycle reference dates, ² No matching turn in inventories.

Source: Based on material from Department of Commerce.

Accordingly, turns in the individual industry goods-in-process series have been compared first with turns in purchased materials and then with those in finished goods.

Comparisons with purchased-materials turns

	Number	Percent
Total comparisons of goods in process	34	100
Leads, 4 or more months Rough coincidences Leads, less than 4 months Coincidences Lags, less than 4 months Lags, 4 or more months	11 18 (11) (2) (5) 5	32 53 (33) (6) (14) 15

#### Comparisons with finished-goods turns

	Number	Percent
Total comparisons of goods in process	24	100
Leads, 4 or more months Rough coincidences Leads, less than 4 months Coincidences Lags, less than 4 months Lags, 4 or more months	15 5 (2) (2) (1) 4	62 21 (9) (8) (4) 17

The typical relationships are apparent. Goods-in-process stocks turn at approximately the same time as purchased materials in over half the observations; but there is a tendency to lead. This is firmly established in relation to finished-goods turns, with leads noted in 71 percent of the comparisons-4 months or more in 62 percent.

These findings are consistent with the comparisons of timing in the deflated comprehensive series. To the extent that we may generalize from these postwar observations, goods in process may be expected to lead or coincide with business cycle peaks and to coincide or lag by a short interval at troughs.

## GOODS-IN-PROCESS INVENTORY INVESTMENT

The overall behavior of total manufacturers' inventory investment in goods in process (chart 15) resembles in several respects that of purchased-materials investment (chart 16). During the period 1945-49 the patterns of movements in the two undeflated series are similar. Investment moves sharply upward until mid 1946, declines, and rises to a second and lesser peak in the first half of 1948. Thereafter each series declines to a trough in 1949. Following this last trough, goods-in-process behavior differs from that observed in the purchased-materials series. There is a major upward movement in goods-in-process investment lasting until the second quarter of 1951, following which there occurs an erratic, generally declining movement until the second quarter of 1953, and then a sharp drop to the trough in 1954.⁸

#### CHART 15

Goods-in-Process Inventory Investment: Total Manufacturing, Durableand Nondurable-Goods Industries, 1946-58



Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated cycles; circles, of undeflated cycles. Source: Department of Commerce. Data deflated by the author.

⁸ The pattern of movement in the deflated series is essentially the same except that the peak in 1948 becomes "submerged" and the peak in 1951 occurs 6 months later.

The purchased-materials series during this last period shows a very sensitive reaction to cyclical forces in the economy in 1951-52, declining sharply to a well marked trough in the second quarter of 1952.

The durables goods-in-process series (undeflated and deflated) show the same general pattern as total manufacturing, except that there are no irregular movements in 1952-53 (chart 16). The nondurables series, on the other hand, show a much less well-defined pattern, being especially choppy from the beginning of 1951 to the second quarter of 1953, so that peaks and troughs cannot be marked with confidence.

## TIMING AND CONFORMITY OF INVESTMENT: COMPARISON WITH OUTPUT, SALES, AND REFERENCE TURNS

The movements described for total manufacturing and durable-goods investment are reflected in the timing measures shown in table 31. Among the industry series most of the output turns may be matched by those in inventory investment, but the timing is irregular.

TABLE 31Timing	and conformity of	of manufacturers'	goods-in-process	inventory
	investment at o	utput and sales tu	rns	•

	Lead (-) or lag (+) in months, in zones associated with reference turns							
Industry	1948	1949	Korean War		1953	1954	1957	1958
	peak	trough	1951 peak	1952 trough	peak	trough	peak	trough
		г	'iming of	investm	ent to ou	tput turi	ns	
Deflated data: Total manufacturing Durable-goods industries, total Nondurable-goods industries,	$+1 \\ +1 \\ +1$	$-3 \\ -2$	(1) (1)	(2) (2)	(8) (8)	$^{+1}_{+1}$		
total Undeflated data	(3)	(8)	(3)	(8)	(7)	(3)		
Total manufacturing Durable-goods industries, total. Nondurable-goods industries.	+1 +1 +1	$^{-3}_{-2}$	(1) (1)	(2) (2)	8	+1 +1	$-21 \\ -21$	$-2 \\ -2$
total Primary metals Machinery (including electrical). Transportation equipment (in-	(3) -2 +10	(3) -2 -1	(³) +8 (¹)	(²) +4 (²)	(*) (*) (*)	(*) -5 -3	(³) 	(ª) 
Transportation equipment (in- cluding motor vehicles) Stone, clay, and glass Food and beverages Paper Chemical Petroleum and coal Rubber	$ \begin{array}{r} -13 \\ -11 \\ (2) \\ -13 \\ -22 \\ -4 \\ (4) \end{array} $	-4 +2 (2) 0 +4 +4 -4	(1) (2) (2) (1) (1) (1) (1)	(2) (3) (4) (4) (1) (2) (2) (3) (4) (4) (4) (4) (4) (4) (4) (4) (4) (4	(3) (3) (1) (1) -2 -3 -3	$^{-9}_{+7}$ (2) (1) (2) (2) (2) (3) (2) (3) (4) (4) (4) (4) (4) (4) (4) (4) (4) (4		
	Timing of investment to sales turns							
Deflated data: Total manufacturing Durable goods industries, total Nondurable-goods industries	1 4	-4 -2	(1) (1)	(2) (2)	(3) (3)	5 5		
total.	(4)	(3)	(?)	(3)	(2)	(2)		
Durable-goods industries, total Nondurable-goods industries, total	-1 -4	$-4 \\ -2$	(1) (1)	(2) (2)	(3) (3)	-5 -5	-14 -14	$-1 \\ -2$
total Primary metals. Machinery (including electrical) Transportetion acquipment (in	(4) 4 1	(3) -2 -7	(*) +8 (1)	(*) +5 (*)	(²) 5 (³)	(*) -6 -8	( ³ )	(*)
cluding motor vehicles) Stone, clay, and glass Food and beverages Paper Chemical Petroleum and coal Rubber	-10 -11 (2) -13 -13 -18 -4 +1	$ \begin{array}{r} -4 \\ -4 \\ (^3) \\ -2 \\ +4 \\ +1 \\ -8 \end{array} $	(1) (3) (3) (4) (1) (1)	() () () () () () ()	() () () () () () () () () () () () () (	$ \begin{array}{c} -11 \\ +7 \\ (3) \\ (1) \\ -2 \\ (4) \\ -14 \end{array} $		

A. TIMING MEASURES

#### 94 INVENTORY FLUCTUATIONS AND ECONOMIC STABILIZATION

TABLE 31.—Timing and conformity of manufacturers' goods-in-process inventory investment at output and sales turns—Continued

	9 indust	try series	4 durab indu	le-goods stries	5 nondurable-goods industries	
	All turns	All turns except Korean	All turns	All turns except Korean	All turns	All turns except Korean
Number of comparisons Matching inventory turns Percentage of matching turns	38 34 90	28 25 90	20 16 80	16 13 81	18 18 100	12 12 100

B. SUMMARY OF CONFORMITY TO OUTPUT

Inventory investment turn occurs, but there is no activity turn.
 No turn occurs in either activity or inventory investment.
 Activity turn occurs but no matching inventory investment turn.
 Not available.

Source: Based on material from Department of Commerce.

Timing comparisons of the comprehensive series and reference turns are shown in table 32. The total manufacturing and durables series (both deflated and undeflated) lead at all business cycle turns.

TABLE 32.—Timing of manufacturers' goods-in-process inventory investment at reference turns

			Lead (	-) or lag	( <b>+</b> ) in :	months				
Industry	Business cycle		Korean cycle		Business cycles					
	Peak No- vember 1948	Trough Octo- ber 1949	Peak Febru- ary 1951	Trough June 1952	Peak July 1953	Trough Aug- ust 1954	Peak July 1957	Trough April 1958		
Deflated data: Total manufacturing Durable-goods industries, total Nondurable-goods industries, totat	-3 -3	-2 -2	+9 +6	(1) (1) (1)	(1) (1)	-3 -3				
Total Total manufacturing Durable-good industries, total Nondurable-goods industries, total	(1) -3 (1)	(1) $-2 \\ -2 \\ (1)$	(·) +3 0 ( ¹ )	() () ()	(1) (1) (1)	(·) -3 -3 ( ¹ )	-20 -20 (1)	2 -2 (1)		

¹ No turn in inventory investment.

Source: Based on material from Department of Commerce.

For the individual industry series, turn sequences are summarized below.

Com	parisons	with	purchased	materials	investment	turns
-----	----------	------	-----------	-----------	------------	-------

	Number	Percent
Total comparisons: Goods-in-process investment	35	100
Leads, 4 or more months Rough coincidences Leads, less than 4 months Coincidences Lags, less than 4 months Lags, 4 or more months	4 25 (10) (9) (6) 6	11 72 (29) (26) (17) 17

	Number	Percent
Total comparisons: Goods-in-process investment	37	100
Leads, 4 or more months Rough coincidences Leads, less than 3 months Coincidences Lags, less than 3 months Lags, 4 or more months	15 19 (12) (4) (3) 3	40 52 (33) (11) (8) 8

Comparisons with finished goods investment turns

A clear timing tendency is revealed: turns in goods-in-process investment occur roughly coincidently with turns in purchased-materials investment, but lead turns in finished-goods investment.

These findings are sufficiently in agreement with the measures for the other comprehensive series presented in table 35 (ch. 7) to permit a generalization as to postwar timing. Goods-in-process investment, like purchased-materials investment, will tend to lead business cycle turns. The lead may be of considerable duration.

## ANALYSIS AND REVISION OF ABRAMOVITZ' THEORY

If one attempts to pass judgment on Abramovitz' theory in the light of the observed behavior of the data, the results are ambiguous. The general tendency for goods in process to roughly coincide with turns in output is consistent with Abramovitz' explanation. On the other hand, there is nothing in the theory that would lead one to expect the several very long leads and lags that occur in the industry series, or the failure of such goods in process as total nondurables and certain of the industry series to conform to the Korean cycle.

The behavior of inventory investment is even less consistent with the theory. In a comparison of investment turns in the undeflated industry series with turns in rates of change in comparable output series, of the 36 turns in the latter, timing was as follows: 6 leads (3 of which 6 or more months); 9 coincidences; 21 lags (17 of which, 6 or more months).

The leads and lags of 6 months and more would appear to be well outside the range of possible timing according to Abramovitz' theory. Especially difficult to explain is the behavior of investment in the total manufacturing and durable series after 1949. The great upward movement in inventory investment observed in these series in chart 15 lasted until the latter half of 1951, the peak occurring roughly coincidently with the trough in the rate of change in output.

I can suggest four possible explanations for this disagreement between Abramovitz' theoretical statement and the behavior of the data: (1) Errors in the data due to price movements; (2) definitional difficulties leading to a reporting of stocks other than goods in process; (3) changes in composition of goods-in-process stocks since World War II; and (4) theoretical omissions.

Little need be said regarding the first point; it has already been shown that formidable obstacles to deflation exist. While we can be confident of neither the undeflated nor the deflated data, the differences noted between the timing in the series and the timing postulated by Abramovitz cannot be attributed principally to this

76626—61—pt. I——8

difficulty. Experience with the purchased-materials and finishedgoods series, for which deflation is more reliable, indicate that only in a small minority of cases does deflation significantly alter the essential pattern of movements in stocks and inventory investment.

It is quite possible that in the current data goods in process are not identical with those defined by Abramovitz. We do not know for instance how much of these goods are fully fabricated items held by plants in vertically integrated organizations, goods which are in process only in the sense that they will enter into further production in another division of the organization. In a nonintegrated firm such stocks would be reported as finished goods. Although there is no way to determine the extent to which this is true, its significance is doubtful. It is only reasonable to expect integrated corporation reports to be composed of consolidated accounting records kept by member plants or divisions. Since the cost records of individual plants or divisions contain separate statements of inventories held at each stage of fabrication the consolidated report should not provide a different figure from that which would have been reported had the production units been individual firms.

The third possible reason for differences between theory and observed behavior appears to be of greater importance. Abramovitz' analysis rested on the assumption that continuous process operations were the most important in American industry, and that the behavior of goods in process within stages dominated the cyclical pattern of total goods in process. This assumption was based on his estimates of the relative amount of goods in process held by continuous, mixed, and discontinuous industries in 1939.

As shown previously, there has been a marked change in the structure of American industry since 1939; the durables, especially transportation equipment and machinery, having assumed a more prominent place. It is precisely these industries that have discontinuous production processes.⁹

Approximately 50 percent of total stocks were held by durables manufacturers at the close of 1939. Comparable figures for December 31, 1953, show durables with 58 percent of total stocks.¹⁰ The dis-

	Value of total stocks by production process of minor industries			
	Continuous	Discontinuous	Mixed	
Nondurable goods manfacturing: Food and tobacco products	88	12		
Textiles and textile products Leather and leather products Rubber products	36 57 100	2 43	62	
Paper and allied products	73 100		27	
Durable goods manufacturing: Lumber and wood products	20	43	37	
Stone, clay, and glass. Ferrous and nonferrous metals and their products	15	38	47	
(excluding machinery) Machinery (including electrical) Transportation equipment (including auto)	45 5 17	30 62 83	25 33	

 This is clearly seen in Abramovitz' estimates of the value of total stocks held by production processes on Dec. 31, 1939 (compiled from table 105, with dollar figures converted to percentages, ibid., p. 560).
 ¹⁰ I derived the 1939 figure from ibid., table 105, p. 560, omitting the miscellaneous category and classifying the industries as durable or nondurable. The 1953 figure was obtained similarly using the 1953 Annual Survey of Manufactures data.
parity between the amounts of goods-in-process stocks held by the durables and nondurables categories on the two dates is even greater because the discontinuous processes, so important to the durables industries, carry a higher proportion of total stocks as goods in process than do the mixed or continuous processes. Using Abramovitz' assumptions, it is estimated that durable manufactures held 57 percent of all goods in process in 1939 and 65 percent in 1953.¹¹ But the latter figure, although comparable with Abramovitz', is too low. According to the census estimates (Annual Survey of Manufactures), goods in process held in 1953 by these same industries amounted to 81 percent of the total.¹²

The effect of this increased role of durables is, of course, to increase the proportion of total goods in process held by discontinuous process industrial activities. Using Abramovitz' assumptions and method, comparable estimates for 1939 and 1953 are as follows:

	1939	1953
Continuous	37. 7	32
Discontinuous	36. 1	46
Mixed	26. 2	22

An estimate made directly from the postwar census goods-in-process data, rather than the Abramovitz method, results in a slightly higher proportion of goods-in-process stocks held by industries engaging in discontinuous processes, roughly 50 percent. The significance of this finding is clear. The increased importance of that type of goods in process most loosely tied to the rate of production could account, in part, for the disparity noted between the behavior hypothesized and that observed in the data.

Finally, it appears that Abramovitz in his theoretical discussion omitted some possible types of behavior which may be of importance. The first omission concerns the behavior of goods in process between As pointed out previously, Abramovitz' treatment of the bestages. havior of these stocks was somewhat sketchy; only surplus stocks could move in an inverted fashion, but they need not. In the main, he holds, these between-stage stocks may be expected to move fairly closely with output. I suggest that when these goods in process are standardized component parts they need not behave either in an inverted fashion or move with the rate of output, but may be treated in a manner similar to purchased materials; that is, increased in advance of actual utilization on the basis of orders received. Further, as in the case of purchased materials, it may be desirable for the manufacturer to allow these goods in process to accumulate when there is a rapidly rising backlog of orders or when a large order backlog is not being significantly reduced. Such a hypothesis is particularly helpful in explaining the long expansion in durables goods-in-process investment from 1949 to 1951.

The second omission concerns goods in process held within stages of production. Abramovitz assumes that these will be effected by

¹¹ In preparing this estimate I made use of Abramovitz' estimates of the value of each industry's total stocks held in continuous, discontinuous, and mixed type operations and converted these by using his average ratios of goods in process to total stocks for the 3 types of processes. See ibid., p. 164, for these ratios. ¹³ Excluding the miscellaneous category. See footnote 10 above.

changes in the rate of output. I suggest that this need not be the case if output is varied simply by working more or fewer days per week or more or fewer shifts per day, without increasing or decreasing the number of machines used. Under such conditions the amount of material in process at any time need not be altered. This practice is quite important in some industries (e.g., in the cotton broad-woven goods industry, varying the number of days worked per week is the principal device for varying output). Further, it is a well-established fact that the length of the workweek in manufacturing varies with the business cycle. To the extent that output is varied in this fashion, fluctuations in the level of goods in process within stages will tend to be muted, and the type of movement visualized by Abramovitz will play a lesser role.

If these previously omitted cases are added to the theoretical description of goods-in-process inventory behavior and due account is taken of the changed composition of these stocks, the composite picture is altered substantially. The movements of within-stage stocks become less important in determining the overall pattern because these stocks are smaller and because they are not so completely tied to movements in the rate of change in output as was formerly supposed. The movements of between-stage stocks, on the other hand, become more important, for they are much larger than in prewar years. Moreover, they are no longer so closely tied to output movements under the revised theory, but are related in a larger measure to the volume of incoming orders and to the levels of unfilled orders.

These revisions do not leave us with a very precise theory, but certain conclusions may be drawn as to expected behavior. Goods-inprocess stocks may be expected to conform closely to cyclical movements. Expected timing of these movements is somewhat indeterminate, except that there would seem to be little reason to anticipate the development of inverted tendencies. Goods-in-process investment movements would be expected to resemble those of purchasedmaterials investment, moving with a lead at both peak and trough and thereby contributing to the early set of forces which brings about the end of the expansion and of the contraction phases.

Such a theory goes far toward explaining the difference in behavior of durable and nondurable goods-in-process investment. Investment series for durables manufacturers' goods-in-process inventories, which contain a large proportion of between-process stocks, are relatively sensitive to business cycles; those for nondurables, which contain a small proportion of such stocks, are relatively insensitive.

### SUMMARY

Although problems of deflation render conclusions regarding goodsin-process behavior less dependable than those for purchased materials and finished goods, certain characteristics may be noted. The inventory series conformed to business cycles with virtually coincident timing at four of the six reference cycle turns, and inventory investment led all business cycle turns. Analysis of sequence of timing among the industry series indicates that, for both inventory and investment, goods in process lead or turn coincidently with purchased materials and lead finished-goods series. Although these findings are generally consistent with Abramovitz' theory among the individual industry investment series, irregularities in timing appear which cannot be explained by his analysis. Three factors appear to contribute to this behavior:

(1) The composition of postwar goods in process is different from that observed by Abramovitz. Owing to the increased role of durables, there is a much larger proportion of these stocks held by industries engaging in discontinuous processes, roughly 50 percent compared with his estimate of 36 percent.

(2) Goods-in-process stocks in discontinuous-process industries are likely to be held in substantial quantities between stages. These stocks may be expected to rise and fall in a manner similar to that of purchased materials.

(3) Goods held within stages need not fluctuate in as close conformity to changes in output as Abramovitz maintained.

# CHAPTER 7

# CYCLICAL BEHAVIOR OF INVENTORY AND INVENTORY INVESTMENT MOVEMENTS AT THE THREE STAGES OF FABRICATION: SUMMARY

It is now possible to bring together summary statements regarding manufacturers' inventories and investment, and to note how movements at each stage of fabrication form the composite patterns of total manufacturers' inventory and inventory investment behavior.¹

 TABLE 33.—Timing of inventories and inventory investment at business cycle turns:

 Summary

	Lead $(-)$ or lag $(+)$ , in months						
Business cycle turn	Purchased materials	Finished goods	Goods in process	Total in- ventories			
Peak, November 1948 Trough, October 1949 Korean rough, June 1952 Peak, July 1953 Trough, August 1954 Peak, July 1957 Trough, April 1958 Averages: Peaks and troughs Prewar average, peaks and troughs	$\begin{array}{r} -3.0 \\ +6.0 \\ +9.0 \\ +4.0 \\ +1.0 \\ +8.0 \\ +4.0 \\ +5.0 \\ +2.8 \\ +5.8 \\ +4.5 \end{array}$	$\begin{array}{r} +7.0 \\ +11.0 \\ +14.0 \\ +4.0 \\ +8.0 \\ +1.0 \\ +6.0 \\ +8.0 \\ +8.8 \\ +6.0 \\ +7.4 \end{array}$	$\begin{array}{c} -14.0 \\ +2.0 \\ (1) \\ (1) \\ (1) \\ (1) \\ +1.0 \\ +2.0 \\ +7.0 \\ +7.0 \\ -4.0 \\ +3.3 \\3 \end{array}$	(1) +4.0 +4.0 +6.0 +6.0 +1.0 +2.0 +1.0 +2.0 +2.0 +2.0 +2.0 +2.0 +2.0 +2.0 +3.8 +8.6 +2.7 +5.0 +3.8 +8.6 +8.6 +3.8 +8.6 +8.6 +8.6 +8.6 +8.6 +8.6 +8.6 +8			

### A. INVENTORIES, ALL MANUFACTURERS

B. INVENTORIES, DURABLE AND NONDURABLE MANUFACTURERS

	Lead (-) or lag (+), in months									
Business cycle turn	Purchased materials		Finished goods		Goods in process		Total in- ventories			
	Dur- able	Non- durable	Dur- able	Non- durable	Dur- able	Non- durable	Dur- able	Non- durable		
Peak, November 1948 Trough, October 1949 Korean peak, February 1951 Korean trough, June 1952 Peak, July 1953 Trough, August 1954 Peak, July 1957 Trough, April 1958	$\begin{array}{r} -16.0 \\ +4.0 \\ +13.0 \\ +2.0 \\ +1.0 \\ +10.0 \\ -4.0 \\ +4.0 \end{array}$	$\begin{array}{c} -1.0 \\ +8.0 \\ +9.0 \\ (1) \\ (1) \\ +7.0 \\ +1.0 \\ +10.0 \end{array}$	+8.0+11.0(1)+5.0+1.0+6.0+9.0	$ \begin{array}{c} +6.0 \\ +11.0 \\ +14.0 \\ +10.0 \\ (1) \\ (1) \\ +6.0 \\ +5.0 \end{array} $	-14 +4 (1) (1) +1 +1 +1 +7	(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)	-14 +4 (1) (1) +2 +9 +3 +8	(1)(1)(1)(1)+2.0+1.0+1.0+5.0		
Averages, peaks and troughs	+1.8	+5.7	+6.7	+8.8	0	+3.2	+2	+3.2		

¹ Timing measures mentioned in this section are presented in table 33; inventory investment movements are shown in chart 16.

101

#### TABLE 33.—Timing of inventories and inventory investment at business cycle turns: Summary—Continued

ed Finished ls goods	Goods in process	Total inventory
		investment
$\begin{array}{c c} & & +3\\ 5.0 & -2\\ 3.0 & +6\\ 2.0 & -1\\ 2.0 & +1\\ 9.0 & -3\\ 3.0 & -14\\ 1.0 & +1\\ 1.0 & +1\\ 8.5 & -1\\ 2.8 & -1\\ 5.6 & -1\\ \end{array}$	$ \begin{array}{c} 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 $	$\begin{array}{c} -6.0\\ -2.0\\ +3.0\\ -1.0\\ -1.0\\ -2.0\\ 0\\ -14.0\\ +1.0\\ -4.8\\2.6\\ -2.6\end{array}$
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

#### C. INVENTORY INVESTMENT, ALL MANUFACTURERS

#### D. INVENTORY INVESTMENT, DURABLE AND NONDURABLE MANUFACTURERS

	Lead (-) or lag (+), in months								
Business cycle turn	Purchased imaterials		Finished goods		Goods in process		Total in- ventories		
	Du- rable	Non- durable	Du- rable	Non- durable	Du- rable	Non- durable	Du- rable	Non- durable	
Peak, November 1948 Trough, October 1949 Korean peak, February 1951 Korean trough, June 1952 Peak, July 1953 Trough, August 1954 Pak, July 1957 Trough, April 1958 Averages, peaks and troughs	$\begin{array}{c} -6.0 \\ -5.0 \\ -3.0 \\ -1.0 \\ -2. \\ 0 \\ -23.0 \\ +1.0 \\ -4.9 \end{array}$	$ \begin{array}{r} -6 \\ -11 \\ -3 \\ (1) \\ (1) \\ -6 \\ -2 \\ +4 \\ -8 \end{array} $	$ \begin{array}{r} 0 \\ -2 \\ +6 \\ +2 \\ +1 \\ -3 \\ +4 \\ 0 \end{array} $	$\begin{array}{r} -6.0 \\ -2.0 \\ +6.0 \\ -1.0 \\ +3.0 \\ -14.0 \\ +1.0 \\ -14.5 \end{array}$	$\begin{array}{c} -3 \\ -2 \\ +6 \\ (1) \\ (1) \\ -3 \\ -20 \\ -2 \\ -4 \end{array}$	3366663336	$\begin{array}{r} -6.0 \\ -2.0 \\ +6.0 \\ -1.0 \\ -2.0 \\ -20.0 \\ +1.0 \\ -3.4 \end{array}$	$\begin{array}{c} -6.0 \\ +7.0 \\ +3.0 \\ (1) \\ (1) \\ -9.0 \\ -14.0 \\ +4.0 \\ -2.5 \end{array}$	

No matching inventory turn.

Source: Tables 8, 9, 12, 14, 21, 23, 28, and 30. Source of prewar (1919–39) averages, based on annual data: Abramovitz, "Inventories and Business Cycles," pp. 95 and 338. All 1948-54 data are deflated (1947 dollars) undeflated (book value) data are used thereafter.





See notes on p. 105.



See notes on p. 105.



Chart 16—Continued

Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of deflated cycles; circles, of undeflated cycles. Source: Department of Commerce. Data deflated by the author.

### PURCHASED MATERIALS

Purchased materials show a high degree of sensitivity to business cycles, particularly in the durables group. The stocks proper tend to turn roughly coincidently with business cycle peaks and to lag at business cycle troughs. But the rates of change (the investment series) show a single well-developed tendency: They lead at all turns. It is interesting to observe that the timing is very similar to that of new orders. When orders begin to turn up late in the recession, the rate of disinvestment declines. As the trough is reached and early recovery is noted, purchased-materials stocks continue to decline but

at a decreasing rate. Several months after the trough, declining disinvestment changes to rising investment as the stocks turn and move up. A similar movement occurs at the peak, except that stocks proper turn relatively earlier.

The amplitude of the investment movements appears to be influenced by the level of unfilled orders. When orders are rising at a faster pace than shipments, so that the backlog of unfilled orders is growing rapidly, or when the backlog is very large and not significantly diminishing, upward movements of new orders prompt much heavier investment in purchased materials than otherwise.

### FINISHED GOODS

Finished stocks are, of course, the lagging series. They lag behind business cycle turns from 1 to 11 months. Finished-goods inventory investment, however, reaches its peak or trough roughly coincidently with the business cycle peak or trough.

The relatively early timing of finished-goods investment turns is a well-established finding, based upon observation of commodity series as well as the Department of Commerce data. In the activity expansion phase, the latest occurrence of the trough in inventory investment is the quarter in which shipments spurt forward at their most rapid rate; frequently it occurs somewhat earlier. The converse is true for the peak in investment: it rarely occurs later in the activity contraction phase than the quarter during which shipments fall at their most rapid rate. Since it is a well-established characteristic of the business cycle that the highest rate of increase in activity typically occurs during early recovery, and the highest rate of decrease during the early or middle stages of recession, it may be concluded that any countercyclical movement of finished-goods investment is terminated early in each business cycle phase.

### GOODS IN PROCESS

Problems of deflation render conclusions regarding goods-in-process behavior less dependable than those for purchased materials and finished goods. Several characteristics may be noted, however. The deflated inventory series conform to business cycles with virtually coincident timing at four of the six business cycle turns. Analysis of timing sequence indicates that goods in process lead or turn coincidently with purchased-materials inventory turns, and lead finishedgoods turns. Goods-in-process inventory investment movements are cyclically sensitive and give evidence of being influenced by unfilled order levels. Although timing is somewhat irregular, the typical sequence is one in which these series turn roughly coincidently with turns in purchased materials, but lead turns in finished-goods investment.

TOTAL STOCKS AND INVENTORY INVESTMENT

The three components combine to form a total manufacturing inventory complex which is highly responsive to cyclical forces. During the period studied, total stocks lagged behind business cycle turns from 1 to 8 months, and total inventory investment turned coincidently or led by as much as 14 months (table 33).

Of course, the pattern of behavior for total stocks is merely an average of the component patterns, and there is actually a consider-

able diversity in the timing of inventory turns at each stage of fabrication. The striking feature is that inventory investment at each stage is so sensitive to cyclical forces. Throughout the analysis it has been noted that inventory investment conformed somewhat better to business cycles than did the inventory series proper. Even the finished-goods investment series lags behind the other investment series by no more than 6 months at four of the six business cycle turns, and shows almost no countercyclical tendencies.

### Amplitude of Investment Movements Compared

Tables 34 and 35 present relative and dollar value measures of change during movements in inventory investment for each stage of fabrication. It may be observed that changes in each type of inventory investment have contributed significantly to cyclical instability. Comparing total manufacturers' purchased materials and finished goods investment, we find that the amplitude of total change is somewhat greater for the latter in six of the seven phases. This is true for measures of change in dollars and on a relative basis. Amplitude of changes in goods-in-process investment is more variable. In the earlier phases it ranked behind purchased materials and finished goods investment, but more recently it has tended to move with greater amplitude than either.

Industry	Peak- trough (1948-49)	Trough- peak (1949-50)	Peak- trough (1950-52)	Trough– peak (1952–53)	Peak- trough (1953-54)	Trough- peak (1954-56)	Peak- trough (1956-58)			
		Total change (millions of dollars)								
Total manufacturing: Purchased materials Goods in process Finished goods	-843 -208 -835	+1,220 +753 +1,821	-1, 017 -1, 703	+336 +666	-627 1 -779 -790	+1, 143 +1, 273 +1, 190	-1, 136 -1, 649 -1, 436			
Purchased materials Goods in process Finished goods Total nondurables manufactur-	-415 -224 -315	+700 +734 +908	-626 	+406 +360	-513 1 -739 -668	$+921 \\ +1,328 \\ +856$	-1, 089 -1, 550 -700			
ng: Purchased materials	588	+680			1 -320	+507	-415			
Finished goods	-561	+923	-920	+379	278	+602	-882			
	Change per month (millions of dollars)									
Total manufacturing: Purchased materials Goods in process Finished goods Total durables manufacturing: Purchased materials	-70 -15 -139 -35	+68 +27 +76 +39	-48 -142 -35	+112 +55 +34	-52 1-52 -88 -29	+54 +71 +49 +44	54 61 60 52			
Finished goods Total nondurables manufactur- ing:	-12 -52	+24 + 38	-68		-74	+29	-33			
Purchased materials	-98	+28			1 - 36	+34	-11			
Finished goods	-37	+38	-102	+25	-18	+33	-37			

TABLE 34.—Amplitude of change in inventory investment cycles, by phase, 1948-58

¹ To permit comparison of amplitudes of downward movements during recession of 1953-54, purchasedmaterials investment change in nondurable goods has been measured from 2d quarter 1953 to 1st quarter 1954; goods in process (except nondurable) change has been measured from 1st quarter 1953 to 2d quarter 1954.

Source: Based on deflated material from Department of Commerce. All 1948-54 data have been deflated (1947 dollars). Measures for the 2 most recent phases are based on undeflated (book value) data.

Industry	Peak- trough (1948-49)	Trough- peak (1949-50)	Peak- trough (1950-52)	Trough- peak (1952-53)	Peak- trough (1953-54)	Trough- peak (1954-56)	Peak- trough (1956–58)				
		Total change (relative)									
Total manufacturing: Purchased materials Goods in process Finished goods Total durables manufacturing:	-7.15 -3.06 -7.72	+10. 28 +9. 73 +15. 95	-7.82 -13.51	+2.43	-4.57 -7.65 -5.83	+7.44 +9.76 +6.93	-7.20 -11.44 -7.60				
Purchased materials Goods in process Finished goods Total nondurables manufac-	-8.60 -4.85 -7.35	+14 94 +14. 21 +19. 44	-12.12 -15.74	+7.23 +6.43	-9.69 2-9.67 -11.23	+12.72 +12.83 +10.16	-14.78 -13.45 -7.60				
turing: Purchased materials Goods in process Finished goods	-8. 27 -9. 37	+9.28 +13.55	-12.40	4. 88	² -3. 93 -3. 56	+6.38 +6.74	5. 00 9. 00				
	Change per month (relative)										
Total manufacturing: Purchased materials Goods in process. Finished goods Total durables manufacturing: Purchased materials Goods in process. Finished goods Total nondurables manufac	60 20 -1. 29 72 40 82	+. 57 +. 36 +. 66 +. 83 +. 59 +. 81	37 -1. 13 67 -1. 31	+.81 +.41 +.60	$\begin{array}{r}38\\ 251\\65\\65\\ 264\\ -1.25\end{array}$	+.35 +.54 +.29 +.61 +.72 +.34	34 42 32 70 50 36				
turing: Purchased materials Goods in process Finished goods	-1.38 62	+. 39 +. 56	-1.38	+. 33	² 44 24	+.42	13 38				
Finished goods	62	+. 56	-1.38	+. 33	24	+. 37	-				

 TABLE 35.—Amplitude of change in inventory investment cycles, 1948–58, as percent of mean level of inventories during each phase 1

 Mean level of inventories was computed separately for each phase using average of beginning and ending level of inventories of terminal quarters of the phase.
 See note (1) to table 34.

Source: Based on deflated material from Department of Commerce.

# POSTWAR AND PREWAR TIMING COMPARED

Since frequent mention has been made of Abramovitz' study, a brief comparison of his timing measures with those for the postwar period is relevant.²

Lagging movements in total inventories have been characteristic in the postwar period, but relative to business cycle turns they have been shorter, averaging about 4 months compared with 8 or 9 months during the previous period. In part, this earlier timing may be due to the shift in composition toward durables and goods in process, resulting in greater sensitivity and consilience with the cycle. It may also be due to a tighter control of inventories by management.

Abramovitz' estimate of short lags (about 3 months) in purchasedmaterials inventories is fairly consistent with postwar experience. Lags at troughs (averaging about 6 months) have been longer than at peaks (averaging about 3 months).

Abramovitz drew no conclusions regarding the timing of finishedgoods inventories as a whole, but dealt with the behavior of the component classifications. It is important to observe that the largest

² Timing measures mentioned in this section are presented in table 33.

component (demand-oriented staples made to order), which demonstrates marked inverted characteristics, has been relatively smaller since the war, and that the positively-conforming, made-to-order category is larger as a result of the increased share of durable-goods production. Accordingly, there seems to be little doubt that lags in aggregate finished stocks have been shorter in the postwar period. They have averaged about 7 months, always turning before the end of the phase even when the phase is short.

The roughly coincident postwar behavior of goods-in-process stocks is consistent with Abramovitz' estimate, but inventory investment behaves somewhat differently.

Leads in total manufacturers' inventory investment have been characteristic in the postwar period, whereas Abramovitz calculated rough coincidences from prewar annual data. Abramovitz concluded that investment would be likely to show a substantial lag behind rates of change in output, and this has been confirmed by the postwar data. He expected that normally the business cycle peak would occur at about the same time as the inventory peak. This would be due, however, to the reciprocal effect of inventory investment on national product. He did not rule out the possibility that leads might occur.

Abramovitz expected short leads in purchased-materials investment relative to business activity, but lags relative to the rate of change in output. Postwar experience is reasonably consistent with this, although some lags behind rates of change in output were longer than could be explained by Abramovitz' theory.

As in the case of finished-goods inventories, it is not possible to compare postwar and prewar timing in finished-goods investment directly, but the evidence points to earlier turns since the war. For the postwar period as a whole, rough coincidence has been the rule. Investment in goods made to stock has moved inversely to the rate of change in output, frequently with a lead. Since turns in the rate of change have occurred rather early in expansions or contractions, investment has moved positively to output during most of the phase (i.e., lags on a positive basis, when they have occurred, have been fairly short). Investment in goods made to order, which probably moves positively, has been more important since the war because of the greater importance of durables.

Abramovitz expected long leads, in goods-in-process investment because it should turn coincidently with the rate of change in output. But short leads have been characteristic since the war, goods in process having lagged behind the rate of change in output. These goods are not tied so closely to output as Abramovitz thought.

#### CONCLUSION

It is well to note the degree to which these findings attest the value of Abramovitz' approach to the study of manufacturers' inventories. This analysis of the behavior of various component stocks, constructing the composite from knowledge so derived, has made it possible to observe the characteristic behavior of various types of inventory and, thereby, to evaluate the effects of shifts in composition. Without such a study it would have been impossible to assess the significance of declines in the relative size of finished stocks and of increases in the importance of durables, or to explain why manufacturers' stocks, which are smaller today than before the war, have played no less important a role in business cycles.

The analysis must be regarded as incomplete, however, on the grounds that the business cycle has been assumed as given. Inventory movements have been regarded as responsive to the cycle, aggravating the fluctuations, but not themselves causing either upturn or downturn. In short, the income effect of inventory investment has not been treated, nor has the possibility of a self-generating inventory cycle. These matters are the concern of the chapter which follows.

### CHAPTER 8

### HOW INVENTORY MOVEMENTS CONTRIBUTE TO INSTABILITY

In observing the behavior of manufacturers' inventories, it has been assumed that fluctuations in general business activity are caused by forces unrelated to inventory movements, and that the latter may be regarded as dependent fluctuations, superimposed upon the business cycle.

It has been shown, however, that inventory investment movements are among the major variables in the economy. The demonstrated cyclical sensitivity of this important component of the national product raises two fundamental questions: In what manner do changes in inventory investment contribute to the cumulative forces of expansion and contraction? Is it possible that movements in inventory investment may spark the upswing and bring about the downturn, thereby constituting an underlying cause as well as an aggravating force?

### METZLER'S INVENTORY CYCLE THEORY

To answer these questions it is useful to examine Lloyd Metzler's theory of the inventory cycle.¹ Metzler presents his study, not as a theory of the business cycle, but as a contribution to the understanding of its nature. In the American Economic Review article in which he summarizes his theory and explains its significance, Metzler reviews several of the earlier cycle theories and points to what he considers their common property: each conceives of expansion and contraction as highly cumulative in nature and capable of being terminated only through certain limiting forces which are operative only toward the end of each phase. These older theories hold "that the economy is essentially unstable."²

In Metzler's alternative explanation, business cycles may be regarded as recurrent deviations from equilibrium levels of income, deviations which come about as a result of the structure of the economic system. In this view the economy is essentially stable, but any increase in demand (through, say, a change to a permanently higher level of autonomous investment) will set up oscillations around a new equilibrium.

The inventory cycle is offered as an example of such a structural cycle. It comes about as a result of the attempts of businessmen to bring inventories into a desired relationship with sales. Initially, efforts to replace or increase depleted inventories increase income and consumption, thereby pulling down the level of stocks further and causing additional increases in inventory demand, income, and con-

76626-61-pt. I-9

¹Metzler's elaboration of the nature and significance of inventory cycles is set forth in three articles: "The Nature and Stability of Inventory Cycles," Review of Economic Statistics, August 1941, pp. 113-129; "Factors Governing the Length of Inventory Cycles," ibid., February 1947, pp. 1-15; "Business Cycles and the Modern Theory of Employment," American Economic Review, June 1946, pp. 278-291. ³ Metzler, "Business Cycles and the Modern Theory of Employment," p. 280.

sumption. Ultimately, however, efforts to adjust stocks become more successful in spite of the derived increases in consumption, and at some point the rate of inventory accumulation reaches a peak and This brings a corresponding decline in income and begins to decline. consumption, resulting in unwanted stocks. Efforts of businessmen to reduce stocks cause further declines in income and consumption, again defeating inventory objectives. But once again inventory adjustment ultimately begins to be achieved. Disinvestment reaches its maximum and begins to diminish (i.e., there is an increase in the inventory investment series), bringing an increase in aggregate demand, and a new cycle follows.

It is essential to the argument that the interaction of the multiplier and accelerator causes the rate of change in sales to decline after a time during expansion (and rise after a time during contractions). The accelerator is defined below. The multiplier is simply the reciprocal of the marginal propensity to save. This reversal in direction of movement in the rate of change in sales brings in its wake a reversal of inventory investment and of income.³

In his two articles dealing with the properties of inventory cycel models, Metzler varies assumptions regarding the size of the marginal propensity to consume, the nature of expectations, and the inventory accelerator.⁴ In all the models the marginal propensity to consume is equal to the average propensity to consume, and consumption is not lagged (i.e., all consumption expenditures are made from current income). On the other hand, production is lagged, in the sense that production in a given period is guided by the sales of the preceding period. In the simplest models, production for sales in period t is equal to sales in t-1, and businessmen attempt to maintain a constant level of stocks. But in the more complex models, the coefficient of expectations and the inventory accelerator are introduced. The coefficient of expectations is a percentage figure, applied to the change

* The argument may be stated in terms of a relatively simple model in which sales in period t are expected to be those observed in period t-1 and desired inventories are a constant proportion of sales: (1) Output  $(Y_i)$  = autonomous investment (A) plus output for sale, i.e., sales of last period  $(S_{t-1})$ , plus planned inventory investment  $(I_t)$ 

$$Y_i = A + S_{i-1} + I_i$$

(2) Planned inventory investment is the sum of: (a) Unplanned inventory disinvestment of last period  $(I_{I'})$  which is the difference between expected and actual sales last period. Since the expected sales last period are  $S_{I-3}$ , and the actual sales are  $S_{I-1}$ ,

I'=S1-1-S1-3

(b) The increase in required stocks made necessary by the increase in sales  $(I_i'')$ . Stocks required in period t equal expected sales in t times the inventory sales ratio (a). Stocks required in period t-1equal expected sales in t-1 times a. Since  $I_i'$  would only bring stocks into line with sales in t-2,  $I_i''$ must bring stocks into line with  $S_{i-1}$  (expected  $S_i$ ). Therefore,

(c) 
$$I_{i}=I_{i}'+I_{i}''=a(S_{i-1}-S_{i-3})$$
$$I_{i}=I_{i}'+I_{i}''=(I_{i}+a)(S_{i-1}-S_{i-3})$$
$$I_{i}=(1+a)(S_{i-1}-S_{i-3})$$

(3) Total output, therefore, is:

whe

$$Y_{i}=A+S_{i-1}+(1+a)(S_{i-1}-S_{i-1})$$

(4) From the above, the conditions under which the movement of output is revised may be noted. In period t the change in output from the previous period is  $Y_t - Y_{t-1}$ 

$$\begin{array}{c} Y_{i}=A+S_{i-1}+(1+a)(S_{i-1}-S_{i-3}) \\ Y_{i-1}=A+S_{i-3}+(1+a)(S_{i-3}-S_{i-3}) \end{array}$$
 and

As long as sales are rising at an increasing rate Y must rise, and at an increasing rate. But if the rate of change in sales begins to decline,  $Y_{i-} Y_{i-1}$  will remain positive only so long as  $S_{i-1}$  exceeds  $S_{i-2}$  by more than  $(1+\alpha)$  times the difference between the rates of change in sales. If the rate of change in sales is declining,  $S_{i-1} - S_{i-2}$  is becoming a smaller positive quantity while  $(S_{i-1} - S_{i-2}) - (S_{i-2} - S_{i-2})$  is becoming a larger negative quantity. Given a decline in the rate of increase in sales, the sign of  $Y_i - Y_{i-1}$  must, therefore, eventually become negative.

: - ·

in sales in the current period  $(S_t - S_t - 1)$ , to give the expected change in the next period. The inventory accelerator is expressed as a percentage figure which, when applied to expected sales, gives the inventory objective for the next period.

Metzler concludes that the propensity to consume is a more important factor governing the length of the cycle than are the inventory accelerator and the coefficient of expectations. Expansion and contraction phases of the cycle are shown to be more prolonged the larger the propensity to consume or the larger the inventory accelerator, but less prolonged with larger values of the coefficient of expectations. The cycle can only occur for certain values of the marginal propensity to consume and the accelerator. At values below the critical limits, a disturbance merely leads to a new position of equilibrium; above the critical limits, the cycle is unstable, aggregate output moving continuously upward with wavelike fluctuations in the rate of increase.

### OBSERVATIONS ON THE VALIDITY OF METZLER'S THEORY

That it is rarely possible to submit an economic theory to a conclusive empirical test is an all too familiar observation. Useful theory requires a high level of abstraction and the generous use of ceteris paribus assumptions. When the theorist makes use of the rigorous language of mathematics these difficulties may be multiplied, for special assumptions must be kept at a minimum, and the language permits of few ambiguities. In short, theory attempts to cut through to essential and recurrent processes, but empirical tests must be conducted in a complex and changing world, in which parameters change and essential processes are fused with accidental or irregular ones.

If Metzler's theory were to be tested it is expected that the following specific problems would be encountered:

(1) There is no means available of determining either the size or the degree of stability of the accelerator or the expectations coefficient. Moreover, only ex post facto observations can be made of the relationships between income and consumption, thus making it impossible to compute in advance the value of the marginal propensity to consume. An additional complication involving parameters is that there is no means of determining the duration of lags between the receipt and the spending of a given income dollar and, consequently, no means of determining directly the duration of a "period" as Metzler visualizes it in his model.

(2) The problem is rendered especially difficult by the fact that Metzler is concerned with inventory, rather than business cycles; his theory is directed toward understanding the process by which an economy seeks a new equilibrium. He recognizes that the business cycle is a complex phenomenon and that the establishment of a single causal relationship for any given cycle is unlikely.⁵ Yet the data needed to establish or to disprove the Metzler theory bear the stamp of the additional forces present in the business cycle itself.

(3) Associated with this difficulty is the fact that Metzler does not elaborate a theory of durable-goods investment. He makes only brief reference to it, stating that his models may be readily abridged to remedy the omission by combining the induced demand for invest-

1

Metzler, "Business Cycles and the Modern Theory of Employment," p. 282;

ment goods with the demand for consumers goods in a general "propensity to spend."

There are, however, at least three grounds for suspecting that the relationship is more complex than Metzler implies:

(a) Evidence indicates that the lag between the decision to invest and the investment expenditure is longer than that which he hypothesizes for the decision to consume (derived immediately from income) and consumption expenditure.⁶

(b) Inventory requirements arising out of changes in durables investment are likely to be smaller than those in consumption expenditures. An examination of institutional arrangements supports this view: In the marketing of producers investment goods, the channels from manufacturer to ultimate purchaser are largely direct, with no middlemen's stocks held. Moreover, it is well established that such investment goods as heavy machinery are produced principally to order, and that manufacturers' finished-goods requirements for such products are small.

(c) It is unlikely that durable-goods investment is derived merely from changes in income. Very likely it is influenced importantly by a number of other factors, including the existing stock of equipment, changes in technology, changes in population, changes in costs of production, and changes in availability of funds.

(4) Metzler makes simplifying assumptions regarding price movements and supply conditions. He assumes that prices remain unchanged and that supply is perfectly expansible. In fact, most prices have a degree of cyclical sensitivity, and it is reasonable to suppose that price movements influence businessmen's anticipations and inventory objectives. Moreover, supply conditions deteriorate during certain stages of the business cycle and improve during others, and these changes influence inventory objectives and the purchasing process.

Although it appears unlikely that conclusive tests of Metzler's theory can be devised, it stands as an hypothesis of major importance: (a) It suggests that attempts to adjust inventories may, because of income feedback, be partially self-defeating, thereby providing a cumulative force in expansion and in contraction; (b) it suggests a way in which movements in inventory investment may be reversed prior to reversals in aggregate demand.

Is there, then, any evidence which will permit at least a tentative judgment as to the validity of the Metzler thesis in explaining the nature of forces at work in the cyclical process? Some such evidence may be found in the behavior of consumption and saving, and in timing comparisons of turns in inventory investment with turns in rates of change in final purchases.

#### BEHAVIOR OF CONSUMPTION AND SAVING

During the postwar years a number of studies have been undertaken to determine the relations of consumption or saving to income and other

⁶ Victor Zarnowitz has compiled series which portray both aggregate fixed investment commitments (equipment orders placed and industrial construction contracts let) and aggregate fixed investment expenditures. During the period covered by these series (1949-58) timing comparisons reveal a systematic lag between orders and expenditures ranging between 6 and 12 months, with the exception of a lag of 30 months associated with the unusual demand and supply conditions arising out of the Korean war. Victor Zarnowitz, "The Timing of Manufacturers' Orders During Business Cycles," Business Cycle Indicators, New York, Princeton for NBER, pp. 476-477.

variables, and to measure the stability of such relationships.⁷ While, to a considerable extent the conclusions drawn from these studies remain controversial, there seems to be general agreement on at least the following:

1. The relationship between consumption and income is markedly less stable for short periods than for long.

2. Consumption is significantly related to a number of variables other than current income. These variables include liquid asset position, past income (or some concept of permanent income), and expectations regarding future income and prices.

In the years since the war consumer spending has been maintained at high levels for a number of months after each decline in aggregate activity.⁸ Moreover, there is evidence that on occasions such stability existed before the war. Unpublished quarterly estimates prepared for the National Bureau of Economic Research by Harold Barger indicate that consumption increased during the recession of 1923-24 and remained virtually unchanged during the recession of 1926-27.

What are the implications of these findings? Certainly it can be held that the behavior of consumption during minor business cycle recessions (the sort which have characterized the postwar period) is distinctly different from that described by Metzler. If the marginal propensity to consume is very low (i.e., if the response of consumption to a decline in aggregate income is very small), the feedback process which is the essential element of Metzler's theory will be broken or become relatively unimportant. The expansion would terminate at the point at which inventories are brought into the desired relationship with sales. A decline in inventory investment, and thus in output, would follow; but an equilibrium adjustment of aggregate income would be readily attained, since the decline in investment would not set off a downward movement in sales.

Under such circumstances, the full extent of the observed declines in both inventory investment and inventories proper would not be explainable in terms of the Metzler process. Such declines might be due to a correction of earlier speculative overcommitment in stocks and to the downward revision of inventory objectives (i.e., a reduction in the desired inventory-to-sales ratio) arising out of changes in supply conditions. They might be due also to the adjustment of stocks to an independently determined decline in the production of durable capital goods. This latter cause would appear to be a partial explanation at best, for marked inventory disinvestment during recession is characteristic of all sectors of industry, and of wholesaling and retailing as well.

In regard to expansion periods, the findings which show consumption to be influenced by factors other than current income, and which

Tothe to be initiate of the transmission of the series of the transmission of the tra

demonstrate a very considerable degree of short-run instability in the consumption function, argue that Metzler's thesis is an abstraction which neglects significant relationships and provides, at best, a very imperfect description of the forces at work.

This is not to say, however, that the income effect is of no importance during expansions. The marginal propensity to consume may be unstable, but aggregate consumption usually does rise with rising Moreover, business spending also rises. Under such conincomes. ditions it is to be expected that efforts to increase inventories will be self-frustrating to a degree, and the process of inventory investment may, in some measure, feed upon itself. If all planned inventory investment is not realized, then investment demand has contributed more significantly to the cumulative forces of expansion than the data This does not mean that these expansions could have been show. sustained by a Metzler-type process alone. But it is a suggestion that the demand arising out of attempted investment is probably well in excess of that which can be measured from the (realized) inventory investment data.

### TIMING OF INVENTORY INVESTMENT AND RATES OF CHANGE IN FINAL PURCHASES

In Metzler's theory, not only must final purchases respond to changes in aggregate output and income, but the response must be large enough to defeat inventory objectives significantly. During the expansion phase increased final purchases, along with such inventory investment as may be realized, bear aggregate activity The peak in the rate of inventory accumulation cannot be upward. attained until the rate of increase in final purchases has begun to sub-Similarly, it is the unanticipated decline in final purchases side. during the contraction phase, along with realized inventory disinvestment, that carries aggregate activity downward. The highest rate of inventory liquidation cannot be attained until the rate of decrease in final purchases has begun to subside, for only then may adjustment of stocks reach a maximum.9

We have already had occasion (ch. 4) to observe the timing relationship between movements in rates of change in manufacturing activity and in purchased-materials investment. A similar relationship exists: (1) between rates of change in manufacturing activity (either output or sales) and investment in manufacturers' total stocks, and (2) between rates of change in activity (sales) and inventory investment for both retailers and wholesalers (chart 17, table 36). In the manufacturing series, inventory investment lags behind rates of change in activity at every comparison. In the retailing and wholesaling series, the same lag occurs in 7 out of 10, and 6 out of 8 comparisons, respectively.

$$U_{t} = S_{t-1} - S_{t} = -(S_{t} - S_{t-1}) \tag{4}$$

and total inventory investment is:  

$$I_{t}+U_{t}=(1+a)(S_{t-1}-S_{t-2})-(S_{t}-S_{t-1})$$
 (5)

[•] The point may be demonstrated by reference to equation 2c (see note 4), which shows that planned inventory investment is dependent on the rate of change in sales with a one-period lag. Thus total inventory investment would show a one-period lag if it depended only on planned inventory investment. Actually it also includes unplanned investment (Ui).

Since the second term  $(S_t-S_{t-1})$  is increasing as long as the rate of growth of sales is increasing, total inventory investment is prevented from rising as fast as the rate of increase in sales during the period of acceleration. Thereafter, the offsetting effect of unplanned investment diminishes. Accordingly, the lag of investment behind the turning points in the rate of increase would, in general, be more than one period.

	[Current dollars]										
	Peak	Trough	Peak	Trough	Peak	Trough	Peak	Trough	Peak	Trou	ıgh
Quarter-to-quarter change final purchases (less services)	Aug. 1946 	Feb. 1949 +9 Feb. 1949 +6 Feb. 1949 +9 May 1949 0	Aug. 1950 +9 Aug. 1950 +3 Feb. 1951 0 Aug. 1950 -3	Aug. 1952 -3 Aug. 1951 +9 May 1951 +12 May 1951 +6	Nov. 1952 0 Nov. 1952 +6 Nov. 1952 +9 Aug. 1952 +3	Feb. 1954 -3 Nov. 1953 +9 Aug. 1953 +15 Nov. 1953 +9	Aug. 1955 +3 May 1955 +12 May 1955 +6 Nov. 1955 +3	Feb. 1956 ( ¹ ) Feb. 1956 Feb. 1956 +6 ( ² ) ( ³ )	Feb. 1957 (1) Nov. 1956 (1) Aug. 1957 -9 (3) (1)	Feb. Feb. Feb. Feb.	1958 (1958 +8 1955 1955 +3

TABLE 36.—Timing of selected inventory investment series to related purchase or sales series

¹ No matching turn. ² No turn.

1

ï

Source: Based on material from Department of Commerce.

• •

CHART 17

QUARTER-TO-QUARTER CHANGE IN FINAL PURCHASES AND SALES, AND INVENTORY INVESTMENT, SELECTED SERIES, 1946-58



ъ.



Shaded areas represent business contractions; unshaded areas, expansions. Dots identify peaks and troughs of specific cycles. Source: All series except change in average index of manufacturers' output were compiled from Department of Commerce data. Manufacturers' output series from Federal Reserve Board.

It should be noted, however, that there is not a completely consistent relationship between movements in the two types of series. Looking once again at the total manufacturers, total wholesalers, and total retailers data, it will be observed that there are two instances of this lack of correspondence. The first occurs early in the postwar period, when each inventory investment series shows a second cycle following a trough in the third or fourth quarter of 1947. There is no full-fledged corresponding cycle to be found in any of the series representing rates of change in activity, although each of these series do turn up for a single quarter at the end of 1947. In the second instance, the manufacturing and retailing activity series trace out an

76626-61-pt. I---10

"extra" cycle after forming a trough in the first quarter of 1956. The manufacturers' inventory investment series, however, moves downward from second quarter 1956 to first quarter 1958. The retailing investment series shows only a poorly defined and somewhat erratic cyclical movement beginning in the third quarter of 1956 and ending in the first quarter of 1958.

In the nonfarm inventory investment series, movements are found which conform in most instances to the pattern of quarter-to-quarter change in final purchases, in a manner similar to that observed between inventory investment and rates of change in activity for manufacturing, retailing and wholesaling (chart 17).¹⁰ During the first business cycle, inventory investment shows two distinct cycles. The first peak occurs within 3 months of the peak in the final purchases series, but the trough and peak which follow have no counterpart in the latter series. In the second business cycle there are once again two well-established cycles in inventory investment, but in this instance both conform to roughly similar movements in the final purchases series. In the third business cycle, inventory investment moves in conformity to changes in final purchases during approximately the first half of the expansion. Following the first quarter of 1956, a second cycle is noted in the final purchases series, but inventory investment moves steadily downward.

In contrast to the behavior of the manufacturing, wholesaling, and retailing series, nonfarm inventory investment does not show a welldefined tendency to lag behind rates of change in its corresponding purchases series. Turns in the nonfarm investment series coincided with final purchase turns in two of the eight comparisons, led by one quarter in three comparisons, and lagged by one to three quarters in the remaining three comparisons.

There is no present method to reconcile statistically the consistent lagging relationship of inventory investment to rates of change in activity in the manufacturing, wholesaling, and retailing series, with the irregular timing of total nonfarm inventory investment relative to rates of change in final purchases. It should be recognized, however, that the manufacturing and wholesaling activity series are not closely comparable to final purchases. The former series reflect, in part, variations in inventory investment of their customers as well as production or sale of goods destined for end use, whereas the final purchases data are free of inventory investment changes. Whatever may be the explanation of the differences, the aggregative data conform more closely to the concepts envisioned in the theory and must be given great weight.

Although these findings are inconclusive as regards the relevance of Metzler's theory, they provide the basis for a general conclusion of some importance: in spite of the lack of a well-established timing characteristic, movements in aggregate nonfarm inventory investment appear to be significantly related to movements in the rate of change in activity. There have been two postwar instances in which this has not been true, but the overall consistency of the relationship, despite its looseness, is impressive. There are no observations of timing

¹⁰ The final-purchases series used in this analysis excludes consumer purchases of services. It covers only spending for commodities, construction and Government (Government spending on services is included). It should be noted, however, that some inventory investment occurs in the service industries or in response to spending on services.

differences in nonfarm inventory investment and final purchases which are greater than 9 months.

1

The question arises, however, whether or not this observed conformity necessarily implies that the acceleration principle is at work. Bert G. Hickman takes the position that it does not.¹¹ In an analysis of the role of the acceleration principle in business cycles he shows that most of the retardation in manufacturing production and sales and in retail sales (observed above) appears to be accounted for by diffusion, i.e., by actual expansions or contractions in individual industries or retail store classifications. Such a finding leads him to conclude that "acceleration-induced inventory investment may be a comparatively unimportant initiating (as contrasted with amplifying) factor in business contractions."¹²

It is important to note that the emphasis here is on the words, "acceleration induced." His evidence does not rule out other models in which inventory investment is an initiating factor (one such model is presented later in this chapter). Moreover, until he examines the data for individual industries, his evidence against even accelerationinduced investment is indirect.

### BEHAVIOR OF INVENTORY INVESTMENT UNDER CONDITIONS OF CHANGING INVENTORY OBJECTIVES

One major gap in Metzler's theory has not been examined: its failure to treat the influence of supply conditions and the ordering process on the determination of inventory behavior. In the economic world of Metzler's models there is only one stage of economic activity, and it is assumed implicitly that all sales are made for immediate Firms have no suppliers; there are no purchase orders and delivery. no backlogs of unfilled orders. The entrepreneur's output plans are guided by his anticipated sales to consumers and by his inventory The latter (except in the model in which stock objectives objectives. are held constant) are based entirely upon the level of sales anticipated in the subsequent period.

In practice, however, inventory objectives are influenced by other factors. In chapter 4 it was shown that order backlogs vary cyclically, especially in the important durables group, and that sellers' willingness to hold purchased-materials stocks is thereby affected. Moreover, these fluctuations in order backlogs are indicative of changing supply conditions which alter buyers' desired inventory target levels. Such changes in supply conditions have a well-defined cyclical pattern.¹³

Thus we find that forces operating through supply conditions act upon individual manufacturers in their roles as both buyers and sellers, causing them to attempt to adjust their stocks. These changes in inventory demand are superimposed upon cyclical changes in final demand. Moreover, they are likely to both set up price movements and to be accentuated by them. In a sense the result is a sort of

¹¹ Bert G. Hickman, "Diffusion, Acceleration, and Business Cycles," American Economic Review, September 1959, pp. 551-558. ¹³ Ibid., p. 588.

¹³ Ibid., p. 588.
¹⁴ As indicated earlier the pattern was outlined as follows:

Availability of materials reaches its maximum at approximately the trough of the business cycle and begins to deteriorate with the beginning of recovery, or very shortly thereafter. Supply conditions deteriorate at an accelerating rate during early expansion and then at a diminishing rate.
2. By midexpansion, supply conditions begin to improve, although high levels of unfilled orders attest to substantial delays. Improvement may proceed at varying rates, but it is continuous throughout the latter half of the expansion and during early recession, availability improves at an accelerating rate. The maximum rate of improvement is attained by midrecession.

accelerator, but the process is quite different from that visualized in Metzler's theory.

# THE OCCURRENCE OF "EXTRA" INVESTMENT CYCLES

It will be recalled that during the 1945–48 and 1949–53 expansions there were additional upward movements in inventory investment not matched by a movement in rates of change in final purchases, but that in the 1954-57 expansion there was not. Is it possible that differences in the availability of materials may provide a key to explaining inventory investment behavior on these occasions? Such evidence as does exist is indirect, but it points to supply conditions as a significant factor in determining the response of inventory investment movements to changes in final purchases during the latter half of expansion periods. In the 1945–48 expansion, the durable-goods Durables sector adjusted quickly to immediate postwar demand. unfilled orders began to decline after October 1946. The ratio of unfilled orders to sales reached a peak of 5.99 in the first quarter of 1946; by the second quarter of 1947 it had fallen to a level of 4.10, and continued to fall thereafter (see chart 7).¹⁴ Under these conditions, there was no significant upward movement in purchasedmaterials investment after mid-1947. In fact, the deflated data show continuous disinvestment.

This behavior is in complete contrast to that of the nondurables sector during the same period. For the latter, the immediate postwar period was one of continued strong demand and limited capacity. Unfilled orders rose to near-peak levels in early 1947, and remained virtually unchanged throughout the remainder of the year. Unfilledorders-to-sales ratios are not available prior to 1948 but in the first quarter of that year the ratio stood at 1.78, a level which was subsequently exceeded only in first quarter 1951. It was under these conditions of sustained high demand and short supply that nondurable purchased-materials inventory investment moved upward in a second expansion during the second half of 1947.

In the  $1949-\overline{53}$  expansion, it was the nondurables sector that adjusted quickly to demand conditions. A flood of orders immediately followed the outbreak of the Korean war, and the unfilled-orders-tosales series attained a peak level of 1.91 in first quarter 1951. But unfilled orders began to decline following the peak in March 1951, and by the fourth quarter of 1952 the ratio had fallen to 1.10. Thus, the initial upsurge in purchased-materials investment was short lived. Investment reached its peak in fourth quarter 1950, and there was no upward movement thereafter.

In the durables sector, demand outstripped supply during almost the entire expansion period. Unfilled orders rose until September 1952 and remained at a high level until mid-1953. The unfilled-ordersto-sales ratio reached a maximum of 7.02 in August 1952. In contrast to the nondurables, durables purchased-materials investment moved upward during the latter half of 1952 and first half of 1953.

In the 1954–57 expansion, neither durables nor nondurables showed a record rise in unfilled order backlogs. The nondurables unfilled-

¹⁴ Absolute levels of unfilled orders to sales differ substantially between the durables and nondurables categories, the former being much higher, due to longer periods of production and a much larger percentage of business done on order. In the postwar period durables ratios have ranged between 6.8 and 2.41, non-durables between 1.8 and 0.85. It is the relative movements in these ratios, not their absolute levels, which are significant.

orders-to-sales ratio reached a peak at 1.11 in the fourth quarter of 1955 and fell continuously thereafter. Unfilled orders rose for a longer period in the durables sector, but the levels of order backlogs never reached the peak of the preceding expansion. In third quarter 1956, the peak ratio of 4.43 was attained.

In the same period, total manufacturers' purchased-materials investment failed to respond to the sharp upward movement in rates of change in final purchases. Unfortunately, there are no deflated data subsequent to mid-1956, and the behavior of the durable and nondurable categories cannot be ascertained with certainty. In the undeflated data the peak in durables purchased-materials investment was reached in third quarter 1955, and there was no second cycle thereafter. The nondurables undeflated series shows a gentle rise to a peak in second quarter 1957. The inventory investment movement had but little amplitude, however, and it is possible that deflated data would show no significant movement.

To summarize: the first two postwar expansion periods were characterized by double cycles in total manufacturers' purchased-materials inventory investment. In contrast, there was no second movement in the 1954-57 expansion, in spite of the well-defined upturn in rates of change of final purchases in 1956. The two "extra" cycles were not generalized throughout the manufacturing sector, but are traceable to the nondurables group in the first instance and to the durables group in the second. These movements occurred only under conditions of relatively high levels of unfilled orders.

There is evidence here that general availability of materials influences the behavior of inventory investment in the later stages of expansion, when the upward movement in final purchases continues because certain types of final expenditures still rise; under conditions of "tight" supply, there may be a resurgence of inventory investment, whereas under conditions of ready availability there will not.

This is by no means a solid finding. I am not able to establish the precise relationship between supply and inventory investment. Moreover, I generalize from the behavior of purchased materials to that of total nonfarm inventory investment. Nevertheless, the observed behavior is interesting. It suggests the hypothesis that after the initial surge in inventory investment, the relative availability of materials will determine whether or not it may once again become a source of strength during the expansion.

### THE RELATIVE IMPORTANCE OF CHANGES IN INVESTMENT DURING RECESSION

In chapter 2, it was pointed out that movements in inventory investment were greater, relative to changes in national product, during early than during later expansion, and were greatest during recession. These observations may readily be interpreted in the light of observed supply conditions. Shifts in the vendor-performance and purchasing-policy series are most abrupt during the first half of expansion phases and during contractions (chart 9). Inspection of these data indicates that changes in supply conditions affect the initial stages of expansion and contraction about equally, but the expansions having been much longer, the effect is weakened during the latter part of these phases. In addition, inventory movements are larger relative

to change in national product during recession because final purchases, particularly consumption, remain relatively stable in this phase but rise during expansion. The interesting point is that inventory investment can show such sharp declines during recessions under conditions of relatively stable consumption.

It would appear that these declines are due in part to the abrupt shifts in purchasing policy already noted. When purchase based on requirements extending well into the future is changing abruptly to purchase based on immediate needs, it seems likely that firms are cutting inventory objectives drastically and attempting to trim stocks. The observed behavior of vendor-performance and purchasing-policy series provide evidence to support this thesis. Thus, in Metzler's terminology, there occurs a sharp increase in the accelerator (i.e., an increase in the amount of inventories to be liquidated for a given decline in sales or output) resulting in a sharper inventory disinvestment than would otherwise occur under conditions of relatively slight declines in final purchases.

### FACTORS CONTRIBUTING TO TURNING POINTS

The present section is concerned with the manner in which inventory investment may contribute to business cycle turning points.

In early expansion, as sales rise, purchase orders also rise sharply. Two reasons for this rise in orders have already been discussed: (a) The firm places orders to meet rising sales requirements; and (b) the firm places orders to increase the volume of stocks on hand. A third reason may be added: (c) The firm may place orders simply to increase its order commitments on the supplier's books. At a time when delivery periods are being extended, the purchaser will realize that his order may be desirable, not only to build up his stocks, but also to provide him with additional protection against the possibility of price increases or against being "uncovered" at some future date. Thus the purchasing firm looks not only to its stock on hand or in transit from supplier in assessing the degree to which it is protected; it looks also to its "goods on order." It is this "total ownership position" along with sales requirements which guides purchasing policy.¹⁵

Throughout the buying movement, firms increasingly place orders in excess of their requirements in order to provide themselves with this kind of protection. At some point the ownership position approaches a "satisfactory" condition (given existing anticipations as to sales, prices, and vendor performance) and orders tend to recede toward a level which covers sales alone. It is highly likely that such a decline will contribute to the termination of the expansion.

It is conceivable, of course, that sales will continue to increase; additional purchase orders to cover sales requirements and, perhaps, to keep the ratio of stocks and orders to sales at a satisfactory level, may be enough to keep total orders rising. But the increase in demand would need to be quite sharp; presumably it would stem from some vigorous new source, such as a shift in the propensity to consume or a sharp increase in investment spending.

On the other hand, if something were to cause delivery periods to shorten, or anticipations of vendor performance to improve, the in-

¹⁴ I am indebted to Ruth Mack for introducing me to this concept and stressing its importance. See Ruth P. Mack and Victor Zarnowitz, "Cause and Consequence of Change in Retailers' Buying," American Economic Review, March 1958, pp. 27ff.

ventory objective and the desired ratio of ownership to sales would be reduced. Under these conditions, it would be highly probable that orders would fall. Such changes are easily visualized: delivery conditions might improve as a result of expansion of capacity as cyclical expansion goes on. Or, as has already been suggested, a decline in the current rate of deterioration of vendor performance may affect anticipations favorably.

Given the initial downturn in orders and sales, the decline in purchasing orders proceeds during recession under the combined impetus of forces paralleling those noted during expansion: (a) there is a decline in sales requirements, (b) the firm reduces orders in order to reduce the volume of stocks, and (c) the improvement in vendor performance, leads to a reduction in the desired ratio of outstanding orders to sales. As recession deepens, purchasing policy shifts very sharply, and within a relatively short period of time a majority of firms are purchasing on a short-range basis. When this occurs, inventory policy is no longer influenced significantly by considerations of delays in delivery. The target level for stocks is then related principally to anticipated sales requirements.

At this point it is to be expected that forces (b) and (c) are no longer operative and that inventory objectives will no longer decline. Orders must turn up unless the decline in final sales alone is sufficient to decrease them further. If the only reason for the decline in total sales and output has been the attempt to achieve the goals mentioned under (b) and (c) (that is, if final sales were constant), then the level of orders and of activity may be expected to rise. This seems to approximate what has occurred during postwar recessions. Consumption expenditures have been relatively insensitive to declines in national income, and Government expenditures have tended to be stable or to rise. As a result, the postwar recessions have been characterized by an early upturn in inventory investment. Of course, if final purchases were sensitive to declines in income, in the manner assumed by Metzler, the decline in inventory investment would set up a cumulative process and the trough would be reached under conditions described at the beginning of this chapter.

This discussion of the impact of changing supply conditions has been stated in terms of manufacturers' purchased-materials inventory policy alone. Chapter 6 indicates that the analysis can probably be extended to certain types of goods in process, since these stocks, when held between stages, behave similarly to purchased materials (see The behavior of finished goods must be explained in chart 16). somewhat different terms, but we have seen that investment in these stocks does not lag behind that of purchased materials and goods in process by many months (ch. 7, table 33, chart 16). No confident statement regarding retailers' and wholesalers' stocks can be made, since this study has dealt with them only in passing. On an a priori basis, however, the analysis would appear to be applicable, for distributors, to the extent that they purchase from manufacturers, have as much reason to respond to changing supply conditions as do manufacturers' purchasing agents.

#### CONCLUSIONS

It is now possible to bring together certain observations made in this chapter and to draw conclusions regarding their usefulness in explaining cyclical fluctuations. For this task it is convenient to arrange the various points according to the light they shed upon the two questions posed previously. How do changes in inventory investment contribute to the cumulative forces of business cycle expansion and contraction? Do movements in inventory investment spark the upswing and bring about the downturn, thereby constituting an underlying cause of business cycles as well as an aggravating force?

There is no doubt that inventory fluctuations contribute to the cumulative movements which constitute business cycles. Both the prewar and postwar data show clearly that changes in the rate of realized inventory accumulation are an important source of increase in demand during the greater part of expansions, and are a principal source of decline in demand during contractions. (See ch. 2.) Moreover, it is probable that the contribution is greater than may be observed in these data, since they show only realized inventory investment. One of the great contributions made by Metzler is his conclusion that adjustments in inventory contribute more to instability than may be observed, because they set up increases or decreases in final purchases which, in part, defeat the attempted adjustments. It would appear, however, that this income effect has been more important in expansions than contractions.

Another source of instability has been traced to the interaction between availability of materials and purchasing policy. An important reason for the cumulative force developed by buying movements is that purchasing policy alters with any increase in activity: The volume of orders rises not only to provide for sales, but to provide for inventories and to increase the time-range of advance purchasing. Increased demand gives rise to a deterioration of supply conditions which in turn, causes firms to raise their inventory objectives and their commitment to suppliers. The process occurs in reverse during the contraction phase.

This interaction alone could not long sustain a buying movement or a curtailment phase, but coupled to an income effect, speculative tendencies, or effect on plant and equipment investment (particularly through expectations) the result is a mechanism capable of generating very powerful cumulative forces. It is fortunate that the process has within it the means of its own termination within a relatively short period of time, and that the income feedback effect is limited during recession. Otherwise, business cycles would be much more severe than we have known them to be in recent times.

Turning points in planned inventory investment must be regarded as reversals in demand. Ceteris paribus, when firms cease to attempt to accumulate stocks at as high a rate as previously, their purchases from suppliers decline; when firms cease to attempt to reduce stocks at as fast a rate as previously, their purchases increase.

According to Metzler, the explanation of turning points in the business cycle lies in this process.¹⁶ Turns in inventory investment

¹⁶ For the purposes of his analysis, Metzler treats the business cycle as an inventory cycle. He fully realizes, of course, the complexity of business cycle relationships.

occur after rates of change in activity have reached a peak or trough, thereby permitting accumulation or liquidation of stocks to proceed at a rate more closely approximating the desired level.

Timing comparisons of turns in nonfarm inventory investment and final purchases show that, on a number of occasions, investment has turned earlier than would be expected in Metzler's theory. Under existing institutional arrangements, however, there is another way in which inventory investment turns can come about. Management considers not only stocks on hand when examining its inventory position, but also its outstanding purchase orders. As expansion proceeds, the quantity of goods on hand and on order increases. Ultimately, management must consider itself sufficiently protected and, other things being equal, the volume of orders which it places with suppliers will decline.

The observation that inventory investment has led all three postwar business cycle peaks and two of the three business cycle troughs lends support to the argument that turns in inventory investment play a role in sparking the upturn from recession and in halting the process of expansion. During expansion, continued increases in fixed investment and consumption expenditures seem characteristic of the modern business cycle, but the inventory requirements traceable to such expenditures are not likely to be large.

It is my conclusion that these observations are useful in explaining both the cumulative forces of expansion and contraction and the processes by which these phases may be terminated. This is not to deny the complexity of the forces which are combined in the business cycle. Evidence abounds that there are other contributory processes at work, some of which are also capable of causing cyclical reversals. But this, in turn, does not detract from the proposition that the processes which surround the accumulation and liquidation of inventories also contribute significantly to an explanation of cyclical fluctuations.

In a period such as the years since the war, when the underlying forces of growth have been strong and the recessions interrupting the general rise in national output have been only a year or less in duration, the inventory investment mechanism, as modified above, appears as a highly likely candidate for explaining turning points and, in part, the cumulative character of the business cycle.

### APPENDIX A

### SOURCES AND PROCESSING OF DATA

The principal data studied are Department of Commerce end-ofmonth estimates of the book value of manufacturers stocks.¹ Series for total inventories covered the 3 comprehensive groupings (total manufacturing, durable-goods industries, nondurable-goods industries) and 16 industry groups. Stage-of-fabrication series covered the comprehensive groupings and nine industry groups. All data were deflated by the author for the period 1947-56. Only material for the comprehensive groupings were studied for the period after 1956. These more recent data were not deflated. Prior to dating turns in the inventory series, the end-of-month figures were centered by use of a 2-month moving average. These data comprised the principal source for the analysis in

These data comprised the principal source for the analysis in chapters 3 through 7. In addition, chapters 2 and 3 make use of Department of Commerce inventory data for total nonfarm, total manufacturing, and durable-goods industries, deflated (in 1954 dollars) by the Department for purposes of computing the inventory investment component of gross national product. These figures cover the period 1947 to date and are roughly comparable to those deflated by the author; they differ only in that coverage and procedures used in deflation were not identical.

Finally, activity (shipments or output) and finished-goods inventory data for 25 commodities were analyzed in chapter 5. These figures, shown in table 21, required seasonal adjustment only.

### DEFLATION PROCEDURE.

The approach used in deflating the book value series was as follows: 1. In constructing deflators the Bureau of Labor Statistics Wholesale Price Indexes were used. No deflation was attempted before 1947 since the Wholesale Price Indexes were revised in that year.

1 The Department of Commerce book value series were as follows:

Total-stocks series	Stage-of-fabrication series
Total manufacturers	Total manufacturers.
Durable-goods manufacturers	Durable-goods manufacturers.
Nondurable-goods manufacturers	Nondurable-goods manufacturers.
Primary metals	Primary metals.
Fabricated metals	
Electrical machinery	Total machinery.
Machinery, excluding electrical	<b>m</b> + <b>t</b> + + + + + + + + + + + + + + + + + + +
Motor vehicles and parts	Total transportation equipment.
Transportation equipment, excluding motor vehicles	<b>6</b> • • • • • • • • • • • • • • • • • • •
Stone, clay, and glass products	Stone, clay, and glass products.
Food and kindred products	rood and beverages.
A proval	
Lasthan and lasthan products	
Papar products	Paper products
Chamical products	Chamical products.
Printing and publishing	Onemical products.
Petroleum and coal products	Petroleum and coal products
Rubber products	Rubber products.

2. Inventories were deflated separately for each stage of fabrication in both the durables and nondurables classifications. The more aggregative series (i.e., total manufacturers, durable-goods and nondurablegoods industries; total purchased materials, total goods in process, total finished goods) were secured by adding the appropriate deflated components.

Purchased-materials and finished-goods deflators were built up from deflators constructed separately for each industry. In combining these, the weights assigned were based on the proportions of the stage-of-fabrication inventory held by the various industries according to census data.² (See table A-1 for industry weights.) The goodsin-process deflators were constructed by averaging the purchasedmaterials and finished-goods deflators.

3. Purchased-materials industry deflators were constructed by examining materials inputs to each industry as revealed by the Bureau of Labor Statistics interindustry study for the year 1947, selecting and weighing price indexes accordingly. Code numbers and weights of the price indexes chosen are presented in table A-2. Finishedgoods industry deflators were constructed by examining the composition of finished stocks within each industry according to the 1947 Census of Manufactures and selecting and weighing the price indexes accordingly. Code numbers and weights of the price indexes are shown in table A-2.

4. Prior to combining the industry deflators the months covered by the turnover period ³ were averaged for each industry. Thus, if the turnover period were 3 months, the adjusted industry deflator for the month of March would be secured by averaging the deflators in the months of January, February, and March.

5. There were difficulties in making allowance for the LIFO component of stocks and for the fact that certain companies make cost or market adjustments at the end of fiscal periods. In general, the procedure used was to remove, monthly or quarterly, a fixed LIFO component and deflate the remainder, adding the LIFO component to the deflated portion to secure the final deflated stock value. The value of the LIFO component was taken from published estimates by Daly in the Survey of Current Business (July 1953, p. 17). No cost or market adjustments could be made in monthly series, but year-end deflations were checked with comparable National Income Division deflations for which cost of market adjustments had been made.

6. The total stocks series for 16 individual industries, which are discussed in chapter 3, were deflated in a somewhat different manner; deflators were constructed as described above but with assumptions of a fixed composition of total stock by stage of fabrication.

52

² Finished-goods weights were based upon the composition of finished goods according to the 1947 Census of Manufactures. Purchased-materials weights were derived from 1953 Annual Survey of Manufacturers' material.

¹ A turnover period is a period of time during which the stocks held as of a given date are presumed to have been accumulated. Information concerning the duration of these periods for the several industries was secured from the Department of Commerce.

# TABLE A-1.—Weights and turnover periods assigned to industry deflators DURABLE-GOODS INDUSTRIES

1. 	Turnover	Weights			
Industry	period (in months)	Purchased materials	Finished goods		
Primary metals	4 4 6 4 4 4 4	20. 6 16. 7 29. 2 21. 2 8. 1 4. 2	$\begin{cases} 15.3 \\ 14.4 \\ 41.0 \\ 0.5 \\ 3.0 \\ 6.5 \end{cases}$		

### NONDURABLE-GOODS INDUSTRIES

Food and beverage Tobacco products Textile mill products Apparel Leather products Paper Chemicals and allied products Petroleum and coal products Rubber and rubber products	3 6 (3) 1 4 4 3 4 3 4 3 4 3 4	22.7 13.6 12.2 8.8 2.9 9.8 19.8 6.8 3.4	41.0 9.9 8.0 9.4 2.5 2.6 13.7 9.6 3.3
------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------	-----------------------------------------------------------------	---------------------------------------------------------------

¹ Figure enclosed by parenthesis relates to finished-goods turnover period; other figure relates to purchased materials turnover period only.

+ 1

	Purchased	l materials	Finished goods			
Industry	Index numbers	Weight	Index numbers	Weight		
Primary metals	10-11 05-2 10-13 10-12 10-14 10-23	6 10 27 8 29 4	10-14 10-15 10-22 10-24 10-25	59 11 9 9 12		
Fabricated metals products	10-22 10-24 10-14 10-25	13 3 85	 10-4 10-6 10-7			
Total machinery	06-21 06-73 08	3 2 5	10-8 11-7 12-5 11	27 17 8 63		
Total transportation equipment	10-14 10-15 10-2 10-81 10-82 03 07-2 08 10-14 10-15	25 24 26 8 7 4 8 6 32 26	12-4  			
Stone, clay, and glass products	10-2 10-4 10-82 09-3 13-1 13-2 13-4 13-7	9 5 10 13 18 24 14 31				
Lumber and lumber products, furniture and fixtures	08	100	132 08 12-1	76 76 24		
NONDURABL	E-GOODS IN	DUSTRIES	I,,	<u> </u>		
Food and beverages Tobacco Textile mill products	01-3 01-52 01-1 01-2 02-5 02-71 01-32 01-4 03-31	9 5 14 47 14 11 100 78 	$\begin{array}{c} 02-2\\ 02-3\\ 02-4\\ 02-1\\ 102-5\\ 14-4\\ 02-8\\ 14-1\\ 03-1\\ 03-2\\ 03-3\end{array}$	20 5 24 8 10 25 8 100 48 30 22		
Apparei	03-12 03-24 03-33	40 38 22	03-5	100		
Leather and products Paper and allied products	04-2 09-1 09-2	100 94 6	04-3 04-4 09-3 09-5	35 34 22		
Chemicals and allied products	06 06–3 06–4	25 4 18	09-4 06-1 06-3 06-71	44 27 17 9		
Petroleum and coal products	06-7 06-22 01-73 05-56 05-20 09-7 05-56 05-12 07-1 03-12-76	5 5 28 5 5 65 35 71 29	05-4 06-21 06-7 06-5 	14 14 14 		

# TABLE A-2.—Index numbers and weights used in constructing industry deflators DURABLE-GOODS INDUSTRIES

.

105-5 (weight 100) minus 05-56 (weight 18) divided by 82.

# APPENDIX B

# **INVENTORY DATA**

· . .

# TABLE B-1.—Total manufacturing, inventories

[Billions of dollars, seasonally adjusted]

	Janu- ary	Feb- ruary	March	April	Мау	June	July	Au- gust	Sep- tem- ber	Octo- ber	No- vem- ber	De- cem- ber
		Total inventories (book value)										
1946 1947 1948 1949 1950 1951 1952 1953	18. 21 25. 28 29. 02 31. 91 28. 96 35. 30 43. 53 43. 96	18. 90 25. 81 29. 16 32. 08 29. 02 35. 94 43. 76	19. 27 26. 33 29. 48 31. 85 29. 19 36. 87 43. 93	19. 61 26. 95 29. 64 31. 50 29. 30 38. 17 43. 88	19.96 27.39 29.99 31.17 29.52 39.08 43.73	20. 37 27. 64 30. 27 30. 63 29. 75 39. 94 43. 32	21. 36 27. 80 30. 67 30. 19 29. 81 40. 78 42. 93	22. 11 28. 17 30. 98 29. 82 30. 12 41. 58 43. 13	22. 69 28. 19 31. 26 29. 34 31. 02 41. 91 43. 15	$\begin{array}{c} 23.\ 73\\ 28.\ 37\\ 31.\ 51\\ 29.\ 02\\ 31.\ 95\\ 42.\ 30\\ 43.\ 30\end{array}$	24, 19 28, 72 31, 66 28, 74 33, 38 42, 58 43, 46	24, 46 28, 87 31, 69 28, 86 34, 31 42, 82 43, 80
1953 ¹ 1954 1955 1956 1957 1958 1959 1960 1961	43. 93 45. 21 43. 15 46. 80 52. 43 52. 91 49. 49 53. 31 53. 67	44.00 44.88 43.14 47.56 52.92 52.44 49.92 53.90 53.60	$\begin{array}{r} 44.\ 25\\ 44.\ 53\\ 43.\ 22\\ 47.\ 98\\ 53.\ 33\\ 52.\ 01\\ 50.\ 45\\ 54.\ 34\\ 53.\ 31\\ \end{array}$	$\begin{array}{r} 44.56\\ 44.07\\ 43.24\\ 48.65\\ 53.66\\ 51.49\\ 51.05\\ 54.66\\ 53.88\end{array}$	45.04 43.77 43.54 49.32 53.91 50.90 51.60 54.95 53.37	45. 35 43. 65 43. 73 49. 62 53. 85 50. 25 52. 14 55. 10 53. 36	$\begin{array}{r} 45.54\\ 43.14\\ 44.08\\ 50.04\\ 54.09\\ 49.78\\ 52.24\\ 54.90\\ 53.58\end{array}$	46.06 42.88 44.77 50.38 54.20 49.42 52.12 54.98	46.05 42.65 44.86 50.84 54.17 49.30 51.89 54.71	45. 85 42. 98 45. 79 51. 75 54. 10 49. 34 51. 52 54. 38	45.76 43.18 45.92 52.21 53.87 49.30 51.62 54.01	45. 43 42. 98 46. 36 52. 30 53. 52 49. 18 52. 43 53. 74
	Total inventories (1947 dollars)											
1947 1948 1949 1950 1951 1952 1952	27.86 29.52 28.21 29.96 36.53 37.44	27.82 29.69 28.19 29.99 36.83	27.97 29.56 28.17 30.35 37.00	27. 32 28. 18 29. 45 28. 19 31. 18 37. 04	27. 69 28. 50 29. 30 28. 27 31. 94 36. 94	28. 02 28. 78 29. 15 28. 42 32. 82 36. 79	28. 24 29. 05 29. 05 28. 23 33. 84 36. 75	28. 54 29. 10 28. 84 28. 12 34. 70 36. 85	28. 43 29. 27 28. 60 28. 50 35. 25 36. 89	28. 27 29. 28 28. 36 28. 88 35. 76 37. 01	28. 30 29. 26 28. 09 29. 59 35. 92 37. 06	28. 21 29. 33 28. 24 29. 87 36. 04 37. 35
1953 1. 1954 1955 1956	37.50 37.65 35.92 37.53	37. 56 37. 37 35. 84 37. 95	37. 69 37. 11 35. 90 38. 09	37. 88 36. 78 35. 83 38. 42	38. 17 36. 56 36. 06 38. 79	38. 33 36. 50 36. 19 38. 96	38.30 36.08 36.29	38.55 35.81 36.71	38. 38 35. 57 36. 60	38. 14 35. 92 37. 11	38. 05 35. 97 37. 08	37.80 35.80 37.31
		Purchased materials (book value)										
1946 1947 1948 1949 1950 1951 1952 1953	8.10 11.23 12.18 12.88 11.01 14.91 16.56 15.74	8.28 11.46 12.28 12.84 11.05 15.26 16.57	8.53 11.62 12.37 12.46 11.04 15.71 16.47	8.77 11.79 12.48 12.22 11.04 16.15 16.34	8.86 11.97 12.62 12.00 11.18 16.28 16.26	9.07 12.02 12.92 11.72 11.35 16.55 16.00	9.47 11.91 13.04 11.51 11.66 16.72 15.86	9.83 11.96 13.02 11.43 12.07 16.79 15.76	10.04 11.95 13.06 11.36 12.69 16.56 15.77	10. 44 11. 96 13. 01 11. 17 13. 21 16. 76 15. 79	10. 72 12. 17 12. 89 11. 07 13. 88 16. 59 15. 83	11.00 12.28 12.77 11.08 14.48 16.42 15.87
1953 1 1954 1955 1956 1957 1958 1959 1960 1961	15. 76 15. 54 14. 50 15. 70 17. 24 17. 14 16. 27 17. 63 16. 77	$\begin{array}{c} 15.\ 66\\ 15.\ 25\\ 14.\ 41\\ 15.\ 92\\ 17.\ 42\\ 17.\ 10\\ 16.\ 47\\ 17.\ 75\\ 16.\ 69\end{array}$	$\begin{array}{c} 15.\ 71\\ 15.\ 01\\ 14.\ 36\\ 16.\ 05\\ 17.\ 45\\ 16.\ 93\\ .16.\ 74\\ 17.\ 81\\ 16.\ 70\\ \end{array}$	$\begin{array}{c} 15.\ 80\\ 15.\ 05\\ 14.\ 40\\ 16.\ 21\\ 17.\ 26\\ 16.\ 75\\ 17.\ 03\\ 17.\ 90\\ 16.\ 69\\ \end{array}$	$\begin{array}{c} 16.\ 09\\ 15.\ 08\\ 14.\ 59\\ 16.\ 44\\ 17.\ 34\\ 16.\ 57\\ 17.\ 37\\ 17.\ 93\\ 16.\ 75\\ \end{array}$	$\begin{array}{c} 16.\ 07\\ 15.\ 17\\ 14.\ 61\\ 16.\ 45\\ 17.\ 33\\ 16.\ 36\\ 17.\ 88\\ 17.\ 82\\ 16.\ 57\\ \end{array}$	$\begin{array}{c} 16.14\\ 15.05\\ 14.81\\ 16.57\\ 17.36\\ 16.21\\ 17.90\\ 17.68\\ 16.68\\ \end{array}$	16. 27 14. 84 15. 08 16. 59 17. 42 16. 15 17. 70 17. 58	16. 14 14. 68 15. 17 16. 68 17. 43 16. 14 17. 26 17. 31	15.91 14.53 15.53 17.08 17.56 16.37 17.00 17.12	15.83 14.60 15.50 17.20 17.49 16.23 17.04 16.83	15. 55 14. 38 15. 54 17. 20 17. 09 16. 07 17. 24 16. 80
	Purchased materials (1947 dollars)											·
1947 1948 1949 1950 1951 1952 1953	11. 62 12. 04 11. 05 12. 62 14. 04 13. 80	11. 64 12. 02 11. 02 12. 63 14. 09	11. 76 11. 78 10. 98 12. 82 14. 09	11. 95 11. 86 11. 65 10. 95 13. 13 14. 09	12. 12 12. 01 11. 53 10. 99 13. 24 14. 09	12. 21 12. 20 11. 38 11. 06 13. 49 13. 93	12.16 12.29 11.32 11.19 13.72 13.82	12. 18 12. 23 11. 32 11. 39 13. 94 13. 74	12.08 12.25 11.30 11.74 13.91 13.73	11. 93 12. 18 11. 15 11. 96 14. 17 13. 72	11. 96 12. 04 11. 08 12. 30 14. 10 13. 79	11. 88 11. 95 11. 11 12. 56 13. 94 13. 87
1953 1 1954 1955 1956	13. 83 13. 40 12. 46 13. 08	13.77 13.13 12.36 13.16	13. 81 12. 93 12. 31 13. 18	13. 85 12. 95 12. 32 13. 19	14.06 12.97 12.50 13.32	14.00 13.04 12.51 13.29	14.02 12.94 12.61	14. 04 12. 74 12. 80	13.88 12.58 12.81	13.65 12.47 13.05	13.60 12.55 13.00	13. 39 12. 39 12. 98

¹ Data revised January 1953.

132

# TABLE B-1.—Total manufacturing, inventories—Continued

			<b>L</b>					,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				
	Janu- ary	Feb- ruary	March	Åpril	Мау	June	July	Au- gust	Sep- tem- ber	Octo- ber	No- vem- ber	De- cem- ber
	Goods-in-process (book value)											
1946 1947 1948 1949 1950 1951 1953	5. 22 6. 64 7. 33 7. 61 6. 87 8. 82 11. 57 12. 92	5. 22 6. 76 7. 30 7. 60 6. 90 9. 01 11. 80	5.27 6.90 7.28 7.54 6.97 9.29 11.94	5. 34 7. 00 7. 30 7. 44 7. 05 9. 63 12. 01	5.59 7.02 7.32 7.37 7.16 9.83 12.10	5.78 7.06 7.36 7.27 7.32 10.05 12.13	5. 96 7. 12 7. 42 7. 25 7. 48 10. 19 12. 22	6.05 7.31 7.58 7.09 7.63 10.30 12.49	6. 21 7. 34 7. 62 6. 97 7. 84 10. 60 12. 60	6. 27 7. 33 7. 69 6. 89 8. 05 10. 79 12. 77	6. 34 7. 43 7. 73 6. 80 8. 30 11. 14 12. 67	6. 46 7. 43 7. 65 6. 82 8. 54 11. 34 12. 86
1953 ¹ 1954 1955 1956 1957 1959 1959 1960 1961	$\begin{array}{c} 2.96\\ 13.16\\ 12.44\\ 13.96\\ 15.76\\ 15.39\\ 14.38\\ 15.31\\ 15.15 \end{array}$	13. 19 13. 05 12. 44 14. 17 15. 86 15. 08 14. 38 15. 52 15. 69	$\begin{array}{c} 13.\ 37\\ 12.\ 88\\ 12.\ 54\\ 14.\ 38\\ 16.\ 01\\ 14.\ 83\\ 14.\ 50\\ 15.\ 69\\ 14.\ 95\\ \end{array}$	$\begin{array}{c} 13.54\\ 12.69\\ 12.52\\ 14.65\\ 16.37\\ 14.66\\ 14.70\\ 15.73\\ 15.00\\ \end{array}$	$\begin{array}{c} 13.58\\ 12.52\\ 12.62\\ 14.90\\ 16.39\\ 14.53\\ 14.83\\ 15.82\\ 15.10\\ \end{array}$	$\begin{array}{c} 13.\ 73\\ 12.\ 29\\ 12.\ 72\\ 14.\ 89\\ 16.\ 31\\ 14.\ 35\\ 14.\ 93\\ 15.\ 95\\ 15.\ 28\end{array}$	$\begin{array}{c} 13.\ 71\\ 12.\ 02\\ 12.\ 88\\ 15.\ 00\\ 16.\ 40\\ 14.\ 25\\ 15.\ 03\\ 15.\ 79\\ 15.\ 35 \end{array}$	13. 83 12. 08 13. 16 15. 09 16. 52 14. 24 14. 99 15. 77	13. 59 11. 97 13. 16 15. 23 16. 39 14. 24 15. 01 15. 48	13. 39 12. 24 13. 55 15. 60 16. 21 14. 21 14. 88 15. 35	13. 37 12. 47 13. 75 15. 86 16. 13 14. 22 14. 85 15. 26	13. 38 12. 45 13. 85 15. 73 15. 79 14. 32 15. 11 15. 14
	Goods-in-process (1947 dollars)											
1947 1948 1949 1950 1951 1952 1953	7.10 6.86 6.48 7.31 9.34 10.42	7.02 6.85 6.49 7.34 9.52	6. 96 6. 80 6. 52 7. 48 9. 63	7. 18 6. 95 6. 74 6. 57 7. 70 9. 68	7. 17 6. 95 6. 73 6. 65 7. 84 9. 76	7.20 6.94 6.70 6.74 8.03 9.79	7.25 6.96 6.74 6.83 8.15 9.90	7. 41 7. 04 6. 64 6. 89 8. 28 10. 09	7.38 7.01 6.56 6.97 8.54 10.18	7. 31 7. 02 6. 49 7. 04 8. 72 10. 31	7. 31 7. 01 6. 41 7. 15 9. 00 10. 21	7. 24 6. 92 6. 44 7. 22 9. 16 10. 34
1953 ¹ 1954 1955 1956	10. 44 10. 27 9. 66 10. 25	10. 62 10. 19 9. 64 10. 36	10. 73 10. 06 9. 69 10. 44	10. 85 9. 95 9. 66 10. 59	10. 85 9. 83 9. 72 10. 72	10. 94 9. 65 9. 78 10. 71	10. 85 9. 42 9. 84	10. 87 9. 44 10. 02	10. 63 9. 33 9. 93	10. 45 9. 52 10. 12	10. 42 9. 70 10. 21	10. 43 9. 66 10. 22
					Finish	ed good:	s (book	value)				
1946 1947 1948 1949 1950 1951 1952 1953	5. 28 7. 48 9. 43 11. 38 10. 99 11. 40 15. 25 15. 18	5. 49 7. 66 9. 51 11. 57 10. 99 11. 56 15. 30	5.50 7.81 9.65 11.68 11.03 11.74 15.38	5. 48 8. 06 9. 80 11. 71 11. 03 12. 16 15. 36	5. 51 8. 31 10. 01 11. 66 11. 03 12. 77 15. 17	5. 52 8. 52 10. 13 11. 64 11. 08 13. 33 15. 12	5.83 8.72 10.34 11.47 10.72 14.09 15.06	6. 16 8. 85 10. 44 11. 33 10. 50 14. 70 15. 07	6.46 8.93 10.74 11.18 10.65 15.04 15.02	6. 94 9. 07 10. 84 11. 12 10. 85 15. 03 14. 99	7.08 9.12 10.98 10.97 11.25 14.96 15.04	⁷ 7.14 9.28 11.26 11.02 11.31 15.05 15.08
1953 ¹ 1954 1955 1956 1957 1959 1959 1960 1961	15. 20 16. 50 16. 21 17. 14 19. 44 20. 38 18. 84 20. 37 21. 75	$\begin{array}{c} 15.\ 15\\ 16.\ 58\\ 16.\ 29\\ 17.\ 47\\ 19.\ 64\\ 20.\ 26\\ 19.\ 06\\ 20.\ 63\\ 21.\ 82 \end{array}$	$\begin{array}{c} 15.\ 16\\ 16.\ 64\\ 16.\ 32\\ 17.\ 56\\ 19.\ 87\\ 20.\ 24\\ 19.\ 21\\ 20.\ 84\\ 21.\ 67\\ \end{array}$	$\begin{array}{c} 15.\ 23\\ 16.\ 32\\ 16.\ 32\\ 17.\ 79\\ 20.\ 04\\ 20.\ 08\\ 19.\ 32\\ 21.\ 03\\ 21.\ 69\\ \end{array}$	15.37 16.16 16.33 17.98 20.18 19.80 19.40 21.20 21.52	$\begin{array}{c} 15.55\\ 16.18\\ 16.40\\ 18.29\\ 20.21\\ 19.54\\ 19.33\\ 21.33\\ 21.51\\ \end{array}$	15. 69 16. 07 16. 39 18. 47 20. 29 19. 32 19. 30 21. 43 21. 55	15. 96 15. 96 16. 52 18. 70 20. 26 19. 04 19. 43 21. 64	16. 32 16. 01 16. 53 18. 93 20. 35 18. 92 19. 62 21. 92	16. 55 16. 21 16. 71 19. 08 20. 33 18. 76 19. 64 21. 91	16.56 16.10 16.68 19.15 20.25 18.85 19.73 21:91	16. 49 16. 16 16. 97 19. 37 20. 64 18. 80 20. 07 21. 81
	Finished goods (1947 dollars)											
1947 1948 1949 1950 1951 1952 1953	9. 13 10. 62 10. 68 10. 03 13. 14 13. 22	9. 16 10. 82 10. 68 10. 02 13. 22	9. 25 10. 98 10. 67 10. 04 13. 28	8. 19 9. 36 11. 05 10. 66 10. 35 13. 28	8. 40 9. 55 11: 04 10. 64 10. 85 13. 09	8.62 9.64 11.07 10.62 11.30 13.08	8.83 9.80 10.99 10.20 11.97 13.04	8. 95 9. 83 10. 88 9. 84 12. 49 13. 02	8. 97 10. 02 10. 74 9. 80 12. 80 12. 99	9.04 10.09 10.73 9.88 12.87 12.99	9.04 10.21 10.60 10.14 12.82 13.06	9.09 10.47 10.69 10.09 12.94 13.14
1953 ¹ 1954 1955 1956	13. 23 13. 99 13. 80 14. 20	13. 17 14. 06 13. 84 14. 43	13. 15 14. 12 13. 90 14. 48	13. 17 13. 88 13. 85 14. 65	13. 25 13. 76 13. 84 14. 75	13. 39 13. 81 13. 90 14. 96	13. 44 13. 72 13. 83	13. 64 13. 62 13. 89	13. 87 13. 66 13. 86	14. 03 13. 92 13. 95	14. 03 13. 73 13. 87	13. 98 13. 76 14. 10

[Billions of dollars, seasonally adjusted]

¹ Data revised January 1953.

Source: Department of Commerce; deflations by the author.
## TABLE B-2.—Durable-goods industries, total inventories

	Janu- ary	Feb- ruary	March	April	May	June	July	Au- gust	Sep- tem- ber	Octo- ber	No- vem- ber	De- cem- ber
					Total ir	ventori	es (bool	r value)				
1946 1947 1948 1949 1950 1951 1952 1953	8.55 12.47 14.27 16.01 13.95 17.31 23.32 24.61	8.99 12.76 14.29 16.23 14.05 17.71 23.57	9.29 13.06 14.39 16.16 14.13 18.10 23.76	9.52 13.32 14.46 15.97 14.20 18.70 23.87	9.72 13.62 14.53 15.70 14.34 19.38 24.00	10.06 13.87 14.69 15.35 14.57 20.08 23.70	10. 40 14. 00 14. 92 15. 11 14. 65 20. 70 23. 30	10. 77 14. 18 15. 16 14. 80 14. 77 21. 38 23. 55	11. 15 14. 19 15. 35 14. 47 15. 12 21. 84 23. 65	11. 49 14. 33 15. 50 14. 16 15. 53 22. 19 23. 89	11. 76 14. 34 15. 71 13. 89 16. 29 22. 51 23. 98	12.00 14.30 15.74 13.97 16.78 22.81 24.41
1953 1 1954 1955 1956 1957 1957 1958 1958 1960 1961	24. 55 26. 00 24. 08 26. 91 30. 63 30. 62 28. 11 30. 76 30. 76	24. 66 25. 72 24. 08 27. 45 30. 96 30. 27 28. 41 31. 26 30. 65	24. 97 25. 48 24. 20 27. 87 31. 18 29. 86 28. 92 31. 77 30. 30	25. 24 24. 99 24. 16. 28. 28 31. 46 29. 42 29. 36 31. 92 30. 15	$\begin{array}{c} 25.\ 61\\ 24.\ 70\\ 24.\ 32\\ 28.\ 74\\ 31.\ 57\\ 28.\ 98\\ 29.\ 73\\ 32.\ 07\\ 30.\ 15 \end{array}$	$\begin{array}{c} 25.\ 88\\ 24.\ 41\\ 24.\ 43\\ 28.\ 76\\ 31.\ 44\\ 28.\ 53\\ 30.\ 23\\ 32.\ 23\\ 30.\ 20\\ \end{array}$	26. 09 23. 97 24. 74 29. 00 31. 70 28. 31 30. 35 32. 05 30. 41	26. 48 23. 86 25. 23 29. 12 31. 74 28. 07 30. 14 32. 08	26. 49 23. 74 25. 38 29. 45 31. 82 28. 05 29. 82 31. 84	26. 44 23. 90 26. 09 30. 23 31. 75 27. 93 29. 25 31. 43	26. 41 24. 04 26. 23 30. 65 31. 51 27. 88 29. 35 31. 07	26. 24 24. 08 26. 66 30. 66 31. 15 27. 82 30. 08 30. 86
		·	·		Total in	ventori	es (1947	dollars)				
1947 1948 1949 1950 1951 1952	13.92 13.94 12.46 13.79 17.98	13.82 14.01 12.45 13.93 18.17	13. 74 13. 90 12. 46 14. 10 18. 31	13. 72 13. 72 13. 80 12. 49 14. 45 18. 37	13. 94 13. 73 13. 68 12. 61 14. 90 18. 42	14. 12 13. 80 13. 52 12. 76 15. 48 18. 32	14. 23 13. 86 13. 41 12. 75 16. 06 18. 21	14. 32 13. 84 13. 23 12. 74 16. 63 18. 34	14. 24 13. 83 13. 00 12. 88 17. 02 18. 38	14.26 13.79 12.79 13.10 17.34 18.49	14. 19 13. 80 12. 53 13. 49 17. 53 18. 47	14. 10 13. 84 12. 59 13. 66 17. 71 18. 72
1953 1 1954 1955 1956	18. 83 19. 28 17. 65 18. 45	18.90 19.09 17,62 18.73	19. 10 18. 94 17. 71 18. 92	19. 26 18. 58 17. 59 19. 09	19. 47 18. 38 17. 66 19. 34	19. 57 18. 14 17. 74 19. 31	19. 59 17. 78 17. 80	19.74 17.67 18.00	19.66 17.53 17.92	19.58 17.60 18.22	19.56 17.67 18.20	19.45 17.69 18.38
			·	·.	Goods-i	in-proce	ss (bool	value)		<u> </u>		·
1946 1947 1948 1948 1950 1951 1952 1953	3.56 4.83 5.15 5.37 4.69 6.33 8.82 10.15	3.58 4.92 5.07 5.38 4.72 6.49 9.05	3. 64 4. 98 5. 05 5. 37 4. 75 6. 70 9. 19	3.71 5.04 5.07 5.31 4.82 6.95 9.32	3.87 5.07 5.24 4.93 7.12 9.47	4.04 5.07 5.12 5.14 5.03 7.31 9.55	4. 18 5. 10 5. 17 5. 10 5. 12 7. 54 9. 58	4.30 5.22 5.27 4.95 5.24 7.72 9.76	4. 40 5. 24 5. 33 4. 83 5. 43 7. 98 9. 84	4. 44 5. 24 5. 38 4. 78 5. 58 8. 19 9. 95	4.56 5.26 5.41 4.70 5.80 8.46 9.88	4. 66 5. 21 5. 40 4. 72 5. 99 8. 63 10. 12
1953 ¹ 1954 1955 1956 1957 1958 1959 1969 1960	10. 22 10. 55 9. 76 11. 18 12. 76 12. 40 11. 38 12. 26 12. 07	10. 36 10. 44 9. 76 11. 37 12. 89 12. 11 11. 37 12. 47 12. 47 12. 05	10. 60 10. 28 9. 86 11. 62 13. 04 11. 90 11. 51 12. 66 11. 90	$\begin{array}{c} 10.\ 69\\ 10.\ 02\\ 9.\ 84\\ 11.\ 84\\ 13.\ 42\\ 11.\ 76\\ 11.\ 67\\ 12.\ 63\\ 11.\ 87\\ \end{array}$	10.72 9.88 9.91 12.06 13.44 11.61 11.81 12.70 11.91	10. 83 9. 68 9. 98 11. 98 13. 33 11. 41 11. 89 12. 80 12. 03	10. 87 9. 47 10. 13 12. 06 13. 50 11. 34 11. 92 12. 63 12. 06	11.01 9.51 10.32 12.16 13.55 11.28 11.91 12.60	10. 85 9. 43 10. 37 12. 34 13. 45 11. 32 11. 95 12. 39	10. 68 9. 64 10. 75 12. 65 13. 24 11. 26 11. 78 12. 25	10. 71 9. 79 10. 92 12. 92 13. 14 11. 27 11. 78 12. 12	10. 73 9. 84 11. 09 12. 78 12. 73 11. 31 12. 08 12. 06
		<u> </u>	<u> </u>	<u> </u>	Goods-i	in-proce	ss (1947	dollars)		<u> </u>	<u> </u>	<u> </u>
1947 1948 1949 1950 1951 1952 1953	5.00 4.66 4.19 5.06 6.80 7.74	4.89 4.66 4.20 5.11 6.98	4.83 4.64 4.21 5.22 7.08	5. 22 4. 82 4. 60 4. 25 5. 38 7. 17	5. 21 4. 78 4. 58 4. 34 5. 49 7. 29	5. 19 4. 80 4. 53 4. 39 5. 64 7. 36	5. 21 4. 81 4. 53 4. 44 5. 82 7. 40	5. 28 4. 84 4. 42 4. 50 5. 97 7. 52	5.26 4.82 4.32 4.60 6.17 7.56	5. 21 4. 79 4. 28 4. 68 6. 33 7. 62	5. 18 4. 76 4. 22 4. 79 6. 54 7. 55	5. 10 4. 72 4. 23 4. 87 6. 66 7. 73
1953 ¹ 1954 1955 1956	7.79 7.77 7.09 7.59	7.90 7.69 7.07 7.68	8.06 7.59 7.14 7.80	8. 11 7. 40 7. 10 7. 92	8. 10 7. 31 7. 13 8. 04	8. 15 7. 15 7. 16 7. 96	8.12 6.98 7.21	8. 16 6. 99 7. 29	8.00 6.91 7.25	7.86 7.04 7.43	7.88 7.13 7.50	7.90 7.16 7.56

[Billions of dollars, seasonally adjusted]

¹ Data revised January 1953.

## TABLE B-2.—Durable-goods industries, total inventories—Continued

[Billions of dollars, seasonally adjusted]

		•										
	Janu- ary	Feb- ruary	March	April	Мау	June	July	Au- gust	Sep- tem- ber	Octo- ber	No- vem- ber	De- cem- ber
				I	urchase	ed mate	rials (bo	ok valu	e)			<u> </u>
1946 1947 1948 1949 1950 1951 1952 1953	3. 25 4. 65 5. 13 5. 83 4. 58 6. 15 7. 51 7. 30	3. 29 4. 81 5. 16 5. 84 4. 62 6. 27 7. 56	3. 44 4. 89 5. 17 5. 68 4. 64 6. 41 7. 57	3.54 4.93 5.22 5.56 4.70 6.55 7.52	3.57 5.07 5.28 5.38 4.80 6.70 7.41	3. 65 5. 14 5. 40 5. 16 4. 93 6. 94 7. 21	3.80 5.18 5.46 5.00 5.06 7.09 7.05	3. 94 5. 20 5. 47 4. 93 5. 20 7. 20 7. 09	4. 12 5. 13 5. 52 4. 86 5. 39 7. 25 7. 14	4. 31 5. 16 5. 57 4. 72 5. 54 7. 39 7. 25	4. 46 5. 14 5. 66 4. 60 5. 82 7. 42 7. 29	4.57 5.19 5.65 4.62 6.10 7.42 7.35
1953 1 1954 1955 1956 1957 1958 1959 1960 1961	$\begin{array}{c} 7.\ 29\\ 7.\ 38\\ 6.\ 64\\ 7.\ 52\\ 8.\ 64\\ 8.\ 35\\ 7.\ 71\\ 8.\ 60\\ 8.\ 03\\ \end{array}$	$\begin{array}{c} 7.23\\ 7.25\\ 6.62\\ 7.71\\ 8.72\\ 8.29\\ 7.84\\ 8.67\\ 8.01 \end{array}$	7.26 7.14 6.60 7.85 8.71 8.71 8.12 8.11 8.75 7.91	$\begin{array}{c} 7.39\\ 7.12\\ 6.61\\ 8.05\\ 8.60\\ 7.98\\ 8.34\\ 8.81\\ 7.81 \end{array}$	$\begin{array}{c} 7.63\\ 7.13\\ 6.64\\ 8.13\\ 8.49\\ 7.76\\ 8.52\\ 8.80\\ 7.78\end{array}$	$\begin{array}{c} 7.\ 67\\ 7.\ 17\\ 6.\ 61\\ 8.\ 13\\ 8.\ 37\\ 7.\ 58\\ 8.\ 86\\ 8.\ 69\\ 7.\ 60\end{array}$	$\begin{array}{c} 7.\ 72\\ 7.\ 00\\ 6.\ 82\\ 8.\ 25\\ 8.\ 39\\ 7.\ 46\\ 8.\ 90\\ 8.\ 63\\ 7.\ 73\end{array}$	7.92 6.89 7.02 8.17 8.38 7.44 8.74 8.56	7.79 6.83 7.15 8.25 8.51 7.54 8.35 8.42	7.68 6.66 7.38 8.50 8.61 7.71 8.05 8.26	7.62 6.65 7.33 8.63 8.57 7.64 8.11 8.05	7. 41 6. 49 7. 42 8. 66 8. 31 7. 52 8. 30 8. 05
				Р	urchase	d mater	ials (194	7 dollar	s)	L	·	
1947 1948 1949 1950 1951 1952 1953	4.98 4.97 4.07 4.84 5.73 5.49	4.97 4.98 4.08 4.86 5.76	4. 93 4. 84 4. 08 4. 91 5. 77	5. 10 4. 92 4. 76 4. 11 5. 00 5. 72	5. 23 4. 94 4. 65 4. 17 5. 10 5. 65	5. 28 5. 02 4. 51 4. 25 5. 28 5. 52	5. 30 5. 05 4. 42 4. 33 5. 40 5. 41	5. 28 4. 96 4. 39 4. 40 5. 49 5. 49 5. 42	5 16 4.92 4.35 4.50 5.54 5.43	5. 13 4. 88 4. 23 4. 57 5. 64 5. 48	5.05 4.89 4.12 4.74 5.66 5.40	5.08 4.86 4.12 4.88 5.66 5.53
1953 ¹ 1954 1955 1956	5. 48 5. 33 4. 72 4. 94	5. 43 5. 24 4. 70 5. 04	5. 44 5. 17 4. 68 5. 10	5. 52 5. 16 4. 66 5. 20	5. 67 5. 17 4. 67 5. 24	5. 66 5. 19 4. 64 5. 23	5.65 5.05 4.74	5.74 4.96 4.82	5.62 4.90 4.85	5. 52 4. 76 4. 94	5. 48 4. 74 4. 88	5. 34 4. 63 4. 91
			·		Finish	ed good	s (book	value)		·		
1946 1947 1948 1949 1950 1951 1952 1953	1.74 2.99 4.04 4.86 4.65 4.75 6.89 7.04	2. 12 3. 03 4. 08 4. 96 4. 62 4. 89 6. 88	2. 21 3. 19 4. 13 5. 02 4. 64 4. 96 6. 92	2. 27 3. 30 4. 16 5. 04 4. 61 5. 12 6. 93	2, 28 3, 42 4, 21 5, 04 4, 58 5, 44 6, 95	2.37 3.58 4.20 5.04 4.64 5.76 6.89	2. 42 3. 67 4. 25 4. 99 4. 50 6. 14 6. 84	2, 53 3, 73 4, 34 4, 92 4, 37 6, 55 6, 83	2. 63 3. 81 4. 44 4. 82 4. 34 6. 72 6. 82	2.74 3.94 4.53 4.75 4.50 6.78 6.84	2. 74 4. 02 4. 63 4. 65 4. 70 6. 73 6. 89	2.77 4.00 4.77 4.70 4.70 6.80 6.92
1953 1 1954 1955 1956 1957 1958 1959 1960 1961	7.05 8.06 7.68 8.20 9.23 9.88 9.01 9.90 10.65	7.06 8.03 7.70 8.36 9.35 9.86 9.19 10.12 10.59	7. 10 8. 07 7. 73 8. 40 9. 43 9. 84 9. 30 10. 36 10. 49	7.16 7.85 7.71 8.39 9.45 9.69 9.35 10.49 10.47	7. 27 7. 70 7. 76 8. 54 9. 64 9. 61 9. 40 10. 58 10. 47	7.38 7.56 7.83 8.65 9.73 9.54 9.47 10.74 10.57	7.49 7.50 7.79 8.69 9.81 9.52 9.53 10.79 10.62	7.55 7.47 7.89 8.78 9.79 9.35 9.49 10.93	7.85 7.48 7.85 8.87 9.86 9.20 9.52 11.03	8.07 7.60 7.97 9.08 9.90 8.96 9.42 10.92	8.09 7.60 7.99 9.10 9.80 8.98 9.45 10.90	8.10 7.75 8.16 9.22 10.11 8.99 9.71 10.76
			·!		Finish	ed good:	s (1947 d	ollars)				
1947 1948 1949 1950 1951 1952 1953	3. 94 4. 31 4. 20 3. 89 5. 45 5. 55	3. 95 4. 38 4. 17 3. 96 5. 43	3. 98 4. 42 4. 18 3. 98 5. 46	3. 40 3. 98 4. 43 4. 13 4. 07 5. 47	3.50 4.01 4.45 4.10 4.31 5.48	3.65 3.98 4.48 4.13 4.56 5.44	3. 72 4. 00 4. 46 3. 98 4. 85 5. 41	3.76 4.04 4.41 3.84 5.18 5.40	3.82 4.09 4.33 3.77 5.32 5.39	3. 92 4. 12 4. 27 3. 86 5. 37 5. 40	3. 96 4. 16 4. 19 3. 97 5. 33 5. 43	3. 92 4. 26 4. 24 3. 91 5. 38 5. 46
1953 ¹ 1954 1955 1956	5, 56 6, 18 5, 84 5, 92	5. 56 6. 16 5. 85 6. 01	5. 59 6. 19 5. 90 6. 01	5. 63 6. 02 5. 83 5. 98	5. 70 5. 91 5. 86 6. 06	5. 76 5. 80 5. 94 6. 12	5. 82 5. 76 5. 84	5. 84 5. 72 5. 89	6.05 5.72 5.81	6. 20 5. 80 5. 85	6. 20 5. 79 5. 82	6. 21 5. 90 5. 91

~

¹ Data revised January 1953.

Source: Department of Commerce; deflations by the author.

76626-61-pt. I----11

•

## TABLE B-3.—Nondurable-goods industries, total inventories

[Billions of dollars, seasonally adjusted]

Janu- ary	Feb- ruary	March	April	May	June	July	Au- gust	Sep- tem- ber	Octo- ber	No- vem- ber	De- cem- ber
			То	tal inve	ntories	(book v	alue)				
9.66 12.81 14.75 15.90 15.00 18.00 20.21 19.34	9. 91 13. 05 14. 87 15. 85 14. 97 18. 23 20. 19	9.98 13.28 15.09 15.69 15.06 18.77 20.18	10.09 13.63 15.19 15.54 15.10 19.47 20.01	10. 25 13. 78 15. 46 15. 46 15. 17 19. 69 19. 73	10. 30 13. 78 15. 58 15. 28 15. 18 19. 86 19. 62	10. 96 13. 81 15. 75 15. 08 15. 16 20. 08 19. 63	11. 34 14. 00 15. 82 15. 02 15. 36 20. 20 19. 58	11. 54 14. 00 15. 92 14. 87 15. 89 20. 07 19. 50	12.24 14.04 16.01 14.87 16.42 20.12 19.41	12. 43 14. 38 15. 95 14. 85 17. 08 20. 07 19. 48	12. 46 14. 58 15. 96 14. 89 17. 53 20. 01 19. 39
19. 38 19. 21 19. 08 19. 89 21. 80 22. 29 21. 38 22. 55 22. 91	$19.34 \\19.16 \\19.06 \\20.11 \\21.96 \\22.18 \\21.51 \\22.64 \\22.95 \\$	$19. 28 \\19. 05 \\19. 02 \\20. 12 \\22. 15 \\22. 14 \\21. 53 \\22. 57 \\23. 01$	$\begin{array}{c} 19.\ 32\\ 19.\ 08\\ 19.\ 09\\ 20.\ 37\\ 22.\ 20\\ 22.\ 06\\ 21.\ 69\\ 22.\ 73\\ 23.\ 22\\ \end{array}$	$19. 43 \\19. 06 \\19. 22 \\20. 58 \\22. 34 \\21. 92 \\21. 86 \\22. 88 \\23. 22$	19. 47 19. 24 19. 30 20. 86 22. 42 21. 72 21. 91 22. 97 23. 16	19. 45 19. 17 19. 34 21. 03 22. 40 21. 47 21. 89 22. 85 23. 17	19.58 19.01 19.54 21.26 22.46 21.36 21.97 22.90	19. 55 18. 92 19. 49 21. 39 22. 35 21. 25 22. 08 22. 87	19. 42 19. 08 19. 69 21. 52 22. 35 21. 40 22. 27 22. 95	19. 35 19. 14 19. 69 21. 56 22. 36 21. 42 22. 28 22. 93	19.19 18.90 19.70 21.64 22.37 21.36 22.34 22.88
				Total in	ventori	es (1947	dollars)		_		
13.93 15.58 15.75 16.17 18.55 18.66	14.00 15.68 15.74 16.06 18.66	14, 23 15, 66 15, 71 16, 25 18, 69	13. 60 14. 46 15. 65 15. 70 16. 73 18. 67	$\begin{array}{c} 13 & 75 \\ 14 & 77 \\ 15 & 62 \\ 15 & 66 \\ 17 & 04 \\ 18 & 52 \end{array}$	$13.91 \\ 14.99 \\ 15.64 \\ 15.65 \\ 17.34 \\ 18.48$	14.01 15.19 15.64 15.48 17.78 18.54	14. 22 15. 26 15. 61 15. 39 18. 07 18. 50	14. 19 15. 45 15. 60 15. 63 18. 23 18. 51	14.02 15.49 15.58 15.78 18.41 18.52	14.11 15.46 15.56 16.10 18.39 18.60	14.11 15.50 15.65 16.21 18.33 18.63
18. 67 18. 38 18. 26 19. 08	18.67 18.38 18.22 19.22	18.59 18.17 18.19 19.17	18. 62 18. 20 18. 24 19. 33	18. 70 18. 18 18. 40 19. 46	18.76 18.36 18.46 19.65	18. 71 18. 29 18. 49	18.80 18.14 18.71	18.71 18.04 18.68	18.56 18.31 18.90	18.49 18.30 18.89	18.36 18.11 18.92
	•	· `•	P	urchase	d mate	rials (bo	ok valu	e)	<u> </u>		
4.85 6.59 7.05 7.06 6.43 8.74 9.03 8.43	4.98 6.65 7.10 7.00 6.42 8.99 9.02	5.09 6.74 7.21 6.84 6.42 9.33 8.93	5. 20 6. 85 7. 27 6. 72 6. 40 9. 66 8. 90	5. 31 6. 92 7. 38 6. 66 6. 42 9. 64 8. 89	5. 42 6. 88 7. 52 6. 56 6. 44 9. 65 8. 82	5. 68 6. 74 7. 58 6. 51 6. 58 9. 62 8. 78	5.89 6.74 7.54 6.48 6.84 9.57 8.67	5. 92 6. 80 7. 52 6. 47 7. 27 9. 27 8. 60	6. 13 6. 78 7. 40 6. 42 7. 62 9. 30 8. 49	6.26 7.03 7.21 6.44 8.01 9.12 8.52	6. 44 7. 09 7. 12 6. 45 8. 38 8. 99 8. 51
8.47 8.16 7.86 8.18 8.60 8.80 8.55 9.02 8.74	8.42 8.00 7.79 8.20 8.70 8.81 8.63 9.08 8.68	8.45 7.88 7.76 8.20 8.74 8.81 8.63 9.06 8.78	8.40 7.93 7.79 8.16 8.66 8.77 8.69 9.09 8.88	8.46 7.95 7.95 8.31 8.85 8.80 8.85 9.13 8.97	8.40 8.00 8.32 8.96 8.78 9.02 9.13 8.97	8.42 8.06 7.99 8:32 8.97 8.75 9.00 9.05 8.96	8.35 7.95 8.07 8.42 9.02 8.71 8.95 9.02	8.35 7.84 8.02 8.44 8.92 8.61 8.92 8.89	8. 23 7. 88 8. 15 8. 58 8. 95 8. 65 8. 95 8. 85	8. 22 7. 96 8. 17 8. 57 8. 92 8. 59 8. 93 8. 78	8. 14 7. 89 8. 12 8. 53 8. 79 8. 55 8. 95 8. 75
	<u> </u>	1	I F	urchase	d mate	rials (19	47 dolla	! rs)	<u> </u>	<u> </u>	<u>.                                    </u>
6.64 7.07 6.97 7.78 8.31 8.31	6. 67 7. 04 6. 94 7. 77 8. 33	6. 83 6. 94 6. 90 7. 91 8. 32	6.84 6.95 6.89 6.84 8.13 8.36	6. 89 7. 07 6. 88 6. 82 8. 15 8. 44	6. 93 7. 18 6. 87 6. 81 8. 21 8. 42	6. 86 7. 25 6. 90 6. 86 8. 32 8. 41	6. 90 7. 27 6. 93 6. 99 8. 45 8. 32	6. 93 7. 33 6. 96 7. 23 8. 37 8. 30	6.80 7.29 6.92 7.39 8.52 8.24	6.90 7.15 6:96 7.56 8.44 8.30	6.80 7.09 6.99 7.68 8.27 8.34
- 8.35 - 8.07 - 7.74 - 8.13	8. 33 7. 88 7. 66 8. 12	8.37 7.76 7.63 8.08	8.33 7.79 7.66 7.99	8.40 7.80 7.83 8.08	8.34 7.85 7.87 8.07	8.37 7.89 7.87	8.30 7.78 7.98	8.26 7.68 7.96	8. 14 7. 71 8. 11	8.12 7.80 8.12	8.05 7.76 8.08
	Janu- ary 9.66 12.81 14.75 15.90 15.00 18.00 20.21 19.38 19.21 19.38 19.21 19.38 19.21 19.38 22.55 19.22 91 38 22.55 16.17 18.55 16.17 18.55 16.17 18.55 16.17 18.55 16.17 18.55 16.17 18.55 16.17 18.65 19.08 19.08 19.08 19.29 21.38 22.55 10.17 18.55 16.17 18.55 18.66 19.08 19.08 19.08 19.70 18.00 20.21 19.38 22.55 10.71 18.55 16.77 18.55 18.66 19.08 19.08 19.08 19.08 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.77 19.57 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.70 19.	Janu- ary         Feb- ruary           9.66         9.91           12.81         13.05           14.75         14.87           15.90         15.85           15.00         14.97           18.00         18.23           20.21         20.19           19.34            19.38         19.34           19.21         19.16           19.03         20.21           22.29         22.18           21.38         21.51           22.91         22.95	Janu- ary         Feb- ruary         March           9.66         9.91         9.98           12.81         13.05         13.28           14.75         14.87         15.09           15.00         14.97         15.06           15.00         14.97         15.06           18.00         18.23         18.77           20.21         20.19         20.18           19.34         19.06         19.02           19.89         20.11         20.12           19.80         21.82         21.82           22.91         22.18         22.15           22.91         22.12         12.23           22.291         22.18         22.15           22.95         23.01           13.33         14.00         14.23           15.55         15.68         15.66           15.75         15.74         15.71           14.77         16.66         18.69           18.667         18.22         18.22           18.67         18.29         19.17           19.08         19.22         19.17           19.08         19.22         19.17           19.08	Janu- ary         Feb- ruary         March         April           3ry         ruary         March         April           -         -         -         -         -           9.66         9.91         9.98         10.09         13.23           12.81         13.05         13.28         13.63           14.75         14.87         15.09         15.10           18.00         18.23         18.77         19.47           19.21         19.16         19.23         18.77         19.47           19.38         19.34         19.22         19.20         19.06         19.02           19.38         19.34         19.22         19.05         19.08           19.08         19.00         19.02         19.09         19.09           19.89         20.11         20.13         21.20         22.21         22.21         22.22         22.20         22.21         22.21         22.20         22.21         22.21         22.21         22.27         33.01         23.22           22.91         22.95         23.01         23.22         23.01         23.22         23.01         23.22         23.01         23.22         23.22	Janu- ary         Feb- ruary         March         April         May           Total inve           Total inve           9.66         9.91         9.98         10.09         10.25           12.81         13.05         13.28         13.63         13.78           14.75         14.87         15.09         15.46         15.10         15.46           15.00         14.97         15.06         15.10         15.17         19.69           20.12         20.19         20.18         20.01         19.73         19.34           19.34            7.73         22.83         21.92         22.23         22.23         22.22         22.23         22.22         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         22.23         <	Janu- aryFeb- ruaryMarch MarchAprilMay AprilJuneTotal inventories9.669.919.9810.0910.2510.3012.8113.0513.2813.6313.7813.7814.7514.8715.0915.1915.4615.5815.0014.9715.0615.1015.1715.1818.0018.2318.7719.4719.6919.9820.2120.1920.1820.0119.7319.6219.3419.3819.3419.2219.0819.0619.2219.3019.0619.0219.0819.0621.9222.3422.4222.2922.2921.8222.1422.0621.9222.3422.7722.8822.8722.9122.1822.1422.0621.9223.2223.161120.3715.6615.6815.6615.6515.6215.6215.6215.6415.6516.7715.7415.7415.7615.6515.6215.6215.6415.7574.15.7115.7015.6615.6515.6215.6515.6216.7317.0417.3418.3618.3718.3618.7718.7618.86718.6718.6718.6718.6718.6718.6218.7018.7618.86718.6618.995.205.315.426.6	Janu- ary         Feb- ruary         March         April         May         June         July           Total inventories (book v.           9.66         9.91         9.98         10.09         10.25         10.30         10.96           12.81         13.05         13.281         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.78         13.77         13.91         14.01         <	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	

¹ Data revised January 1953.

	Janu- ary	Feb- ruary	March	April	May	June	July	Au- gust	Sep- tem- ber	Octo- ber	No- vem- ber	De- cem- ber
					Goods-	in-pree	ess (boo	k value)	)			<u> </u>
1946 1947 1948 1949 1950 1951 1952 1953	1.67 1.82 2.19 2.24 2.19 2.49 2.76 2.78	1.63 1.84 2.24 2.22 2.18 2.52 2.75	1.63 1.92 2.23 2.16 2.22 2.59 2.75	1. 63 1. 96 2. 23 2. 12 2. 23 2. 67 2. 68	1.72 1.96 2.26 2.12 2.22 2.70 2.63	1.75 2.00 2.24 2.12 2.28 2.73 2.57	1.78 2.02 2.24 2.15 2.35 2.64 2.65	1.75 2.10 2.30 2.14 2.39 2.58 2.72	1.81 2.09 2.28 2.14 2.41 2.62 2.76	1.82 2.10 2.31 2.12 2.46 2.61 2.83	1.77 2.16 2.32 2.10 2.50 2.68 2.79	1, 81 2, 21 2, 25 2, 11 2, 55 2, 72 2, 73
1953 1 1954 1955 1956 1957 1958 1959 1960 1961	2.75 2.61 2.68 2.78 3.00 2.99 3.00 3.05 3.07	2.83 2.61 2.68 2.80 2.97 2.97 3.01 3.05 3.04	2.77 2.59 2.67 2.76 2.97 2.94 3.00 3.03 3.05	2. 84 2. 67 2. 68 2. 81 2. 95 2. 90 3. 03 3. 10 3. 12	2.86 2.65 2.71 2.83 2.95 2.92 3.02 3.13 3.20	2.90 2.62 2.74 2.91 2.98 2.94 3.04 3.14 3.25	2.84 2.56 2.75 2.93 2.94 2.91 3.11 3.17 3.29	2.82 2.57 2.85 2.93 2.97 2.96 3.08 3.17	2.74 2.54 2.80 2.90 2.94 2.92 3.06 3.09	2, 71 2, 60 2, 80 2, 95 2, 96 2, 95 3, 09 3, 10	2.66 2.68 2.83 2.94 3.00 2.96 3.07 3.14	2. 65 2. 60 2. 76 2. 96 3. 06 3. 00 3. 03 3. 08
					Gcods-	in-proce	ss (1947	dollars)	•	·		<u></u>
1947 1948 1949 1950 1951 1952 1953	2.09 2.20 2.29 2.26 2.54 2.68	2. 13 2. 19 2. 29 2. 23 2. 54	2. 12 2. 16 2. 31 2. 26 2. 55	1. 97 2. 14 2. 14 2. 32 2. 32 2. 50	1.96 2.16 2.15 2.30 2.35 2.47	2.01 2.14 2.17 2.35 2.39 2.42	2.05 2.15 2.21 2.39 2.34 2.50	2. 13 2. 20 2. 21 2. 39 2. 31 2. 57	2. 12 2. 19 2. 23 2. 36 2. 38 2. 61	2. 10 2. 23 2. 20 2. 37 2. 39 2. 69	2. 13 2. 25 2. 19 2. 36 2. 46 2. 66	2. 14 2. 19 2. 21 2. 36 2. 50 2. 62
1953 ¹ 1954 1955 1956	2. 65 2. 50 2. 57 2. 66	2. 73 2. 49 2. 56 2. 68	2. 67 2. 48 2. 56 2. 63	2. 74 2. 55 2. 57 2. 67	2.75 2.52 2.59 2.68	2.79 2.50 2.62 2.75	2.73 2.44 2.63	2. 71 2. 46 2. 73	2.62 2.42 2.68	2. 59 2. 48 2. 69	2. 54 2. 56 2. 72	2. 54 2. 50 2. 66
		·			Finish	ed good	ı ls (book	value)	1	I	<u> </u>	l
1946 1947 1948 1949 1950 1951 1952 1953	$\begin{array}{r} 3.18\\ 4.44\\ 5.38\\ 6.54\\ 6.37\\ 6.68\\ 8.39\\ 8.15 \end{array}$	3. 32 4. 56 5. 42 6. 62 6. 40 6. 71 8. 45	3. 29 4. 62 5. 52 6. 67 6. 40 6. 78 8. 47	3. 25 4. 76 5. 63 6. 67 6. 44 7. 04 8. 44	3. 26 4. 89 5. 78 6. 60 6. 45 7. 34 8. 22	3. 21 4. 95 5. 92 6. 57 6. 43 7. 55 8. 22	3. 43 5. 06 6. 08 6. 47 6. 23 7. 94 8. 21	3. 65 5. 13 6. 09 6. 40 6. 11 8. 12 8. 20	$\begin{array}{c} 3.87\\ 5.11\\ 6.27\\ 6.34\\ 6.26\\ 8.27\\ 8.19\end{array}$	4. 24 5. 13 6. 30 6. 38 6. 34 8. 27 8. 16	4. 34 5. 14 6. 37 6. 32 6. 56 8. 25 8. 18	4. 26 5. 28 6. 49 6. 34 6. 64 8. 29 8. 18
1953 1 1954 1955 1956 1957 1958 1959 1960 1961	8. 15 8. 44 8. 53 10. 21 10. 50 9. 83 10. 48 11. 10	8.09 8.55 8.58 9.10 10.29 10.40 9.87 10.51 11.23	8.06 8.58 8.59 9.16 10.44 10.40 9.90 10.48 11.18	8.07 8.48 9.40 10.59 10.39 9.97 10.54 11.22	8. 10 8. 46 8. 57 9. 44 10. 54 10. 19 10. 00 10. 62 11. 05	8. 18 8. 62 8. 57 9. 63 10. 47 10. 00 9. 86 10. 60 10. 94	8. 19 8. 56 8. 60 9. 78 10. 48 9. 80 9. 78 10. 63 10. 93	8. 40 8. 49 8. 63 9. 91 10. 46 9. 69 9. 94 10. 71	8.46 8.53 8.68 10.06 10.49 9.72 10.10 10.89	8.48 8.61 8.74 9.99 10.44 9.80 10.22 11.00	8. 47 8. 50 8. 69 10. 05 10. 44 9. 87 10. 28 11. 01	8.40 8.41 8.82 10.15 10.52 9.81 10.36 11.05
					Finishe	ed good	s (1947 d	lollars>			I	
1947 1948 1949 1950 1951 1952 1953	5. 20 6. 31 6. 48 6. 13 7. 70 7. 67	5. 20 6. 45 6. 51 6. 06 7. 79	5. 28 6. 56 6. 50 6. 07 7. 82	4.79 5.38 6.62 6.54 6.28 7.81	4. 90 5. 54 6. 59 6. 54 6. 54 7. 62	4.97 5.66 6.60 6.49 6.74 7.64	5. 10 5. 80 6. 53 6. 22 7. 11 7. 62	5. 19 5. 78 6. 47 6. 00 7. 31 7. 61	5. 14 5. 93 6. 42 6. 03 7. 48 7. 60	5. 12 5. 97 6. 45 6. 02 7. 50 7. 59	5. 07 6. 06 6. 40 6. 17 7. 50 7. 63	5. 17 6. 21 6. 45 6. 18 7. 56 7. 67
1953 1 1954 1955 1956	7.67 7.80 7.96 8.28	7. 61 7. 90 7. 99 8. 42	7. 56 7. 94 8. 00 8. 46	7.54 7.86 8.02 8.67	7.55 7.85 7.98 8.69	7.63 8.01 7.96 8.84	7.62 7.96 7.99	7.80 7.90 8.01	7.82 7.94 8.05	7.83 8.12 8.10	7.83 7.94 8.05	7.77 7.86 8.19
					•							

## TABLE B-3.—Nondurable-goods industries, total inventories—Continued

[Billions of dollars, seasonally adjusted]

¹ Data revised, January 1953.

Source: Department of Commerce; deflations by the author.

### TABLE B-4.—Nonfarm inventory investment

[Billions of dollars, seasonally adjusted at annual rates]

	Total nonfarm							Total manufacturing								
	Current dollars 1954 dollars								Current	t dollars			1954 d	ollars		
ľ	I	п	ш	IV	I	ц	111	IV	I	п	ш	IV	I	п	m	· IV
1947 1948 1949 1950 1951 1953 1955 1956 1956 1958 1959 1959 1960 1960	+1.53 + 2.23 + 2.23 + 2.23 + 2.86 + 2.88 + 2.88 + 2.88 + 2.88 + 2.88 + 2.66 + 2.59 + 6.98 + 2.68 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 + 2.59 +	$\begin{array}{r} +1.5 \\ +2.9 \\ -4.1 \\ +4.2 \\ +14.0 \\ -3.3 \\ +4.0 \\ -3.2 \\ +5.7 \\ +5.2 \\ +5.0 \\ +11.6 \\ +5.1 \end{array}$	$\begin{array}{r} -0.3 \\ +3.9 \\ +3.8 \\ +9.1 \\ +3.4 \\ +1.5 \\ -2.5 \\ +4.4 \\ +1.5 \\ -2.5 \\ +.7 \\ +2.0 \\ \end{array}$	$\begin{array}{r} +2.4 \\ +2.8 \\ -4.7 \\ +13.8 \\ +4.7 \\ -4.3 \\ +4.7 \\ -4.3 \\ +6.7 \\ +6.7 \\ +2.6 \\ +2.6 \\ +2.5 \\ -2.2 \\ \end{array}$	+1.4 + +2.42 + +2.42 + +4.32 + +4.32 + +4.32 + +4.32 + +4.32 + +4.32 + +4.32 + +4.32 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33 + +4.33	$\begin{array}{r} +1.8\\ +3.1\\ -4.6\\ +4.8\\ +13.7\\ -3.3\\ +4.1\\ -3.4\\ +5.8\\ +5.8\\ +5.0\\ +1.7\\ -4.3\\ +10.3\\ +4.7\end{array}$	$\begin{array}{r} -0.3 \\ +4.1 \\8 \\ +9.2 \\ +3.3 \\ +1.5 \\ -2.7 \\ +5.4 \\ +1.3 \\ -2.1 \\ +1.0 \\ +2.0 \end{array}$	+2.99+-5.45+-14.594.31+-4.3.94.31+-6.692.41+-1.3	1.0 2 5 7 0 5 8 8 0 6 6 8 8 7 1 + + + + 6 2 5 8 8 0 6 6 8 8 6 3 + 5 5 6 6 8 8 6 3 + + + + + + + + + + + + + + + + + +	-0.12+2.79+1114+2.89+11.44+2.89+2.44+5.38+2.55+2.00+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.50+2.	$\begin{array}{c} 9.1 \\ +2.57 \\ +2.57 \\ +1.5332 \\ +3.532 \\ +3.532 \\ +3.532 \\ +2.4 \\ +2.4 \\ +2.532 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4 \\ +2.4$	$\begin{array}{c} 0 + 7 \\ -26.55 \\ +42.02 \\ +3.84 \\ +3.80 \\ +3.80 \\ +3.80 \\ +3.11 \\ +3.5 \\ +3.11 \\ +4.05 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5 \\ -3.5$	95267 +-1.67067 ++4.623798 ++4.44 +-2.92 ++1.292 ++1.292 ++1.292 ++1.292 ++1.292 ++1.292 ++1.292 ++1.292 ++1.292 +-1.292 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.202 +-1.2	$\begin{array}{c} 0 \\ +2.3 \\ -3.0 \\ +1.2 \\ +10.8 \\ -1.3 \\ +3.0 \\ -4.1 \\ +2.5 \\ +4.6 \\ +.4 \\ -4.5 \\ +5.0 \\ +1.7 \end{array}$	$\begin{array}{r} & 1.2 \\ +1.2 \\ +2.3 \\ +3.9 \\ +1.5 \\ +1.5 \\ +1.5 \\ +3.3 \\ +2.5 \\ +3.3 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +1.5 \\ +$	$\begin{array}{c} -0.1 \\ +.8 \\ -2.68 \\ +4.6 \\ +2.1 \\ -2.1 \\ +.6 \\ +2.2 \\ -2.6 \\ +1.2 \\ -2.2 \\ +3.2 \\ -2.6 \\ +1.2 \\ -2.6 \\ +1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -1.2 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.6 \\ -2.$

	. Durable-goods manufacturing								Nondurable-goods manufacturing							
		Current dollars								Curren	t dollars			1954 0	lollars	
	I	п	III	IV	I	ш	m	IV	I	п	ш	IV	I	II	ш	IV .
1947 1948 1949 1950 1951 1952 1954 1954 1955 1956 1959 1959 1960 1961	-0.52+.71+.3.3+44.1+.3.22+.4.0+.3.9+.4.0+.3.8+.4.29+.4.29+.4.2+.4.29+.4.29+.4.29+.4.29+.4.29+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4.20+.4	$+0.6 \\ 0 \\ -2.1 \\ +1.6 \\ +2.2 \\ +2.2 \\ +4.0 \\ +1.6 \\ +3.3 \\ +1.9 \\ -4.2 \\ +4.8 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.3 \\ +1.$	$\begin{array}{c} +1.2 \\ +2.3 \\ -2.3 \\ +1.1 \\ +7.9 \\ +.6 \\ +1.7 \\ -2.5 \\ +1.9 \\ 0 \\ -1.5 \\7 \\ +.1 \end{array}$	$\begin{array}{r} -0.7 \\ +.1 \\ -1.8 \\ +3.6 \\ +3.3 \\ +1.8 \\ -1.9 \\ +.3 \\ +2.5 \\ +2.1 \\ -2.9 \\ 0 \\ +2.0 \\ -3.8 \end{array}$	$\begin{array}{r} -0.7 \\ -1.6 \\ +.7 \\ -3.4 \\ +4.3 \\ +2.4 \\ +3.8 \\ +3.8 \\ +1.0 \\ -3.3 \\ +3.8 \\ +1.0 \\ -3.4 \\ -1.4 \end{array}$	$\begin{array}{r} +0.7\\ 0\\ -2.4\\ +1.6\\ +7.1\\2\\ +2.6\\ +3.1\\ +1.6\\ +3.1\\ +.8\\ -3.5\\ +4.0\\ +1.1\end{array}$	$\begin{array}{c} +1.5 \\ +1.4 \\ -2.82 \\ +3.2 \\ +1.67 \\ +1.55 \\ +1.8 \\ +0 \\ -1.3 \\61 \\61 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\ +.1 \\$	$-0.9 \\ -2.1 \\ -2.1 \\ +3.9 \\ +3.5 \\ +1.9 \\ +2.3 \\ +2.3 \\ +2.6 \\ 0 \\ +1.7 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.0 \\ -3.$	1.525 +1.525 +1.+.+.582 +1.+.+.+.50 +1.+.50 +1.+.50 +1.+.30 +1.34 +1.34	$\begin{array}{r} -0.7 \\ +2.3 \\6 \\4 \\ +4.3 \\ -1.2 \\ +.4 \\ 0 \\ +.9 \\ +1.5 \\4 \\ -1.1 \\ +1.0 \\ +.6 \end{array}$	$\begin{array}{c} -0.3 \\ +1.8 \\ -3.3 \\ +3.3 \\ +3.3 \\ +1.2 \\ -7.7 \\ +1.8 \\ -5.5 \\ +1.1 \\ +3.3 \\ +1.5 \\ -5.5 \\ +1.1 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ -1.5 \\ +1.3 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1.5 \\ -1$	$\begin{array}{c} +0.7\\ +.6\\5\\ +3.0\\ +1.1\\ +.2\\3\\ +1.3\\ +.3\\ +1.3\\ +.2\\ 0\\ 0\\ +1.2\\ +1.6\\ +.4\end{array}$	$\begin{array}{r} +1.6\\ +1.1\\ +.72\\ +1.82\\ +.6\\ +1.8\\ +.6\\ +1.4\\ +1.3\\ +1.5\\ +1.3\end{array}$	$\begin{array}{c} -0.7 \\ +2.3 \\5 \\ +3.7 \\ -1.1 \\ +.4 \\ 0 \\ +1.0 \\ +1.0 \\ +1.0 \\ +1.0 \\ +1.0 \\ +.6 \end{array}$	-0.3824 -14022773955118 -1955118	+0. +-:-2:+1:+ +-:-+:+:+:+:- +:+:+:+:- +:+:+:+:+:+:- +:+:+:+:

Source: Department of Commerce, National Income Division.

505 14.55 360 505

#### TABLE B-5.—Inventory investment: Summary

[Millions of dollars, seasonally adjusted]

#### A. TOTAL INVESTMENT

#### TOTAL MANUFACTURING

		Book	value		1947 dollars				
	I	II .	ш	IV	I	II	III	IV	
1946	$\begin{array}{r} +880\\ +1,877\\ +608\\ +161\\ +328\\ +2,556\\ +1,117\\ +232\\ +450\\ -900\\ +230\\ +1,620\\ +1,039\\ -1,511\\ +1,275\\ +1,910\\ -430\end{array}$	$\begin{array}{c} +1,097\\ +1,310\\ +790\\ -1,221\\ +559\\ +3,070\\ -607\\ +1,104\\ +519\\ +514\\ +519\\ +514\\ +1,684\\ +760\\ +40\\ \end{array}$	$\begin{array}{r} +2,319\\ +543\\ +992\\ -1,295\\ +1,271\\ +1,971\\ +1,971\\ +1,971\\ +1,971\\ +1,137\\ +1,217\\ -9999\\ +1,133\\ +1,217\\ +313\\ -950\\ -246\\ -390\end{array}$	$\begin{array}{r} +1,771\\ +687\\ +429\\ -478\\ +3,296\\ +904\\ +645\\ \hline \\ -616\\ +334\\ +1,502\\ +1,453\\ -646\\ -117\\ +538\\ -970\\ \hline \end{array}$		$\begin{array}{c} +813\\ -409\\ +246\\ +2,467\\ -204\\ \end{array}$	$\begin{array}{c} +409\\ +400\\ -548\\ +88\\ +2,439\\ +100\\ -929\\ +408\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\$	$\begin{array}{c} -219 \\ +61 \\ -362 \\ +1,366 \\ +784 \\ +456 \\ -571 \\ +228 \\ +708 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -571 \\ -$	
							-		

1946.         1947.         1948.         1949.         1950.         1951.         1953.         1953.         1953.         1954.         1955.         1956.         1957.         1958.         1959.         1959.         1959.         1960.	$\begin{array}{r} +520\\ +1,058\\ +90\\ +424\\ +156\\ +1,316\\ +951\\ +454\\ +561\\ -759\\ +113\\ +1,204\\ +525\\ -1,284\\ +1,110\\ +1,690\end{array}$	+777 +813 +306 -813 +438 +1,981 -55 +907 -1,079 +229 +892 +253 -1,336 +1,302 +460	$\begin{array}{r} +1,087\\ +323\\ +654\\ -879\\ +557\\ +1,767\\ -51\\ \end{array}$	$\begin{array}{r} +846\\ +107\\ +389\\ -495\\ +1,655\\ +962\\ +761\\ \end{array}$	$\begin{array}{c}363\\ +65\\ -128\\ +446\\ +604\\ +403\\ +368\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +24\\ +533\\ -507\\ +26\\ +53\\ +56\\ +56\\ +56\\ +56\\ +56\\ +56\\ +56\\ +56$	$\begin{array}{c} +57\\ -383\\ +302\\ +1,873\\ +7\\ +7\\ +473\\ -799\\ +26\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ -28\\ +392\\ +392\\ -28\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +392\\ +39$	$\begin{array}{r} +124 \\ +30 \\ -516 \\ +113 \\ +1,548 \\ +62 \\ \end{array}$	$\begin{array}{c} -138 \\ +99 \\ -410 \\ +778 \\ +685 \\ +338 \\ -215 \\ +159 \\ +466 \\$
1960 1961	+1,690 -560	+460 -130						

#### NONDURABLE-GOODS INDUSTRIES

				· · · ·				4
1946 1947 1948 1949 1950 1951 1952 1952	+360 +819 +518 -263 +172 +1,240 +166 -223	+320 +497 +484 -408 +121 +1,089 -552	+1,232 +220 +338 -416 +714 +204 -120	+925 +580 +40 +17 +1,641 -58 -116	+118 +164 +55 +32 +356 +14	+756 -26 -56 +1,094 -211	+285 +460 -32 -25 +891 +38	-81 +52 +48 +588 +99 +118
1953         1           1954	$-111 \\ -141 \\ +117 \\ +416 \\ +514 \\ -227 \\ +165 \\ +230 \\ +130$	+197 +198 +285 +749 +266 -427 +382 +300 +170	$^{+81}_{-328}$ $^{+184}_{+526}$ $^{-69}_{-470}$ $^{+164}_{+164}$	$\begin{array}{r} -367 \\ -15 \\ +213 \\ +244 \\ +26 \\ +116 \\ +265 \\ +10 \end{array}$	-16 -186 +74 +248 	+165 +188 +269 +474	-44 -313 +227	-356 +69 +242

¹ Data revised January 1953.

DURABLE-GOODS INDUSTRIES

## TABLE B-5.-Inventory investment: Summary-Continued

#### [Millions of dollars, seasonally adjusted]

#### B. PURCHASED-MATERIALS INVESTMENT

#### TOTAL MANUFACTURING

		Book	value		1947 dollars			
	I	п	III	IV	I	п	ш	IV
1946	+425 +613 +95 -308 -43 +1, 231 +49 -40	+542 +407 +553 -749 +315 +841 -469	$^{+968}_{-78}$ $^{+135}_{-361}$ $^{+1,344}_{+14}$ $^{+232}_{-232}$	+969 +329 -286 -277 +1,785 -142 +97	-122 -166 -138 +207 +150 +40	+442 -401 +81 +667 -156	-122 + 46 - 80 + 681 + 418 - 198	$ \begin{array}{r} -204 \\ -296 \\ -189 \\ +819 \\ +28 \\ +138 \\ \end{array} $
1953 1	-60 -542 -19 +516 +258 -161 +670 +570 -100	$^{+356}_{+161}_{+252}_{+396}_{-120}_{-578}_{-578}_{+1, 140}_{+10}_{+130}$	+72 -497 +558 +232 +93 -212 -620 -510		-70 -466 -80 +195	+195 +116 +205 +113	-121 -464 +297	489 193 +175
	·	DUR	ABLE-GOO	DS INDUST	RIES	1	·	<u> </u>
1946	$\begin{array}{r} +245 \\ +320 \\ -20 \\ +27 \\ +26 \\ +309 \\ +150 \\ +20 \\ -20 \\ -278 \\ +115 \\ +434 \\ +494 \\ -184 \\ +500 \\ +450 \\ -140 \\ \end{array}$	$\begin{array}{r} +210 \\ +249 \\ +229 \\ -525 \\ +284 \\ +538 \\ -364 \\ \hline \\ +404 \\ +35 \\ +97 \\ +276 \\ -343 \\ -548 \\ +750 \\ -60 \\ -310 \\ \end{array}$	$\begin{array}{r} +465\\ -4\\ +119\\ -301\\ +462\\ +307\\ -65\\ \hline \\ +124\\ -338\\ +541\\ +118\\ +143\\ -40\\ -510\\ -270\\ \hline \end{array}$	$\begin{array}{r} +454 \\ +58 \\ +134 \\ -236 \\ +173 \\ +206 \\ \hline \\ -380 \\ -344 \\ +265 \\ +418 \\ -205 \\ -20 \\ -370 \\ \hline \end{array}$	$\begin{array}{c}153\\ -16\\ -47\\ +35\\ +105\\ -12\\ -40\\ -173\\ +47\\ +196\\$	+87 -328 +171 +367 -254 +222 +24 -36 +123 		-773 -633 -223 +372 +130 +96 -276 -274 +53 
		Nond	URABLE-GO	oods Indu	STRIES			
1946	$^{+184}_{+305}_{+125}_{-278}_{-31}_{+947}_{-59}_{-39}$	$ \begin{array}{r} +331 \\ +137 \\ +309 \\ -273 \\ +27 \\ +320 \\ -111 \end{array} $	+498 -84 +3 -92 +827 -372 -219	+522 +291 -409 -23 +1,107 -282 -92	$ \begin{array}{r} +31 \\ -150 \\ -91 \\ +232 \\ +45 \\ +52 \\ \end{array} $	+355 -73 -90 +300 +98	$ \begin{array}{r} +1 \\ +143 \\ +86 \\ +424 \\ +162 \\ -115 \end{array} $	$-131 \\ -233 \\ +34 \\ +447 \\ -102 \\ +42$
1953 1 1954 1955 1966 1967 1958 1959	-122 -264 -134 +82 +209 +23 +80	-48 + 126 + 243 + 120 + 223 - 30 + 390		-205 +46 +103 +97 -128 -60 +30	-30 -293 -127 -1	-27 +92 +241 -10	-76 -173 +82	-213 +81 +122

 $-100 \\ -240$ 

+30

140

. . . . .

. . . . .

-----

+60 + 190

-30

¹ Data revised January 1953.

1960.....

1961.....

# TABLE B-5.—Inventory investment: Summary—Continued [Millions of dollars, seasonally adjusted]

. .

#### C. GOODS-IN-PROCESS INVESTMENT

#### TOTAL MANUFACTURING

		Book	value		1947 dollars			
	I	п	ш	IV	I	п	щ	IV
1946	+234 +436 -150 -117 +149 +752 +604 +388	+510 +156 +85 -267 +345 +764 +186	+430 +285 +257 -298 +527 +543 +469	+252 +86 +32 -146 +691 +741 +261	-285 -112 +85 +254 +477 +294	$-14 \\ -106 \\ +218 \\ +548 \\ +152$	+187 +68 -140 +225 +518 +388	-140 -96 -118 +258 +613 +169
1953 '         1954         1955         1956         1957         1958         1959         1960         1961	$+486 \\ -508 \\ +88 \\ +521 \\ +276 \\ -959 \\ +180 \\ +580 \\ -190$	+358 - 583 + 187 + 514 + 306 - 480 + 430 + 260 + 330	$\begin{array}{r} -141 \\ -327 \\ +442 \\ +344 \\ +77 \\ -116 \\ +80 \\ -470 \end{array}$	$\begin{array}{r} -207 \\ +481 \\ +690 \\ +498 \\ -600 \\ +80 \\ +100 \\ -340 \end{array}$	+362 -370 +35 +216	+204 417 +88 +270	-309 -312 +151	-193 +323 +289 
	·	Dur	ABLE-GOO	os Indust	RIES			
1946           1947           1948           1949           1950           1951           1952           1953           1954           1955           1956           1956           1957           1958           1960           1961	$\begin{array}{r} +100\\ +325\\ -157\\ -36\\ +360\\ +705\\ +564\\ +333\\ +431\\ -449\\ +220\\ +528\\ +264\\ +280\\ +200\\ +580\\ -160\end{array}$	$\begin{array}{r} +399\\ +85\\ +72\\ -225\\ +276\\ +614\\ +355\\ \end{array}$	$\begin{array}{r} +360 \\ +177 \\ +212 \\ -317 \\ +400 \\ +662 \\ +291 \\ \end{array}$	$\begin{array}{c} +254\\ -37\\ +72\\ -112\\ +566\\ +654\\ +285\\ \hline \\ -113\\ +416\\ +720\\ +411\\ -724\\ -10\\ +130\\ -330\\ \hline \end{array}$	$\begin{array}{c}263\\80\\15\\ +348\\ +418\\ +239\\ +239\\26\\ +239\\26\\ +239\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\$		+75 +20 -204 +530 +198 	$\begin{array}{c} -165 \\ -97 \\ -97 \\ -99 \\ +262 \\ +167 \\ -104 \\ +251 \\ +315 \\ \\ \\ \\ \\$
<u></u>		Nond	URABLE-GO	ods Indu	STRIES			
1946 1947 1948 1949 1950 1951 1952 1953	$\begin{array}{c} +128 \\ +109 \\ +18 \\ -82 \\ +105 \\ +40 \\ +34 \\ +46 \end{array}$	$\begin{array}{c c} +120 \\ +74 \\ +13 \\ -47 \\ +62 \\ +143 \\ -181 \end{array}$	$\begin{array}{c} +62 \\ +98 \\ +41 \\ +27 \\ +133 \\ -109 \\ +190 \end{array}$	+5 +119 -38 -32 +136 +95 -34	$ \begin{array}{c} -22 \\ -32 \\ 0 \\ -94 \\ +59 \\ +55. \end{array} $	+22 +6 +39 +125 -130	+112 +48 +64 +10 -12 +190	+25 +1 -19 -4 +118 +2
1953 1           1954           1955           1956           1957           1958           1959           1959           1960           1961	$ \begin{array}{c} +55 \\ -59 \\ +68 \\ -7 \\ +12 \\ -129 \\ 0 \\ 0 \\ -30 \\ \end{array} $	$\begin{array}{ c c c c } +& 129 \\ & +& 25 \\ & +& 66 \\ +& 153 \\ & +& 12 \\ & +& 9 \\ & +& 40 \\ & +& 110 \\ & +& 200 \end{array}$	$ \begin{array}{c c} -153 \\ -79 \\ +58 \\ -13 \\ -39 \\ -26 \\ +20 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -50 \\ -$	$\begin{array}{c c} -94 \\ +65 \\ -30 \\ -43 \\ +124 \\ +80 \\ -30 \\ -10 \end{array}$	$\begin{array}{ c c c } +64 \\ -61 \\ +61 \\ -23 \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\$	+120 +24 +61 +114	-162 -75 +63 	

¹ Data revised January 1953.

## TABLE B-5.-Inventory investment: Summary-Continued

#### [Millions of dollars, seasonally adjusted] D. FINISHED GOODS INVESTMENT TOTAL MANUFACTURING

		Book	t value			1947	dollars	
	I	п	ш	IV	I	п	III	IV
1946           1947           1948           1949           1949           1950           1951           1952           1953	+178 +663 +365 +421 +3 +426 +331 +184	$+18 \\ +717 \\ +480 \\ -48 \\ +57 \\ +1,590 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -259 \\ -$	$\begin{array}{r} +937 \\ +402 \\ +606 \\ -451 \\ -437 \\ +1,708 \\ -100 \end{array}$	$\begin{array}{r} +686 \\ +359 \\ +526 \\ -160 \\ +664 \\ +16 \\ +52 \end{array}$	+162 +507 -20 -43 +333 +83	$ \begin{array}{r} +385 \\ +98 \\ -53 \\ +1,252 \\ -200 \end{array} $	$ \begin{array}{c} +344 \\ +376 \\ -328 \\ -818 \\ +1,503 \\ -90 \end{array} $	$\begin{array}{c} +125 \\ +453 \\ -55 \\ +289 \\ +143 \\ +149 \end{array}$
1953 1         1954         1955         1956         1957         1958         1959         1959         1960         1961	+106 +150 +161 +583 +505 -391 +410 +770 -140	$\begin{array}{c} +390 \\ -459 \\ +75 \\ +731 \\ +333 \\ -705 \\ +120 \\ +490 \\ -160 \end{array}$	$\begin{array}{r} +763 \\ -175 \\ +133 \\ +641 \\ +143 \\ -622 \\ +290 \\ +590 \\ \end{array}$	$\begin{array}{c} +176 \\ +151 \\ +444 \\ +440 \\ +287 \\ -120 \\ +450 \\ -110 \end{array}$	+60 +143 +143 +370	+239 -310 +2 +483	+480 -153 -40	+111 +98 +244
		Dur	ABLE-GOO	DS INDUST	RIES			
1946.         1947.         1948.         1949.         1950.         1953.         1953.         1953.         1955.         1956.         1957.         1958.         1957.         1958.         1959.         1959.         1960.         1961.	$\begin{array}{r} +16\\ +299\\ +1251\\ -54\\ +261\\ +251\\ +251\\ +237\\ +115\\ +237\\ +150\\ -32\\ -222\\ +242\\ +212\\ -270\\ +310\\ +650\\ -270\end{array}$	+97 +396 +75 +16 -7 +799 -29 +274 -506 +99 +275 +302 -299 +170 +380 +80	+282 +229 +243 -221 -299 +955 -74 +477 +477 +477 +477 +4216 +123 -350 +290 +290	+294 +190 +325 +85 +85 +109  +244 +277 +304 +350 +257 -210 +190 +190 +190	+53 +161 -66 +63 +81 +176 +110 -255 +33 +98 	+6 +57 -48 +583 -21 +167 -382 +35 +113	+172 +107 -146 -359 +762 -53 +286 -88 -122	$\begin{array}{c} +100 \\ +169 \\ -88 \\ +144 \\ +60 \\ +75 \\ +165 \\ +182 \\ +98 \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\$
		NOND	I RABLE-GO	JODS INDU	STRIES			<u> </u>
1946           1947           1948           1949           1950           1952           1953	+162 +364 +239 +177 +52 +142 +182 -90	-79 +321 +397 -96 +33 +776 -255	+655 +166 +346 -232 -173 +720 -31	+392 +173 +226 +3 +378 +19 -10	+109 +346 +46 -106 +252 -93	+379 +41 -5 +669 -179	+172 +269 -182 -459 +741 37	$\begin{array}{c} +25 \\ +284 \\ +33 \\ +145 \\ +83 \\ +74 \end{array}$
1953 1         1954         1955         1956         1957         1958         1960         1969         1960	-44 +182 +183 +341 +293 -121 +90 +120 +120	+166 +47 -24 +476 +316 -406 -400 +120 -240	$+286 \\ -90 \\ +109 \\ +425 \\ +20 \\ -272 \\ +240 \\ +290$	-68 - 126 + 140 + 90 + 30 + 90 + 260 + 160	50 +168 +140 +272	+72 +72 -33 +370	+194 65 +82	-54 -84 +146

¹ Data revised, January 1953.

Source: Department of Commerce; deflations by the author.

## ANALYSIS OF BUSINESS INVENTORY MOVEMENTS IN THE POSTWAR PERIOD

PREPARED UNDER THE GENERAL DIRECTION OF

### LOUIS J. PARADISO

OFFICE OF BUSINESS ECONOMICS U.S. DEPARTMENT OF COMMERCE

Part I-Factors Influencing Manufacturers' Inventories

BY

MABEL A. SMITH Economic Analyst

PART II-Inventory Investment in Trade

BY

LAWRENCE BRIDGE Chief, Business Structure Division

AND

CLEMENT WINSTON Chief, Consumption and Trade Section Business Structure Division

145

## CONTENTS

-

·____

## ANALYSIS OF BUSINESS INVENTORY MOVEMENTS IN THE POST-WAR PERIOD

	Page
Part I. Factors influencing manufacturing inventories	149
Business inventories in the GNP	149
Manufacturers' share of inventories	150
Postwar inventory movements	150
Durable goods	152
Nondurable goods	153
Pactors affecting inventory movements	153
Dasis of analysis	154
Numerical value of inventories related to sales	154
Inventory abanges related to sales and orders	155
Results for durable, and nondurable goods monufacturing inductrics	155
Stages of fabrication	150
Comparison with manufacturers' expectations based on sample survey	109
Conclusion	162
Tables:	109
1. Changes in GNP and business inventories in postwar recession	
and recovery periods	150
2. Selected regression equations for manufacturing inventory	200
relationships	158
3. Manufacturing inventory change—comparison of estimated	
with actual	161
4. Manufacturers' inventories, sales, and new orders	162
Charts:	
1. Manufacturing industries: Inventories and sales	151
11. Manufacturing industries: Change in inventories, actual	
Compared with calculated	157
111. Manufacturing industries: Inventories by stage of fabrica-	
Part II Inventory investment in trade	160
Section A Destware developments in trade inventories	105
Composition of trade inventories	100
Postwar inventory movements	109
The 1953–57 cycle	172
Trade stocks and GNP	174
Section B. Factors affecting inventory changes	175
Measurement of inventory behavior	176
Additional relationships	178
Tables:	110
5. Retail trade: Inventories, 1947–61	166
6. Retail trade: Sales, 1947–61	167
7. Wholesale trade: Inventories and sales, 1947–61	168
8. Wholesale and retail trade: Regression equations	178
Charts:	
IV. Retail and wholesale trade: Inventories and sales	170
v. Retail stores: Inventories and sales	171
v1. Retail and wholesale trade: Actual inventories compared	
with calculated	177
147	

## ANALYSIS OF BUSINESS INVENTORY MOVEMENTS IN THE POSTWAR PERIOD

## PART I. FACTORS INFLUENCING MANUFACTURING INVENTORIES

Business inventory changes are highly volatile. Although often small in relation to total national product, inventory movements have at times in the past accounted for a large part of the short-term fluctuations in production. Indeed, over short periods fluctuations in business inventories appear to be much more pronounced than those in other principal components of private domestic investment housing and plant and equipment.

Although the volume of stocks held by business is in large part a function of the production and consumption rates, the evidence points to the fact that the relationship is not with coincident variables but rather that there is a tendency for inventory changes to lag behind the course of general business. Inventories tend to rise after the peak in business activity, and adjustment begins some time after contraction has started. Thus, stocks continue to fall after business activity has reached a trough, and tend to move up some time after the general expansion is underway. This phenomenon is due to a number of factors. Investment decisions are likely to lag changes in output or sales due to the difficulty of detecting in time the turning points in demand. Furthermore, even after a change in inventory policy is established, involuntary inventory accumulation may occur due to falling sales and to deliveries of previously ordered materials.

The purpose of the part I of this analysis is to ascertain quantitatively the effect of various relevant factors on short-term fluctuations in manufacturers' stocks by an analysis of historical relationships. This involves an attempt not only to determine lags among the variables, but also to appraise the importance of temporary and sometimes unquantifiable factors. In part II fluctuations in trade inventories are considered.

#### BUSINESS INVENTORIES IN THE GNP

The relative importance of inventory changes in the overall business picture may be seen from the accompanying table I, which compares the movement in gross national product and the shifts in the change in business inventories in each of the postwar recession and recovery periods.

149

1.20

	GNP	Business inventories 1	GNP	Business inventories 1
	Billions	Billions	Billions	Billions of 1954
Recession periods:	dollars	dollars	dollars	dollars
4th quarter 1948 to 2d quarter 1949	-9.5	-9.6	-7.0	-10.1
2d guarter 1953 to 2d guarter 1954	-9.9	-5.8	-13.7	-6.1
3d guarter 1957 to 1st guarter 1958	-15.4	-8.0	-18.0	-6.9
2d guarter 1960 to 1st guarter 1961	-5.6	-9.4	-10.2	-8.2
Recovery periods:		i		
2d guarter 1949 to 2d guarter 1953	112.4	8.4	82.9	9.1
2d guarter 1954 to 3d guarter 1957	89.4	5.2	51.5	5.2
1st quarter 1958 to 2d quarter 1960	73.5	10.9	50.4	9.8

TABLE 1.—Changes in GNP and business inventories in postwar recession and recovery periods

Computed as the change in the inventory component of GNP. Source: Office of Business Economics, U.S. Department of Commerce.

In the 1948–49 downturn, inventory liquidation was equal to the entire \$9½ billion decline in GNP. In the 1953–54 and 1957–58 recessions, reduction in stocks represented more than half of the decrease in total product, and in the most recent business contraction the size of the inventory liquidation—\$9 billion—was larger than the decline in total GNP.

If the corresponding constant dollar estimates are examined, similar conclusions may be drawn concerning the role of the physical volume of business inventories relative to changes in "real" GNP during these periods of recession.

In recovery periods, on the other hand, inventory changes have generally played a smaller role in the total production pattern. In the upturns following the first three postwar recessions, the change in inventory investment contributed at the most less than one-fifth of the total GNP rise. In the current recovery the inventory change from the first to the third quarter of 1961 accounted for one-third of the GNP rise.

#### MANUFACTURERS' SHARE OF INVENTORIES

The book value of manufacturers' stocks at the end of September amounted to \$54 billion. These inventories comprise on the average about three-fifths of total business holdings. Moreover, this proportion is generally exceeded in recessionary periods by the share of the total inventory liquidation experienced by producers. In the last two recessions, for example, about three-fourths of the total reduction in stocks took place at the manufacturing level.

#### POSTWAR INVENTORY MOVEMENTS

The growth of manufacturers' inventories in the postwar period proceeded in a series of waves, which dovetailed with the cyclical behavior of total business activity. It is clear from the accompanying chart I that durable goods have accounted for most of the fluctuations in total manufacturing inventories in the past decade, reflecting the greater volatility of the demand for these goods stemming from both private and Government sources.



CHART I

U. S. Department of Commerce, Office of Business Economics

76626-61-pt. I-----12

A summary of the major developments in durable and nondurable goods manufacturers' inventories over the past 13 years follows. Because of the larger fluctuations in durable goods, the movement of these inventories will be discussed in more detail.

#### DURABLE GOODS

Following a period of accumulation, durable goods manufacturers' stocks were reduced sharply in the second and third quarters of 1949. and liquidation continued at more moderate rates to the end of that year. Inventory movements during this period were greatly affected by strikes or the threat of strikes in the automobile and steel industries and by the filling of the production pipelines.

After the outbreak of the Korean conflict in mid-1950, manufacturers' inventories rose sharply. With wartime shortages still fresh in the public mind, waves of anticipatory buying occurred in the third quarter of 1950 and the first quarter of 1951. This increased demand, as well as business concern over the future course of supplies and prices, were important factors lifting rates of inventory accumulation to record highs in this period.

Stocks of durable goods producers continued to rise at moderate rates until mid-1952 when they leveled off mostly because of the effect of the steel strike. The subsequent buildup through the third quarter of 1953 was mainly a reaction to this temporary halt.

The liquidation of durable goods inventories during the 1953-54 business recession reflected the reduction in the national defense program, and declines in plant and equipment expenditures and in consumer durable goods demand. From the end of 1954 through the fall of 1957 there was a steady accumulation of hard goods stocks. The sharp rise in demand for automobiles and other consumer durables in 1955 provided an impetus to the building up of stocks during this period. In 1957 the rate of accumulation was greatly curtailed primarily due to improvement of supplies relative to demand.

The inventory liquidation in the 1957-58 recession followed a general lessening of the demand for hard goods, particularly producers' durable equipment and new cars. Inventory reduction reached a peak in the spring of 1958, and was halted by the end of that year, concomitant with the recovery in general business activity. This was followed by a sizable buildup in the first half of 1959, attributable in part to an accumulation of primary metals and related products, with a view to forestalling possible shortages due to the steel strike expected in the middle of the year. During the latter half of 1959, changes in producers' hard goods stocks were relatively small.

The year 1960 was characterized by wide swings in manufacturers' durable goods inventories—from substantial accumulation in the first half to liquidation in the latter half. The metal-using industries machinery, transportation equipment, and fabricated metals—accounted for most of the inventory shift. The rebuilding of steel stocks was an important factor in the accumulation in the first half of the year. Following this, inventory policies underwent modification in response to a volume of sales which was short of expectations. Liquidation continued in the first half of 1961, but at a more moderate rate. In the third quarter, inventories advanced in response to the sharp turnaround in business.

#### NONDURABLE GOODS

The movement of nondurable goods manufacturers' inventories over the postwar period reflected some of the same factors which influenced durable goods stocks. On the whole, however, nondurable goods inventories fluctuated within much narrower limits than hard goods, reflecting the comparative stability of the demand for these goods. The largest part of this demand stems from consumers, and has been supported throughout the postwar period by the relative strength of disposable personal income.

Before terminating this discussion of postwar inventory movements, it is of interest to note one more point. In the first quarter of 1949, inventories reached a peak of \$32 billion. The next two prerecession highs were \$46 and \$54 billion, respectively. Thus, in the recovery periods following both the 1949 and the 1954 downturns, manufacturing stocks were built up, not only to their prerecession peaks, but to levels substantially above these. The volume of \$55 billion reached just before the 1960 downturn, however, was only slightly above the previous peak made in 1957. Part of this advance reflects higher prices and the corresponding enhancement of book values of stocks.

#### FACTORS AFFECTING INVENTORY MOVEMENTS

Business inventories react in a highly sensitive and complex manner to shifts in economic conditions and changing patterns of demand. A part of manufacturers' holdings consists of goods actually in process of fabrication. The volume of such stocks obviously depends to a large extent upon the rate of production. The other segments of producers' stocks—purchased materials and finished goods—are also closely related to the rate of operations, since a total inventory must be maintained sufficient to support production and sales requirements.

Other factors which influence manufacturers' inventory changes include the trend of sales, new orders, the volume of unfilled orders, price pressures, the level of inventories, the ratio of stocks to sales, interest rates on business loans, the rate of capacity utilization in manufacturing—all with or without lag—as well as changes in these variables. Some of these may be eliminated a priori, others after some preliminary experimentation. For example, although variations in interest rates may have a bearing on the cost of carrying stocks, interest charges could account for only a small part of the spread between the cost of raw materials and the manufacturers' final price.

Businessmen are accustomed to using inventory-sales ratios in gaging their current inventory position. Thus, a brief description of the trends of these ratios may be of interest. The ratio is, of course, a special type of inventory-sales relation, but one which has certain definite limitations and restrictions for use in analysis.

Inventory book values and sales of all manufacturing industries increased by the same relative amount from 1948 to 1960, so that the stock-sales ratios for all manufacturers for both years are approximately the same. Furthermore, there is no discernable trend over this period. By stage of fabrication, however, the movement of inventories relative to sales shows marked differences in trend. For example, although durable-goods inventories in total kept pace with sales over the postwar years, the increase in purchased material stocks was considerably less. Nondurable purchased material inventories also declined relative to overall nondurable stocks and sales. Consequently, there is a striking downtrend in the stock-sales ratio for purchased materials. This may reflect improved inventory controls, speedier delivery, and better management in the work schedule and in the flow of materials.

#### BASIS OF ANALYSIS

This analysis covers the period from the beginning of 1948 to the first quarter of 1961, and utilizes quarterly seasonally adjusted data, with the manufacturing inventory data being the book values of stocks on hand at the end of the quarter. These differ from the inventory component of GNP discussed earlier, which includes an inventory valuation adjustment designed to remove the effect of changes in replacement costs.

Inventories in book values are used by businessmen in their own accounting. For many purposes, however, business firms judge their stock position relative to sales and orders on the basis of the number of physical units held, and not necessarily on the basis of the book value of the goods. There are certain basic considerations for using the latter series in the evaluation of inventory changes. In addition to the position of the physical volume of goods on hand, inventory decisions must take into account financial considerations—i.e., the availability of working capital and the cost of borrowing. There is also the statistical problem of securing appropriate price data for converting book values into constant dollars.

In this report a correlation analysis approach is used to analyze the differential impact of the various factors which influence inventory fluctuations, thus throwing light on the relative importance of demand and other forces. Other approaches which are primarily designed to yield forward-looking estimates include extensions of past trends and surveys of businessmen's intentions. Later in this paper the projections based on correlation techniques are compared with those of other approaches.

Two general procedures are used: first, a study of the factors affecting aggregate inventories, and second, those influencing changes in inventories. In addition, durable and nondurable stocks are considered separately, and, within these totals, analysis is made of changes in purchased materials, goods-in-process, and finished goods inventories, since the factors affecting each of these categories of stocks differ.

Because of the complexity of the inventory analyses, it was not possible logically to set forth, on theoretical grounds, a single formulation of the specific factors and relationship. The empirical approach used here involved selecting from a multiplicity of correlations the particular one which gave the closest fit, using linear regressions derived by the least squares approach.

#### BOOK VALUE OF INVENTORIES RELATED TO SALES

The most relevant indicator of manufacturing activity to relate to the book value of inventories appears to be manufacturers' sales. The accompanying chart I suggests that these two series move together but that inventories tend to lag behind sales. A comparison of the various peaks and troughs over the postwar period indicates a tendency for inventories to lag, on the average, by two quarters. An exception occurred in the fourth quarter of 1949, when both series were at their lows, apparently reflecting a tighter supply situation due in part to the steel strike that year.

The lag of inventories behind sales may be due to a combination of factors. Materials are supplied under conditions that make it difficult for manufacturers to alter immediately the rate at which they receive goods. Most raw materials are purchased some time before delivery, and some producers may not forecast with a sufficient degree of accuracy the rate at which they will need to consume these materials.

In addition, there appears to be some reluctance on the part of manufacturers to curtail operations until they are certain that business is falling off, and a further reluctance, even then, to cut down sufficiently to liquidate their finished goods stocks. The effect of this is an unplanned accumulation of stocks following a peak in business, and an unplanned liquidation immediately after a trough.

### NUMERICAL RELATIONSHIPS FOR INVENTORY BOOK VALUES

Regressions have been fitted, by the method of least-squares, to the dollar volume of inventories and to sales two quarters earlier, for all manufacturing, and for durable and nondurable goods separately. In each instance the lag was determined by examining various amounts and selecting that which gave the highest correlation coefficient. The fit is reasonably close in each case. A correlation coefficient ( $r^2$ ) of .96 for total manufacturing indicates that sales two quarters earlier accounted for 96 percent of the variations in inventories over the 13-year period. The average deviation between the actual values and those derived from the regression was \$1.3 billion per quarter, or 3 percent.

For durable goods inventories, the proportion explained amounted to 95 percent, with an average deviation per quarter of \$1 billion (4.3 percent).¹ Deviations of more than \$2 billion occurred in the first quarters of 1951, 1953, and 1957, the third and fourth quarters of 1958, and the first and fourth quarters of 1959. Many of these are due to the unusual supply situation at the time.

In the case of nondurable goods stocks, sales two quarters earlier accounted for 93 percent of the total variation over the postwar period. The deviations between actual and "calculated" averaged \$0.6 billion per quarter (3 percent), and exceeded \$1 billion only in the strike period in 1949, in the second quarter of 1951, an aftermath of the buying wave, and in the first quarter of 1952.

### INVENTORY CHANGES RELATED TO SALES AND ORDERS

A basic shortcoming in the application of regression analysis to time series in this form is the possibility of serial correlation. Both inventory book values and sales have a strong upward time trend. As a matter of fact, the values of  $r^2$  between each variable and a time factor are highly significant—0.87 for total manufacturing inventories

¹ When sales in the current and three preceding quarters are used, an additional 3 percent of the variation in durable goods inventories is explained, and the average deviation is reduced to \$0.7 billion. For total manufacturing, the proportion explained increases 2 percent, with an average deviation of \$1.1 billion. There is no appreciable change in the case of nondurables.

and 0.88 for sales. This must be considered in interpreting and using the regression results.

One way of resolving the problem posed by possible serial correlation is to correlate first differences instead of the level of inventories. with the various relevant factors. The following correlations were developed using changes in inventories as the dependent variable.

It is assumed that inventory changes are dependent upon past sales and upon businessmen's expectations regarding the near-term sales prospects, as indicated by their new orders and other information. The trend of the backlog of orders provides some guide to the course of operating in the period ahead. The difference between new orders and sales during any given period is used as a measure of the change in unfilled orders during that period.

In view of the time required to obtain delivery after orders have been placed, the question arises as to whether the change in inventories might reflect the change in unfilled orders in an earlier period more closely than that during the same quarter. Our analysis supports this hypothesis. The best results, of all the combinations tried, are obtained by using the change in unfilled orders in the preceding quarter. The correlations are further improved by including sales in the preceding quarter and the volume of stocks in the second preceding quarter.

#### RESULTS FOR DURABLE AND NONDURABLE GOODS MANUFACTURING INDUSTRIES

Based on linear regressions covering the postwar period, the fore-going three factors explain 72 percent (R=0.85) of the variance of the inventory change for durable goods and 65 percent (R=0.81) for nondurable goods industries. For all manufacturing, using the same variables, the  $R^2$  was 0.80² (R=0.90). These percentages, of course, apply to a much smaller base for inventory changes than is the case when the levels of inventories are considered.

The degree to which the equations reproduced the actual changes in inventories may be seen from the upper curves in each panel of the accompanying chart II which shows, for durable and nondurable goods and for all manufacturers, the actual inventory changes and those calculated from the regression equations. The bottom curve in each panel shows the differences between the actual and calculated These reflect variations not explained by the factors used values. in the equation. The average deviation per quarter is \$0.3 billion for durable goods, and \$0.2 billion for nondurable goods.³ The largest unexplained variations in durable goods inventory changes-\$1 billion or more-occurred during the third quarter of 1959, the steel strike period, in the rapid inventory buildup period of the first quarter in 1960, and in the fourth quarter of that year, when there was an un-usually large inventory liquidation by automobile dealers. In the case of the nondurables, all of the deviations were relatively small, with the largest (\$0.6 billion) occurring in the first quarter of 1951. The estimated inventories for the second quarter of 1961, based on the correlations, are very close to the actual, which have become available since the regressions were developed.

³ Paul G. Darling developed this regression for the postwar period through the third quarter of 1958. See "Manufacturers' Inventory Investment, 1947-1958," American Economic Review, December 1959. The slightly different results are due to the additional 2½ years. ³ The average of the inventory changes for the 53 quarters, without regard to sign, is \$1 billion for all manu-facturing compared with \$0.4 billion as the average deviation from the regression.

#### CHART II

## **MANUFACTURING INDUSTRIES:**

Change in Inventories, Actual Compared With Calculated



*Calculated from regression using average monthly sales and unfilled orders in preceding quarter, and inventories at end of 2d preceding quarter. See text table

U. S. Department of Commerce, Office of Business Economics

In addition to these correlations, another variant was tried, i.e., substituting deviations from trend for actual sales and inventories. The coefficient was about the same as given above for the durables, and was somewhat, but not significantly, improved for the nondurables. These and other correlations examined are summarized in table 2.

TABLE 2.—Selected regression equations for manufacturing inventory relationships

		Average	deviation
	R 3	Billion dollars	Percent
REGRESSIONS FOR INVENTORY LEVELS			
All manufacturing: $I_t = 760+1.811 S_{t-1}$ . $I_t =801+.465 S_t+.367 S_{t-1}+.270 S_{t-2}+.770 S_{t-3}$ . Durable goods industries: $I_t =771+2.210 S_{t-2}$ . $I_t =2152+.710 S_t+.352 S_{t-1}+.216 S_{t-2}+1.046 S_{t-3}$ . Nondurable goods industries: $I_t = 4.020+1.247 S_{t-2}$ . $I_t = 3.797015 S_t+.637 S_{t-1}+.476 S_{t-2}+.163 S_{t-3}$ .	0.96 .98 .95 .98 .93 .93	1.3 1.1 1.0 .7 .6 .6	3.0 2.6 4.3 3.0 3.0 2.9
REGRESSIONS FOR INVENTORY CHANGES			
All manufacturing: $\Delta I_{I} = .381 + .233 \Delta U_{I-1} $	.64 .80 .82 .72 .55 .64 .53 .29 .65 .72 .72 .72 .72 .72 .72 .72 .72 .72 .72	.5 .4 .4 .3 .3 .2 .2 .2 .1 .3 .2 .2 .2 .2	
$\Delta F_t =329 + .208 S_{t-1}260 F_{t-2}$	.53	1.	

NOTE -

t=current quarter, t-1=preceding quarter, etc. I=inventories at end of quarter.  $\Delta I=$ change in inventories during quarter.

S=average monthly sales. dI, dS=deviations from trend. U=unfilled orders.

 $\Delta U$ =change in unfilled orders.

P=purchased material inventories.

F = purchased material inventor. G = goods-in-process inventories. F = finished goods inventories. R = correlation coefficient.

=regressions shown in chart.

Regressions are based on the period from the 1st quarter of 1948 to the 1st quarter of 1961, using quarterly seasonally adjusted data, in billions of dollars.

Source: Office of Business Economics, U.S. Department of Commerce.

#### STAGES OF FABRICATION

Inventory holdings of a firm comprise a mixture of raw materials, goods-in-process, and finished goods, each of which is subject to a different set of influences. Also, the behavior of these three classes of stocks differs because the motives that control inventory policy vary at the different stages of fabrication. Thus, the amount of accumulation or liquidation in a period of changing business conditions is the net result of differing and sometimes offsetting movements in the various categories of inventories. Chart III shows the pattern of variation of these three components for durable and nondurable goods inventories.

Purchased materials account, on the average, for approximately 30 percent of producers' stocks, goods-in-process for another 30 percent, and finished goods for the remainder. Goods-in-process refer to goods the manufacturer has processed to some extent but has not converted completely to the form in which his goods are finally sold. They are to be distinguished from purchased materials which the manufacturer at his stage has not yet processed, and from finished goods which are ready for sale either to other producers or to the ultimate consumer.

Goods-in-process may be expected to fluctuate with current production, while stocks of purchased materials are generally geared to planned future output. Of course, other factors such as price movements, supply expectations, and cost of borrowing, influence businessmen's inventory policies. Each of these categories of working stocks may also contain unintended elements of increase or decrease in periods of changing business activity. As mentioned earlier, this is due to the inability of manufacturers to note such changes immediately and adjust their orders of raw materials accordingly, and to the fact that the rate of supply of some commodities cannot be altered rapidly.

The optimum amount of finished goods held by a firm is obviously the minimum with which orders can be efficiently met. Changes in these inventories generally reflect movements in sales, though less than proportionately. When sales are declining, an accumulation of finished goods is usually indicative also of involuntary additions to these stocks.

In deriving relations for the stages of fabrication within the durable goods total, sales and changes in unfilled orders in the preceding quarter were again the most important factors in accounting for fluctuations in purchased materials and goods-in-process stocks. For the latter, as for the total, the addition of the value of inventories two quarters earlier improved the relationship. For purchased materials, however, the inventory level at the end of the immediately preceding quarter was found to be more effective.

In the case of finished goods—both durable and nondurable—sales and inventories, one and two quarters earlier, respectively, were used. For nondurable purchased materials, the best relation utilized the same variables as for the overall nondurables, namely, sales and change in unfilled orders during the preceding quarter, and stocks at the end of the second preceding quarter. The relationship for each of these stages of fabrication accounted for approximately half of the total variation over the postwar period in the respective inventory change.



CHART III MANUFACTURING INDUSTRIES:

U. S. Department of Commerce, Office of Business Economics

No satisfactory relationship was found for goods-in-process inventories These stocks comprise only oneof nondurable goods industries. seventh of the total nondurables.

It is clear that the relationships for these components separately are less close than those obtained for the durable and nondurable goods It is the total inventory which is usually geared to the volume totals. of available working capital, which in turn is related to sales. If there is a piling up of finished goods when sales unexpectedly fall off, the manufacturer may be forced to reduce other components of stocks. Conversely, a sudden liquidation of finished goods, due perhaps to an unanticipated rise in sales, will increase working capital which may be used to finance additions to purchased material stocks.

#### COMPARISON WITH MANUFACTURERS' EXPECTATIONS BASED ON SAMPLE SURVEY

Since the fall of 1957, the Office of Business Economics of the Department of Commerce has carried on an exploratory survey of manufacturers' sales and inventory expectations. Following a series of tests, the results have proven to be sufficiently reliable to be released this year.⁴ It is of interest to compare the inventory anticipations obtained in this survey with those derived by the use of the foregoing regression equations.

TABLE 3.-Manufacturing inventory change-comparison of estimated with actual [Billions of dollars, seasonally adjusted]

	Actual	Calculated	Anticipated from surveys		
	10044	regression 1	lst	2d	
1959:					
3d quarter	-0.2	1.4	0.3	0.2	
4th quarter	.0	.3	3	i .7	
1st quarter	1.9	.9	1.5	1.6	
2d quarter	.8	.7	.6	.2	
3d quarter	4	.4	8	3	
4th quarter	-1.0	.2	-1.5	-1.0	
1961:					
1st quarter	4	1	0	3	
2d quarter	U 1	.2	3		
Ath quarter	1,1	.9	1.1	1.2	
2011 Quar vor		1. 2	1.1		

¹ Using sales and change in unfilled orders in the preceding quarter, and the value of inventories at the

and of the second preceding quarter. ¹ OBE survey of "Manufacturers" Inventory and Sales Expectations," reported in the August and October 1961 issues of the Survey of Current Business.

Source: Office of Business Economics, U.S. Department of Commerce.

In the accompanying table 3 the inventory changes derived from one of the best correlations of those tried are compared with the published expectations and with the actual inventory changes for all manufacturing industries. The comparison shows, on the whole, that the correlations gave better results compared to actual in two quarters out of nine and somewhat poorer results in the other quarters. However, in using correlations for forecasting purposes, it is essential that the preliminary forecasts obtained from the regressions should be modified, as indicated by an examination of the final residuals.

⁴ The method of deriving the anticipations is described in "Manufacturers' Inventory and Sales Expec-tations" in the August 1961 issue of the Survey of Current Business.

	All manufacturing			Durable goods indus- tries			Nondurable goods in- dustries		
	Inven- tories	Mon aver	thly ages	Inven- tories	Mon aver	thly ages	Inven- tories	Mon aver	thly ages
	quar- ter)	Sales	New orders	quar- ter)	Sales	New orders	quar- ter)	Sales	New orders
10/9+									
1st quarter	29.5	17.3	17.1	14.4	7.4	7.2	15.1	9.9	9.8
2d quarter	30.3	17.6	17.5	14.7	7.4	7.6	15.6	10.2	9.9
30 Quarter	31.3	17.8	17.0	10.4	7.7	7.0	16.0	9.9	10.0
1949:	01.7	11.0	11.2	10.1			10.0	0.0	
1st quarter	31.8	17.0	15.9	16.1	7.5	6.7	15.7	9.5	9.3
2d quarter	30.6	16.5 16.4	15.4	16.3	7.1	6.2	10.3	9.3	9.2
4th guarter	28.9	15.7	16.1	14.0	6.5	6.9	14.9	9, 2	9.2
1950:									
1st quarter	29.2	16.6	17.1	14.1	7.3	7.7	15.1	9.4	9.4
3d quarter	31.0	20.9	24.4	15.1	9.7	12.6	15.9	11.3	11.8
4th quarter	34.3	21.3	23.0	16.8	10.0	11.7	17.5	11.3	11.4
1951:			07.4	10.1	10.4	14.7	10.0	10.0	10 7
2d quarter	30.9	22.0 22.4	27.4	20 1	10.4	14.7	10.0	11.9	12.7
3d quarter	41.9	22.0	22.8	21.8	10.2	11.4	20.1	11.8	11.3
4th quarter	42.8	22.3	23.3	22.8	10.4	11.6	20.0	11.8	11.8
1952:	13.0	22.2	93.1	93.7	10.5	11.5	20.2	11 7	11 7
2d guarter	43.3	22.2	23.4	23.7	10.5	11.6	19.6	11.7	11.8
3d quarter	43.2	22.5	23.5	23.7	10.6	11.7	19.5	11.9	11.8
4th quarter	43.8	24.3	24.1	24.4	12.1	11.8	19.4	12.3	12.3
1900: Ist quarter	44.2	24.6	24.4	25.0	12.5	12.4	19.3	12.1	12.1
2d quarter	45.4	24.8	24.1	25. 9	12.6	12.0	19.5	12.2	12.1
3d quarter	46.0	25.1	22.6	26.5	12.7	10.4	19.6		12.2
4th quarter	45.4	23.0	21.2	20.2	11.0	9.3	19,2	11.9	11.9
1st quarter	44.5	23.5	21.6	25.5	11.4	9.5	19.0	12.1	12.1
2d quarter	43.6	23.4	21.8	24.4	11.2	9.6	19.2	12.2	12.2
3d quarter	42.7	23.5	22.7	23.7	11.2	10.3	18.9	12.4	12.4
1955:	40.0	20.1	20.0	24.1	11. 2	11.4	10.0	12.0	12.0
1st quarter	43.2	25.2	25.8	24.2	12.3	12.8	19.0	12.9	13.0
2d quarter	43.7	26.1	26.4	24.4	12.9	13.2	19.3	13.2	
4th quarter	44.9	20.9	27.9	26.7	13.6	14.0	19.5	13.5	13.6
1956:									
1st quarter	48.0	27.4	27.9	27.9	13.6	14.2	20.1	13.7	13.7
20 Quarter	49.h	27.5	28.2	28.8	13.0	14.4	20.9	13.9	13.8
4th guarter	52.3	28.7	29.2	30.7	14.4	14.9	21.6	14.3	14.4
1957:		~ ~	<u> </u>			14.0		74 8	
Ist quarter	53.3	29.3	28.5	31.2	14.0	14.0	22.1	14.7	14.0
3d guarter	54.2	28.6	27.1	31.8	14.3	12.9	22.3	14.3	14.2
4th quarter	53.5	27.3	25.8	31.1	13.5	12.0	22.4	13.8	13.8
1958:	59.0	95.6	01.1	20.0	19.1	11.0	00 1	13.5	19.5
2d quarter	50.2	25.3	24.4	29.9	11.8	11.5	21.7	13.5	13.6
3d quarter	49.3	26.5	26.5	28.0	12.5	12.5	21.2	14.0	14.0
4th quarter	49.2	27.6	28.0	27.8	13.3	13.6	21.4	14.3	14.4
1909: 1st quarter	50.5	28.6	29.5	28.9	13.9	14.7	21.5	14.6	14.8
2d quarter	52.1	30.8	31.0	30.2	15.5	15.7	21.9	15.3	15.3
3d quarter	51.9	30.0	30.1	29.8	14.5	14.7	22.1	15.5	15.4
4th quarter	52.4	29.7	30.1	30.1	14.2	14.5	22.3	15.5	15.6
1st guarter	54.3	31.2	30.2	31.8	15.4	14.5	22.6	15.7	15.7
2d quarter	55.1	30.9	30.3	32.2	15.0	14.5	22.9	15.9	15.8
3d quarter	54.7	30.2	29.9	31.8	14.5	14.3	22.9		15.6
4611 quarter	00.1	∠9.3	29.0	30.9	10.0	10.0	22.9	10.0	10.0
1st quarter	53.3	29.1	29.2	30.3	13.4	13.4	23.0	15.7	15.8
2d quarter	53.4	30.6	30.9	30.2	14.5		23.2		16.2
au quarter	09.4	51.4	31.9	31.1	1 18.8	10.0	40.4	10.4	10.0

#### TABLE 4.-Manufacturers' inventories, sales and new orders

[Billions of dollars, seasonally adjusted]

Source: Office of Business Economics, U.S. Department of Commerce.

These adjustments may be suggested by studying the residuals for systematic tendencies or regular patterns and by other guides indicated by the correlation as a whole; special or temporary factors which are known to be influences in the forecast period should also be taken into account.

It should be noted that the direct results obtained from the survey approach were not close to the actual until adjustments were made in the light of other information, such as manufacturers' judgments with regard to their inventory position.

#### CONCLUSION

The foregoing analysis of manufacturers' inventories has involved the determination of relationships with the volume of unfilled orders, the direction and strength of demand as reflected by sales, and the prior levels of inventories. As various empirical formulas were examined, including those involving both levels and changes in these factors, and with various leads and lags, it became clear that the basic relationships were perhaps more complex than those utilized, and, more importantly, that other factors and considerations than those used in the correlations had at times greater influence on inventory changes. Among such factors would be manufacturers' views regarding the inventory size relative to sales and orders, costs of carrying inventories, and improved methods of controlling the inventory flow.

The correlation approach using the measurable factors just mentioned yields the following conclusions as to their relative importance in influencing inventory investment:

(1) The most important factor accounting for the variations in the book value of manufacturers' inventories is the level of manufacturers' sales two quarters earlier. This single factor accounted for 96 percent of the total variance for all manufacturing inventories, 95 percent for durables, and 93 percent for nondurables.

(2) In contrast to the analysis of the level of inventories, the correlations utilizing the changes in the book values yield smaller coefficients. In the case of all manufacturing industries, 80 percent of the total variation in the inventory change is accounted for by changes in unfilled orders and the level of sales for the immediately preceding quarter, and the level of inventories two quarters earlier. For durable and nondurable goods manufacturers' stocks, and, within these, by stage of fabrication, the proportion of the total variation in the inventory change explained ranged from 45 to 72 percent.

(3) Changes in manufacturers' inventories during any period reflect not only adjustments dictated by current or anticipated needs, but also unplanned accumulations or liquidations resulting from unforeseen factors. This is apparent from the size of the variation not explained by the correlation procedure.

(4) Because of the large unexplained element in this approach, great care must be exercised in evaluating the correlation results. Further analytical work is needed with a view to sharpening this approach, perhaps by the introduction of additional factors as well as other types of relationships. As additional survey results become available, some combination of the survey and correlation approaches may also yield better results and throw further light on this most dynamic sector of our economy.

#### PART II. INVENTORY INVESTMENT IN TRADE

This part of the paper consists of two sections. Section A deals with the postwar trends in retailers' and wholesalers' inventories, examining their response to changes in economic activity, in distributors' sales and consumer demand, and other developments. The composition of stocks and longer term movements in stock-sales ratios are also discussed.

Section B attempts to quantify the economic factors affecting inventory investment through correlation analyses.

#### A. POSTWAR DEVELOPMENTS IN TRADE INVENTORIES

The book values of merchandise stocks held by retail and wholesale establishments have each grown at a compounded rate of about 4 percent a year since 1948. With consumer and wholesale prices rising at rates of over 1 percent per annum over this period, the growth rate in stocks has been 3 percent in physical volume terms.

This growth in retail stocks was fully matched by the advance in retail sales in the 1948-61 period; wholesale stocks, however, have risen relatively more than sales. This experience is in contrast with that prior to World War II. From 1929 to 1940, for example, there was a noticeable decline in the ratio of stocks to sales. (See tables 5-7.)

•

## TABLE 5.—Retail trade: Inventories, 1947-61

[Seasonally adjusted in billions of dollars]

		D	urable g	oods stor	es	Nor	adurable	goods ste	ores
	Total	Total	Auto- motive	Furni- ture and appli- ances	Lum- ber, build- ing, and hard- ware	Total	Food	Ap- parel	Gen- eral mer- chan- dise
1947: 4th quarter	14.24	5.35	1.53	1.24	1.28	8.90	1.68	1.90	2.82
1993: 1993: 2d quarter 2d quarter 3d quarter 4th quarter	15. 23 15. 44 15. 73 16. 01	6. 04 6. 11 6. 29 6. 57	$1.63 \\ 1.72 \\ 1.65 \\ 1.99$	1.42 1.45 1.50 1.48	$1.53 \\ 1.46 \\ 1.62 \\ 1.53$	9, 20 9, 32 9, 44 9, 44	1.65 1.74 1.77 1.78	1.99 2.18 2.18 2.13	2.97 2.88 2.90 2.86
1949: 1st quarter 2d quarter 3d quarter 4th quarter	15. 76 15. 44 16. 24 15. 47	6.61 6.33 7.14 6.26	2, 22 2, 05 2, 73 1, 88	$1.29 \\ 1.20 \\ 1.23 \\ 1.27$	$1.51 \\ 1.52 \\ 1.53 \\ 1.53 \\ 1.53$	9.15 9.11 9.10 9.21	1.66 1.69 1.66 1.72	2.15 2.11 2.16 2.11	2.76 2.76 2.75 2.87
1950: 1st quarter 2d quarter 3d quarter 4th quarter	15.78 16.58 17.98 19.46	6.26 6.82 7.23 8.29	$1.78 \\ 2.10 \\ 2.05 \\ 2.46$	$1.33 \\ 1.41 \\ 1.62 \\ 1.88$	$1.60 \\ 1.70 \\ 1.85 \\ 2.10$	9.52 9.76 10.75 11.17	1.84 1.94 2.12 2.17	2, 23 2, 19 2, 37 2, 49	2.94 2.96 3.26 3.51
1991: 1st quarter 2d quarter 3d quarter 4th quarter	21. 58 21. 83 21. 21 21. 05	9.26 9.73 9.62 9.63	2, 79 2, 98 2, 99 3, 13	2.14 2.22 2.03 1.95	2.26 2.36 2.44 2.38	$\begin{array}{c} 12.32\\ 12.10\\ 11.59\\ 11.42 \end{array}$	2, 35 2, 19 2, 14 2, 18	2.68 2.70 2.56 2.49	3, 87 3, 93 3, 73 <b>3, 59</b>
1952: 1st quarter2d quarter3d quarter3d quarter3d quarter4th quarter	$\begin{array}{c} 20.\ 65\\ 20.\ 25\\ 20.\ 46\\ 21.\ 03 \end{array}$	9, 32 9, 02 9, 02 9, 49	2.97 2.77 2.74 3.03	1.86 1.84 1.88 1.90	2, 38 2, 35 2, 33 2, 40	11.33 11.23 11.44 11.54	2. 20 2. 22 2. 22 2. 20 2. 20	2, 44 2, 32 2, 41 2, 49	3, 56 3, 54 3, 64 3, 74
1953: 1st quarter 2d quarter 3d quarter 4th quarter	$21.22 \\ 21.57 \\ 21.84 \\ 21.49$	9,68 9,90 10,14 9,78	3.07 3.21 3.52 3.28	1.94 1.93 1.91 1.90	2.46 2.50 2.42 2.34	11.54 11.67 11.70 11.71	2.13 2.11 2.23 2.29	2, 43 2, 55 2, 59 2, 62	3, 83 3, 86 3, 75 <b>3, 69</b>
1954: 1st quarter 2d quarter 3d quarter 4th quarter	21.46 21.30 21.24 20.93	9, 59 9, 54 9, 47 9, 27	3. 21 3. 20 3. 21 3. 01	1.88 1.86 1.81 1.78	2. 27 2. 26 2. 27 2. 28	11.87 11.77 11.77 11.66	2.43 2.46 2.42 2.42 2.47	2. 77 2. 71 2. 73 2. 60	3, 58 3, 44 3, 43 3, 40
1955: 1st quarter 2d quarter 3d quarter 4th quarter	$21.71 \\ 21.98 \\ 22.19 \\ 22.77$	9.78 10.10 10.19 10.53	3.35 3.62 3.66 4.01	1.80 1.84 1.87 1.88	2.32 2.33 2.39 2.36	11.93 11.88 12.00 12.24	2, 44 2, 46 2, 45 2, 56	2. 84 2. 74 2. 74 2. 68	3, 51 3, 56 3, 62 3, 71
1956: 1st quarter	23. 05 23. 13 23. 14 23. 43	10. 73 10. 44 10. 37 10. 53	3.93 3.68 3.60 3.75	1.93 1.93 1.95 1.96	2. 48 2. 42 2. 39 2. 39	12.32 12.69 12.76 12.90	2. 64 2. 73 2. 74 2. 72	2, 65 2, 85 2, 88 2, 91	3, 76 3, 82 3, 85 3, 84
1957: 1st quarter 2d quarter 3d quarter 4th quarter	23. 67 23. 84 24. 43 24. 57	10.58 10.75 11.30 11.41	3.97 4.09 4.64 4.64	1.93 1.93 1.90 1.92	2, 34 2, 35 2, 32 2, 40	13.10 13.09 13.13 13.16	2.72 2.78 2.85 2.85	2.96 2.92 2.96 3.03	3. 84 3. 87 3. 83 3. 84
1998: 1st quarter 2d quarter 3d quarter 4th quarter	24.06 23.81 23.80 24.29	10.88 10.50 10.38 10.71	4. 25 3. 88 3. 68 4. 14	1.87 1.88 1.90 1.88	2.39 2.38 2.43 2.41	13. 18 13. 32 13. 42 13. 58	2.92 2.93 2.92 2.95	2.94 2.91 2.92 2.97	3. 78 3. 78 3. 80 3. 86
1555. 1st quarter	24. 41 25. 16 25. 32 25. 54	10.78 11.27 11.27 11.27 11.27	4.15 4.54 4.42 4.32	1.90 1.91 1.93 1.96	2.38 2.40 2.42 2.45	13. 63 13. 88 14. 05 14. 27	2.90 2.94 2.97 2.98	3.00 3.14 3.16 3.22	3.90 3.96 4.07 4.18
1900: 1st quarter 2d quarter 3d quarter 4th quarter	26. 42 26. 62 26. 82 27. 18	11.82 11.97 12.04 12.33	4.82 4.96 5.02 5.27	1.97 2.00 1.99 1.95	2. 46 2. 45 2. 43 2. 44	14. 61 14. 65 14. 78 14. 85	3.07 3.08 3.08 3.14	3, 31 3, 26 3, 32 3, 36	4.25 4.30 4.31 4.26
1901: 1st quarter 2d quarter	26.07 26.25	11.35 11.38	4. 37 4, 44	1.90 1.90	2. 40 2. 38	14.72 14.87	3.18 3.16	3. 26 3. 33	4.23 4.28

¹ Inventories are book values, end of quarter.

Source: U.S. Department of Commerce.

## TABLE 6.—Retail trade: Sales, 1947-61

[Seasonally adjusted in billions of dollars]

	1		Durable	goods stor	es	N	ondurab	le goods :	stores
	Total	Total	Auto- motive	Furni- ture and appli- ances	Lumber, building, and hard- ware	Total	Food	Ap- parel	General merchan- dise
1947: 4th quarter	10.75	3. 40	1.61	0.60	0.68	7.35	2.40	0. 81	1.39
1st quarter 2d quarter 3d quarter 4th quarter	10. 92 11. 10 11. 27 11. 22	3. 49 3. 48 3. 66 3. 66	1.68 1.62 1.77 1.84	. 60 . 62 . 64 . 60	. 69 . 72 . 72 . 68	7.43 7.61 7.62 7.56	2.46 2.55 2.52 2.53	. 81 . 82 . 84 . 86	1.39 1.43 1.46 1.44
1st quarter 2d quarter 3d quarter 4th quarter	11.08 11.24 11.15 11.22	3, 59 3, 80 3, 85 3, 82	1, 83 2, 04 2, 06 1, 95	. 58 . 59 . 63 . 68	.65 .65 .65 .68	7. 49 7. 44 7. 30 7. 41	2.52 2.51 2.50 2.50	. 84 . 81 . 76 . 77	, 1.37 1.38 1.35 1.34
1st quarter 2d quarter 3d quarter 4th quarter 1951:	11. 55 11. 99 13. 11 12. 47	4. 12 4. 42 5. 07 4. 50	2. 23 2. 40 2. 71 2. 41	. 68 . 68 . 85 . 72	.71 .80 .90 .82	7.43 7.57 8.04 7.96	2.56 2.59 2.74 2.76	. 76 . 78 . 84 . 80	1.33 1.40 1.57 1.46
1st quarter 2d quarter 3d quarter 4th quarter 1952:	13. 54 12. 79 12. 82 13. 04	5. 04 4. 44 4. 44 4. 35	2. 62 2. 30 2. 23 2. 23 2. 23	. 81 . 66 . 68 . 71	. 95 . 86 . 81 . 78	8.50 8.35 8.48 8.69	2.95 2.98 3.00 3.05	. 85 . 83 . 85 . 87	1.56 1.47 1.49 1.54
1st guarter 2d guarter 3d guarter 4th guarter 1953:	13.06 13.60 13.39 14.07	4. 43 4. 77 4. 40 4. 83	2, 18 2, 51 2, 56 2, 60	. 74 . 73 . 74 . 77	. 85 . 85 . 85 . 84	8.64 8.83 8.99 9.24	3.06 3.15 3.24 3.24	.85 .88 .88 .94	1.50 1.54 1.56 1.64
1st quarter 2d quarter 3d quarter 4th quarter 1954:	14. 21 14. 24 14. 04 13. 90	5. 17 5. 13 4. 96 4. 89	2. 83 2. 85 2. 71 2. 72	. 78 . 77 . 75 . 75	. 88 . 85 . 88 . 86	9.04 9.11 9.08 9.02	3. 22 3. 25 3. 29 3. 28	.90 .90 .84 .81	1.56 1.60 1.59 1.58
1st quarter 2d quarter 3d quarter 4th quarter	13. 71 14. 12 14. 05 14. 39	4. 68 4. 88 4. 83 4. 98	2. 51 2. 71 2. 59 2. 74	.77 .76 .74 .75	. 82 . 80 . 86 . 90	9. 04 9. 24 9. 23 9. 40	3. 26 3. 28 3. 34 3. 43	. 84 . 86 . 83 . 85	1.51 1.58 1.57 1.60
1st quarter 2d quarter 3d quarter 4th quarter 1956:	14. 77 15. 19 15. 52 15. 65	5. 27 5. 53 5. 75 5. 71	3. 00 3. 16 3. 30 3. 26	. 82 . 83 . 85 . 85	. 86 . 93 . 93 . 94	9. 50 9. 66 9. 77 9. 94	3. 45 3. 47 3. 50 3. 57	. 88 . 89 . 90 . 91	1.62 1.66 1.70 1.69
1st quarter 2d quarter 3d quarter 4th quarter 1957:	15. 44 15. 67 15. 95 16. 15	5. 42 5. 40 5. 46 5. 66	3. 03 2. 94 2. 93 3. 16	.87 .89 .90 .90	. 92 . 96 . 95 . 89	10. 02 10. 27 10. 48 10. 49	3. 59 3. 62 3. 73 3. 79	. 92 . 95 . 99 . 99	1.69 1.73 1.77 1.72
1st quarter 2d quarter 3d quarter 4th quarter 1958:	16. 32 16. 62 17. 00 16. 71	5. 71 5. 73 5. 76 5. 59	3. 25 3. 24 3. 25 3. 14	- 89 - 89 - 89 - 87	. 86 . 88 . 92 . 88	10.60 10.89 11.24 11.12	3.82 3.95 4.07 4.09	.98 1.00 1.06 1.04	1.73 1.75 1.84 1.74
1st quarter 2d quarter 3d quarter 4th quarter 1959:	16. 29 16. 55 16. 72 17. 17	5. 19 5. 18 5. 18 5. 57	2.83 2.77 2.70 2.99	. 86 . 84 . 87 . 87	.85 .89 .91 .93	11.09 11.37 11.54 11.60	4. 15 4. 22 4. 18 4. 22	1.02 1.02 1.06 1.07	1.71 1.78 1.87 1.82
1st quarter 2d quarter 3d quarter 4th quarter 1960:	17.65 18.12 18.06 17.89	5. 92 6. 13 6. 01 5. 79	3. 28 3. 42 3. 30 3. 11	. 89 . 93 . 94 . 92	.96 1.02 .99 .97	11. 73 11. 99 12. 05 12. 10	4. 28 4. 31 4. 29 4. 35	1.07 1.10 1.11 1.12	1.86 1.96 1.99 1.97
ist quarter 2d quarter 3d quarter 4th quarter 1961:	18. 16 18. 59 18. 14 18. 28	5. 93 6. 09 5. 73 5. 83	3. 36 3. 43 3. 13 3. 24	. 90 . 90 . 89 . 86	. 94 . 97 . 94 . 92	12. 23 12. 50 12. 40 12. 45	4. 42 4. 52 4. 48 4. 53	1. 15 1. 15 1. 15 1. 12	1. 97 2. 02 2. 00 2. 00
1st quarter 2d quarter	17.90 18.04	5. 42 5. 51	2. 93 3. 03	. 83 . 86	. 92 . 89	12. 48 12. 50	4. 58 4. 60	1. 14 1. 10	2.00 2.02

1 Sales are average monthly sales for quarter.

Source: U.S. Department of Commerce.

76626-61-pt. I----13

	Inventories 1			Sales *				
	Total	Durable goods establish- ments	Nondur- able goods establish- ments	Total	Durable goods establish- ments	Nondur- able goods establish- ments		
1947: 4th quarter	7.12	3. 07	4. 05	7. 31	2. 34	4.98		
1948: 1st quarter 2d quarter	7.27 7.58	3. 20 3. 39	4.07 4.19	7. 44 7. 42	2. 41 2. 43	5.02 4.99		
3d quarter	7.90 7.87	3.60 3.68	4.30 4.19	7.57 7.59	2.45 2.44	5.12 5.15		
1949. 1st quarter 2d quarter 3d quarter 4th quarter	7.83 7.61 7.63 7.59	3. 72 3. 57 3. 53 3. 52	4. 11 4. 05 4. 10 4. 08	7.37 7.15 7.01 7.21	2.35 2.30 2.19 2.23	5. 02 4. 85 4. 82 4. 98		
1950: 1st quarter 2d quarter 3d quarter	7.75 8.16 8.36	3.56 3.80 3.78 4.34	4. 19 4. 36 4. 59 4. 78	7.41 7.85 9.36 8.98	2.39 2.68 3.37 3.28	5.02 5.17 5.99 5.70		
4th quarter 1951: 1st quarter 2d quarter	9.12 9.72 10.08 9.88	4.57 4.98 4.96	5.15 5.01 4.92	9.84 9.31 9.16	3. 62 3. 25 3. 13	6. 22 6. 06 6. 03		
4th quarter 1952: 1st quarter	9.71 9.59	4.82 4.79	4.89 4.80	9.20 9.21 9.34	3. 19 3. 19 3. 20	6.01 6.02 6.15		
2d quarter 3d quarter 4th quarter 1953:	9.48 9.72 10.01	4.00 4.71 4.92	5. 02 5. 09	9.64 9.77	3. 25 3. 45	6.40 6.32		
1st quarter 2d quarter 3d quarter 4th quarter	10. 20 10. 39 10. 62 10. 47	5.15 5.10 5.30 5.14	5.04 5.26 5.32 5.32	9.66 9.82 9.95 9.62	3. 48 3. 48 3. 44 3. 35	6.18 6.34 6.50 6.27		
1954: 1st quarter 2d quarter 3d quarter 4th ouarter	10. 51 10. 44 10. 29 10. 39	5. 02 5. 05 5. 08 5. 08 5. 08	5. 49 5. 39 5. 20 5. 31	9.63 9.70 9.60 10.01	3. 21 3. 27 3. 36 3. 53	6. 42 6. 43 6. 24 6. 48		
1955: 1st quarter 2d quarter 3d quarter 4th quarter	10.56 10.87 11.11 11.44	5. 15 5. 40 5. 56 5. 82	5. 41 5. 47 5. 54 5. 63	10. 24 10. 57 10. 63 11. 04	3. 73 3. 93 4. 12 4. 30	6.50 6.65 6.51 6.75		
1956: 1st quarter 2d quarter 3d quarter 4th quarter	11.86 12.18 12.59 12.95	6. 05 6. 20 6. 37 6. 58	5.80 5.99 6.22 6.38	11. 05 11. 21 11. 21 11. 58	4. 43 4. 48 4. 28 4. 38	6.62 6.72 6.92 7.20		
1957: Ist quarter 2d quarter 3d quarter 4th quarter	12.84 12.70 12.82 12.71	6.53 6.57 6.73 6.65	6. 31 6. 14 6. 10 6. 06	11. 54 11. 43 11. 30 10. 94	4.48 4.30 4.16 3.89	7.07 7.13 7.14 7.05		
1958: Ist quarter 2d quarter 3d quarter 4th quarter	12.38 12.12 12.06 11.99	6. 37 6. 21 6. 16 6. 26	6. 01 5. 90 5. 91 5. 73	10. 51 10. 75 11. 17 11. 60	3. 67 3. 74 3. 95 4. 20	6.84 7.01 7.22 7.40		
1959: 1st quarter 2d quarter 3d quarter 4th quarter	11. 95 12. 37 12. 53 12. 65	6. 34 6. 56 6. 49 6. 55	5. 61 5. 80 6. 05 6. 09	11. 96 12. 50 12. 41 12. 36	4. 40 4. 81 4. 68 4. 60	7.55 7.69 7.72 7.76		
1960: 1st quarter 2d quarter 3d quarter 4th ouarter	12.83 13.04 13.09 13.21	6. 78 6. 91 6. 89 6. 81	6. 05 6. 14 6. 19 6. 40	12. 39 12. 54 12. 27 12. 25	4. 62 4. 58 4. 41 4. 27	7.77 7.96 7.86 7.98		
1961: 1st quarter 2d quarter	13. 28 13. 50	6. 69 6. 72	6. 59 6. 77	12.40 12.63	4.23 4.26	8, 18 8, 36		

## TABLE 7.—Wholesale trade: Inventories and sales, 1947-61 [Seasonally adjusted in billions of dollars]

Inventories are book values, end of quarter. Sales are average monthly sales for quarters.

Source: U.S. Department of Commerce.

While the postwar period has been one of general growth, it has also been affected by several important cyclical fluctuations. These recessions and recoveries, though relatively mild in form and short in duration, had important impacts on all sectors of the economy. The impact on trade inventories has generally been indirect and followed the reaction of consumer buying to income changes.

Although consumer expenditures have shown less volatility in cyclical periods than most other types of demand—helped in part by the "built in" income stabilizers and supplementary Government actions their reaction has still been appreciable. This has been especially true of purchases of automobiles and other consumer durable goods, given their high sensitivity to economic fluctuations and uncertainties as to future income on the part of consumers. In this respect, it has been found that changes in income are about as significant as the level of income in determining the size of the auto market.

These movements in consumer demand have their impact on the volume and composition of trade inventories. (Charts IV and V show the postwar pattern of fluctuations in certain categories of trade stocks.) Changes in these stocks, which in aggregate account for about two-fifths of all nonagricultural business investment in inventories, in turn have a significant effect on the behavior of the overall economy at specific periods in time.

An understanding of the relationships involved should be of value as an aid in the analysis of business cycles. In this study the question is examined as to whether there is any consistency in the manner in which wholesale and retail trade organizations invest in inventories and the nature of the relationship of this investment policy to sales, price changes, and other pertinent variables. The form of the relationships derived and their behavior over time would serve to cast some light on the character of overall cyclical patterns.

#### COMPOSITION OF TRADE INVENTORIES

The book value of trade inventories totaled \$39 billion at the end of 1960—with some two-thirds held by retail stores and chainstore warehouses. While there is a close tie-in between wholesale and retail establishments, given the supplier-distributor relationships, there are significant differences in the composition of stocks at the two distributive levels.

First, while retail inventories consist primarily of finished goods going directly to the consumer, wholesale stocks contain not only such finished goods but also a substantial amount of intermediate products, such as metals and inedible farm products, and machinery and equipment destined for the capital goods market.

Secondly, new motor vehicles generally do not pass through wholesale trade but rather move directly from manufacturers to retail dealers. New car stocks account for about one-tenth of total retail inventory book values; more importantly, they are responsible for the preponderant share of the month-to-month change in retail inventories, thus making stocks at retail stores considerably more volatile than those at wholesale establishments.



Coto: Census & OBE

170

INVENTORY FLUCTUATIONS AND ECONOMIC

STABILIZATION

CHART V



i say na tikon a tuk a lu di. Din alama na tahu kaomini ati at

Thirdly, some wholesale functions do not have retail counterparts. Most important among these is exporting, with most commercial exports going through wholesalers. Agents and brokers are also important in wholesale sales, although these commission merchants do not take title to goods, and so are not a factor in the inventory picture.

On the other hand, there is some overlapping in functions between the two types of trade, with both groups selling partly at retail and partly at wholesale. Retail establishments are defined as stores offering goods for sale to the public at retail and where such sales comprise more than half of total sales. A similar sales criterion applies in the classification of wholesale establishments. It is not felt, however, that the amounts of overlapping significantly affect the overall movements.

At the end of 1960 retail durable goods stores held over two-fifths of all retailers' inventories; wholesale establishments, stocks were about equally divided between durables and nondurables. In both cases, durable goods accounted for smaller percentages of sales, so that durable goods distributors were carrying higher stocks per dollar of sales than were the nondurable goods group.

This is not the case throughout trade, however. The stock-sales ratios for apparel and general merchandise stores (which, of course, also carry some durable goods) are generally as high or higher than those for automotive and homefurnishings and appliance dealers. The relatively low ratio for the nondurable goods group stems primarily from food stores where stocks are equivalent to less than 1 month's sales. Inventories at apparel and lumber, building materials, and hardware stores are usually equal to  $2\frac{1}{2}$  to 3 months of sales, general merchandise and furniture and house furnishing stores typically operate with stock-sales ratios of somewhat over 2, while automotive dealers' ratios are generally closer to  $1\frac{1}{2}$ .

Retail automotive dealers in recent years have accounted for about 40 percent of durable goods stocks, with the remainder fairly equally divided between the furniture-appliance, the lumber-hardware, and the "all other" groups. Apparel and food stores each hold somewhat over 20 percent of total nondurable goods stocks, while general merchandise stores account for almost 30 percent.

These proportions have, of course, changed over the postwar period. In 1948, durable goods stocks, and more particularly those of automotive dealers, had still not fully recovered from their abnormally low wartime position. Inventories held at durable goods stores accounted for 40 percent of total retail inventories in 1948 as compared to about 43½ percent in mid-1961. The proportion of automotive dealers' stocks to durable goods stocks rose from 30 to 39 percent over this period. In the nondurables, food stores' inventories have shown greater growth since 1948 than have stocks held by apparel and general merchandise stores. The considerable diversification of food stores into nonfood merchandise lines over this period is largely responsible for these differential movements.

#### POSTWAR INVENTORY MOVEMENTS

In the early postwar years households and businesses alike strove to increase their stocks of goods which had been held down to low levels in the depression and war years. The effect of this pent-up demand is evidenced in both wholesale and retail trade by the strong sales advances as well as by sharply rising inventories with stocks of durables increasing by far the most. The rise in stocks considerably outpaced that in sales: From yearend 1945 to 1948 retail stock-sales ratios rose from 1.2 to over 1.4. The comparable rise in wholesale trade was from 0.9 to 1 month of sales. Book values of trade inventories almost doubled over the 3 years, reaching \$24 billion in late 1948, with about one-half of the rise reflecting the rapid advance in prices following the elimination of price controls.

After recovering to a large extent from these war-induced shortages trade inventories became more responsive to overall economic fluctuations. As demand weakened somewhat in the recession period 1948– 49 the inventory investment policy of businessmen became more cautious. Total trade inventories began to fall off at the end of 1948 declining almost \$1 billion to reach their low point at the end of 1949. Stock liquidation was an important factor in the decline of national output in this period.

Thereafter, as production mounted based upon a resurgence of demand in the upswing in the cycle and then even more rapidly under the impact of the Korean conflict, sales as well as inventory accumulation pressed ever upward. The sharp peaks in sales during the Korean "scare buying" periods of the third quarter of 1950 and the first quarter of 1951 found retailers temporarily short of stocks. As a result a period of sizable stock building ensued. Trade stocks reached \$32 billion by mid-1951, with stocks and stock-sales ratios increasing relatively faster at retail than at wholesale.

After the middle of 1951 consumer buying was scaled down to some degree and businessmen found themselves in an overstocked position This resulted in an inventory subcycle, i.e., disinvestment during a period of advancing economic activity. Trade inventories were drawn down until mid-1952—to a somewhat low position relative to sales—when strong consumer purchasing was once more resumed with an expansive effect on inventory decisions.

#### THE 1953-57 CYCLE

Stocks rose again at a fairly rapid rate, and by the third quarter of 1953 had exceeded the abnormally high point reached in mid-1951. Sales, on the other hand, turned slugglish in early 1953 and showed no further growth before turning down in the final months of that year. The onset of the business recession in mid-1953 was followed by a shift from inventory accumulation to liquidation. The liquidation was steady though not sharply downward—about \$1 billion concentrated in durable goods between the fall of 1953 and the end of 1954. Consumer purchases fell off only slightly in this cyclical downturn, a factor important in dampening the inventory decline, and began to move upward early in 1954.

The increase in consumer demand geared to other expanding elements in the overall economy influenced the resumption of inventory accumulation after 1954. A factor of great importance in trade behavior was the sharp impetus from the record automobile demand throughout the year 1955. Trade inventories, from the end of 1954 to the end of 1956, increased by \$5 billion. The upswing slowed
in 1957—the further rise occurred primarily among retail automotive dealers and durable goods wholesalers—but by the end of 1957 the seasonally adjusted book value of trade stocks had risen by another billion dollars to a total of \$37 billion.

The recession which began in the third quarter of 1957 was uneven in its impact and of short duration. Retail sales fell off only about one-half billion dollars or 3 percent to mid-1958. Wholesale sales, which had declined about 1 percent in the first half of 1957, decreased about 8 percent by the first quarter of 1958.

The impact was sharpest on the durables, a manifestation characteristic of all the postwar recessions. Consumer purchases at nondurable goods stores were down only 1 percent during this decline while those at durables fell off about 10 percent, with the automotive group having one of their worst years as new passenger car sales fell to 4.7 million in 1958 as compared to 6 million in 1957 and 7.2 million in 1955.

Trade inventories had continued to rise during 1957 though at a declining rate and did not begin to fall until the first quarter of 1958. The rate of liquidation was rather sharp but only two quarters in duration. Total trade stocks were down about \$1½ billion, seasonally adjusted, in the first half of 1958—with automotive dealers accounting for one-half of the decline—then moved upward again as conditions improved.

Through 1959 and 1960 stock accumulation was considerable, affected in part by an increase in the number of passenger cars in dealers hands as the introduction of compacts increased the number of car models offered for sale. The book value of trade inventories rose by over \$4 billion by the end of 1960. Automotive dealers' stocks rose by more than \$1 billion as the number of new cars in stock increased from 580,000 at the end of 1958 to over 1 million at yearend 1960.

The period of stock liquidation in the 1960–61 downturn was again of very short duration—lasting only 4 months with the decline amounting to about \$1 billion seasonally adjusted. The reduction, as in earlier downturns, was again concentrated in automotive dealers' stocks as motor vehicle producers slashed output in order to achieve a better balance between stocks and the weakened auto market. Accumulation began again in the second quarter as conditions in the national economy began a sizable recovery. Trade stocks slipped somewhat in the third quarter of 1961 as automotive dealers' inventories were affected by a strike in the auto industry.

#### TRADE STOCKS AND GNP

Stock movements in total trade showed considerable similarity of timing in the three cycles since Korea. In the cycles starting in 1953 and 1957 total inventories lagged the movement of GNP at the peak as well as the trough. Total stocks for all trades turned down one quarter after the peak and turned up two quarters after the trough of economic activity as measured by GNP. In the 1960–61 period the pattern on the downswing was followed generally with stocks turning down two quarters after the activity peak. On the upswing, however, the picture was different with inventories showing no lag behind the advance in the economy. This phenomenon was character-

istic of this recovery: in manufacturing, for example, the inventory lag was shorter than average while shipments atypically showed no lag behind new order receipts.

Durable goods stocks accounted in large part for the patterns exhibited by inventories in the three cycles. When the nondurables are considered separately it is found that there is little consistency in their patterns of behavior during the cycles. In the 1953-54 period nondurable stocks advanced for three quarters after GNP peaked, and declined for one quarter after the trough. In the 1957-58 period the level of nondurable inventories showed little change between peak and trough, and in 1960-61 rose during both the cyclical decline and the recovery.

The expansion of nondurable goods stocks during the 1953-54 downturn in economic activity offset the liquidation in durables so that trade inventories did not constitute a drain on GNP. In each of the other three postwar recessions, however, about one-fifth of the total reduction in output was attributable to the net change in trade stocks.

The contribution of investment in trade inventories during the early stages of economic recoveries has usually been quite small. An exception is in the first quarter of the 1961 recovery when trade stocks (primarily in wholesaling) moved up appreciably.

## **B.** FACTORS AFFECTING INVENTORY CHANGES

The factors affecting inventory investment in trade are many and complex and only a few are readily measurable. The growth in demand as measured by sales is a dominant factor in explaining the level of inventories carried by dealers. In the postwar years with economic activity, and consequently sales, tending generally upward, inventories of goods have risen as well to new high points after each cycle.

The introduction of new products and the change in the nature of the products sold also play a part in raising inventory requirements although sometimes these requirements may be transitory. The introduction of compact passenger cars into the automobile market operated to increase the number of models required to be stocked. In time, however, the product mix may gradually shift with the models for which the demand is weaker tending to be eliminated. To a certain extent the introduction of new products may serve to supplant older products previously carried.

There are other factors that may cause the level of stocks to decline in the long run. Among these are (1) the introduction of new processes that speed up production and increase capacity. This coupled with the improvement in transportation permits lower stocks to be carried at the wholesale and retail level for a given volume of sales. To a certain extent these factors enable distributors to pass back to the manufacturer part of the burden of carrying stocks needed to meet (2) Standardization of lines within products sales commitments. Special market pressures may disturb this tendency from me, as in the automobile case already mentioned. (3) Concarried. time to time, as in the automobile case already mentioned. tinued advances in inventory controls that permit the more efficient movement of goods from the time of purchase to the time of sale. These factors have all tended to control, relatively, the quantity of stocks necessary to meet the steadily increasing demand for goods as

distributors have become increasingly aware of the increased costs entailed in overstocking.

#### MEASUREMENT OF INVENTORY BEHAVIOR

The task of investigating the dynamic factors in inventory management is a very difficult one. The number of variables involved, as indicated above, are many, and frequently it is impossible to assign numerical quantities to these variables in order to evaluate their ultimate effect on inventory variation.

In attempting to determine the nature of the movements of trade inventories over both long and short time intervals and to evaluate critically the contribution of various factors to this variation, a number of relationships were utilized. The most important single factor related to the level of inventories in trade distribution is that of demand for goods as measured by sales. To measure the contribution of sales to the variation in inventories a relationship was introduced of the form

$$I_{\iota} = a + b S_{\iota-1}$$

where  $I_i$  represents the value of inventories at the end of a given quarter and  $S_{i-1}$  represents the average monthly sales for the previous quarter. Seasonally adjusted inventories and sales are used in this and all other relationships presented here.

The lagging of stocks behind sales, representing the time period required to effect the adjustment of stocks to the changing demand situation, was found to significantly improve the relationships in each case. The choice of lag-end-of-period stocks with average sales in the preceding quarter, or about 4½ months-was found to be best for the majority of trade lines after also testing sales in the current quarter (an inventory lag of 1½ months) and sales in the second preceding quarter (a lag of 7½ months). A lag of 1½ months was superior to the 4½-month lag for retail food stores.

In this formulation the coefficient "b" will give the change in stocks associated, on the average, with a unit change in sales at any time period. This value "b" obtained here by correlation procedures is in essence the marginal inventory coefficient. In this equation also the positive slope (always obtained) implies that equilibrium stocks are larger at higher levels of sales. These regression formulas, covering the period from 1948 to the second quarter of 1961,¹ are shown in table 8 for selected wholesale and retail trade groups. Also shown are the indexes of determination  $(\mathbb{R}^2)$ . The coefficients are all highly significant, and the R²'s indicate that from 75 percent to 98 percent of the variations in inventories are explained by changes in sales during the previous period. It may be noted that the higher the aggregation, the larger the R²-so that in total retail and total wholesale trade well over 90 percent of the inventory variations are explained by fluctuations in the levels of sales. Chart VI compares actual inventories held by durable and nondurable goods establishments with the values calculated from the sales regressions.

¹ For retail trade, the Korean period (from mid-1950 to mid-1953) was excluded.

CHART VI



**RETAIL AND WHOLESALE TRADE:** 

*Based on regression equations relating inventories and sales. See text table.

U. S. Department of Commerce, Office of Business Economics

	R²
REGRESSIONS IN CURRENT DOLLARS	
Retail trade:	
Total $I_t =003 + 1.476 S_{t-1}$	0.98
Durable goods stores $I_t = -1.659 + 2.246 S_{t-1}$	. 95
Nondurable goods stores $I_t = 1.657 + 1.059 S_{t-1}$	. 98
Automobiles and parts $I_{t-1} = -1.211 + 1.724$ St-1	. 87
Furniture and appliances It=.186+1.994 St-1-	. 84
Lumber, building, hardware $I_{s} =504 + 3.138 S_{t-1}$	. 89
Food It=.554+.687 St	. 98
Apparel $I_1 = .020 + 2.887 S_{t-1}$	. 81
General merchandise $I_t =046 + 2.183 S_{t-1}$	. 92
Wholesale trade:	~
Total $I_{t} =028 + 1.097$ S ₁₋₁	. 90
Durable goods establishments 1 = 1.602+.789 St-++.035 t	. 97
Nondurable goods establishments 1 _t =.461+.760 S _{t-1}	.08
REGRESSIONS IN CONSTANT (1954) DOLLARS	
Retail trade:	
Total L= .836+1.376 St-1	. 93
Durable goods stores $I_{i}=1.990+1.443$ St-1	. 85
Nondurable goods stores $I_{i} =580 + 1.283 S_{i-1}$	. 94
Wholesale trade:	
Total $I_t = .565 + 1.006 S_{t-1}$	. 92
Durable goods establishments $I_t = 1.080 + 1.177 S_{t-1}$	. 78
Nondurable goods establishments It=1.06+.817 St-1	. 90

TABLE 8.—Wholesale and retail trade—Regression equations 1

NOTE.-

rg.— I₁=Book value of inventories at end of quarter. S₁=Average monthly sales in current quarter. S₁-1=Average monthly sales in preceding quarter. t=Number of quarters since last quarter of 1947; thus, fourth quarter of 1961-56.

R = Correlation coefficient.

¹ Based on the period from 1st quarter 1948 through 2d quarter 1961, quarterly seasonally adjusted data, in billions of dollars. Retail trade regressions exclude Korean period.

Source: Office of Business Economics, U.S. Department of Commerce.

Part of these "closeness of fits" is traceable to the sizable increases in consumer and wholesale prices over the postwar period. Price increases have roughly equivalent effects on sales and stocks-although reflected somewhat slower in stocks. The relationships were therefore recomputed in "real terms" and, as expected, yielded appreciably lower R's-except in the case of wholesale nondurables.

The regression formulas indicate that, on the average for the 1948–61 period, an increase (decrease) of \$1 billion in monthly retail sales was associated with an increase (decrease) of about \$11/2 billion in inventories in the following period; for wholesale a change of \$1 billion in sales was associated on the average with a change of \$1.1 billion in the following period. The average deviation between the actual and calculated values during the postwar years amounted to 21/2 percent for both retailers' and wholesalers' stocks.

It may be noted, given the form of the relationship, that the coefficients of sales are quite close to the average stock-sales ratios in the various lines of trade. Some differences arise from the lag in stocks which has been introduced, and from the "constant" value in the equation.

#### ADDITIONAL RELATIONSHIPS

Because a good deal of inventory variation is not explained by the use of sales alone it was thought advisable to introduce a linear time factor as a catchall to represent the resultant effect of all other variables. The utilization of the relationship

$$I_i = a + bS_{i-1} + ct$$

yielded mixed results. Generally, since the growth factor in the postwar period has been so strong for both sales and inventories, there appeared to be too much intercorrelation between the variables S and t. As a result the coefficients of t in the equations derived were generally found to be not significant and the relationships obtained of doubtful value in most cases.

In another approach, an effort was made to remove the effect of serial correlation, a defect that tends to vitiate the use of sales levels in explaining short-term inventory behavior. This was done by eliminating the effect of level on the relationship between inventories and sales by using an equation of the form

$$\Delta I_{i} = a + b \Delta S_{i-1}$$

where  $\Delta I_i$  is the change in inventories during a given quarter while  $\Delta S_{i-1}$  is the change in average sales between the second preceding and preceding quarters.

In general it was found that the change in inventories tended to lag changes in sales by about a quarter. The variations from the average were very large, however, and the deviations in the inventory changes left unexplained by the use of sales changes alone were much too large to permit the effective use of this formula as an estimating procedure.

For improving this relationship it was then decided to introduce the level of inventories itself as a separate independent variable in the equation. In theory it was felt that the level of inventories at the beginning of a given quarter would in itself influence the stock purchasing policies of businessmen insofar as they applied to the level of sales at that time. Basically a high level of inventories at a given point of time would serve to induce them to increase their stocks by a smaller amount relative to a given sales situation than a lower level of inventories under the same sales conditions. The equation used was

$$\Delta I_i = a + b \Delta S_{i-1} + c I_{i-1}$$

where  $I_{t-1}$  is the value of the level of inventories at the beginning of the quarter and the other variables are as previously defined. Under these conditions the coefficient c would be expected to be negative. The results of the addition of this variable were generally poor, although an inverse relationship with inventory changes did in fact generally appear.

Still another approach would be in the use of the stock-sales ratio directly as a measure of inventory position or condition. The use of the stock-sales ratio in this manner would imply the existence of certain norms to which businessmen would quickly react to correct imbalances. Such norms would naturally differ substantially among the various trades and may shift considerably over time and over the cycle.

The correlations between inventory changes in a given quarter and the stock-sales ratios at the beginning of the periods yielded significant results. These relationships were inverse—i.e., high (low) stock-sales ratios were associated with negative (positive) inventory changes over the near term. The R's, however, were relatively low—running about 0.5.

It is interesting to note that much better results were obtained in manufacturing where direct information on manufacturers' views of inventory conditions could be substituted for stock-sales ratios. This information is collected in the Department of Commerce survey of inventory and sales expectations where manufacturers are asked to classify their inventories (relative to sales and unfilled orders position) as either "high," "low," or "about right."

Many additional correlations were checked, generally with poor results. Some of the independent variables correlated with changes in wholesale and retail inventories were the level of inventories, level of sales, prices, interest rates, changes in manufacturers' unfilled orders, and various combinations of these variables.

Ο